The Effects of Organisational Factors on Employee

Process Innovation:

An Empirical Study of China

Jingjing Wang

A thesis submitted in fulfilment of the requirements for the degree of

Doctor of Philosophy

School of Management

University of Tasmania

October 2013

Declaration of Originality

This thesis contains no material which has been accepted for a degree or diploma by the University or any other institution, except by way of background information and duly acknowledged in the thesis, and to the best of the Candidate's knowledge and belief contains no material previously published or written by another person except where due acknowledgement is made in the text of the thesis.

> Jingjing Wang April 2013

Mypg (eng Oct-2013

ii

Authority of Access

This thesis may be made available for loan and limited copying in accordance with the Copyright Act 1968.

Jingjing Wang April 2013

Mypg Cens Oct-2013

Abstract

Innovation is essential for organisational competitiveness and success (e.g. Edwards, Delbridge & Munday 2005). However, the existing research mainly focuses on the radical innovation of products; few studies have examined the incremental improvement of processes in relation to daily work (Stamm 2003; Axtell et al. 2000; Genaidy et al. 2010). It is therefore imperative to study the process innovation of employees in the workplace.

Employee innovation can be viewed as a process comprising different stages (Scott & Bruce 1994). However, existing research tends to combine different stages of the innovation process together, resulting in erroneous conclusions regarding the effects of organisational factors on different stages of innovation (Magadley & Birdi 2012). The current study therefore aims to examine the effects of organisational factors including the quality of the supervisor–employee relationship, internal communication, external communication, monetary reward, non-monetary recognition and training on the different stages of employee innovation behaviour that produces incremental refinements and improvements to daily work processes.

Using a convenience sample, a questionnaire survey was conducted with 1,299 organisation employees across four industries in China. The data were analysed using multiple regressions to test the hypotheses. The results show that organisational factors have different effects on the stages of employee innovation. Specifically, the quality of the supervisor–employee relationship, external communication, non-monetary recognition and training were found to play a significant role in facilitating opportunity exploration, while non-monetary recognition and training encouraged idea generation and experimentation. Moreover, the quality of the supervisor–employee relationship, internal communication, external communication, non-monetary recognition and training helped to distribute and promote ideas.

iv

The findings suggest that business managers could utilise organisational resources to improve the innovative behaviour of employees in daily work rather than focusing solely on product innovation. Business managers should also use the correct organisational factors to target different stages of employee innovation in order to maximise employee innovative behaviour.

One limitation of the research lies in its focus on China, which restricted the generalisability of the findings in different situations. Another limitation resulted from using the quantitative method, through which the mechanisms of how organisational factors affect the different stages of employee process innovation could not be revealed. Future research could use qualitative data to shed light in this regard.

This research paves the way for future studies to explore whether the findings also hold for other countries and cultures. Future research is needed to consider the direction of the associations in order to provide evidence of causality.

Acknowledgements

There are many people to whom I owe sincere thanks for their assistance and support throughout the journey of this thesis. First, I would like to express my gratitude to my supervisors, Dr Fan Liang and Dr Megan Woods, for their academic guidance, useful suggestions and way of approaching problems. I was lucky to have their assistance and I would not have been able to finish this thesis without their guidance, enthusiasm, encouragement and understanding.

I would like to thank the business staff at the University of Tasmania for their kind, generous and helpful support in every way. Special thanks go to the wonderful people in the Faulty of Business for their support and encouragement. Many names could be mentioned here, but some of them are impossible not to mention: Professor Gary O'Donovan, Dr Mark Dibben, Dr Martin Grimmer, Dr Rob Hecker, Dr Stuart Crispin and Dr Tommy Wong. I would also like to thank Mr Phillip Patman for the advice and guidance regarding data analysis plan he has given me.

My thesis was proofread by Elite Editing, and the intervention was restricted to Standards D and E of the *Australian Standards for Editing Practice*. I would like to thank Elite Editing for their helpful advice.

I wish to convey my gratitude to all of my friends here and in China for their support and encouragement in every possible way throughout the process of this study. I am sincerely grateful to them for enabling me to complete this study and to have a good time during my study in Hobart, Australia.

Finally, I would not have made it without the love of my family. I would like to extend my appreciation to my family for their unconditional love, encouragement and support. My parents have always encouraged me to pursue my education and supported me emotionally and financially. I can never thank my parents enough for their love, support and patience. I would also like to thank my uncle for his support. This thesis is especially dedicated to my beloved family.

Without all of the help from the above-mentioned people, this thesis would not have seen the light of day. Thank you all.

Contents

Declaration of Originality	ii
Authority of Access	iii
Abstract	iv
Acknowledgements	vi
Contents	viii
List of Tables	xi
List of Figures	xii
List of Abbreviations	xiii
Chapter 1: Introduction	1
1.1 Background of research on innovation	1
1.2 Process innovation of employees	4
1.3 Background of employee innovation of China	5
1.4 Research opportunity for the thesis	7
1.5 Structure of the thesis	7
Chapter 2: Literature Deview and Hypotheses Development	Q
2.1 Stages of employee innovation	oور م
2.1 Stages of employee innovation in workplaces	
2.2 Organisation and employee innovation in workplaces	,11 11
2.2.1 Introduction	11 1 <i>1</i>
2.2.2 Leadership.	14
2.2.2.1 Introduction	14
2.2.2.2 LWA theory for process in postion	10
2.2.2.5 LWA theory for process innovation	/ 1
2.2.2.4 Hypotheses development	10
2.2.3 Communication	20
2.2.3.1 Introduction	20 22
2.2.5.2 Internal communication and employee innovation	22
2.2.5.5 External communication and employee milovation	
2.2.5.4 Communication for process innovation	23
2.2.5.5 Hypotheses development. Internal communication	20
2.2.4 Reward and recognition	21 20
2.2.4 Reward and recognition	20 28
2.2.4.1 Introduction	
2.2.4.2 Rewards system and employee mnovation	
2.2.4.5 Hypotheses development	
2.2.5 Halling	
2.2.5.1 Introduction	
2.2.5.2 Training and process innovation	,
2.2.5.5 Hamming and process minovation	
2.2.5.4 Hypotheses development	,
2.5 Chapter summary	

Chapter 3: Research Methodology	.40					
3.1 Research methodology						
3.2 Sampling						
3.3 Survey instrument						
3.3.1 Employees' innovative behaviour	.44					
3.3.2 Quality of the supervisor–employee relationship	.45					
3.3.3 Internal communication at work	.45					
3.3.4 External communication at work	.46					
3.3.5 Monetary reward	.46					
3.3.6 Non-monetary recognition	.46					
3.3.7 Training	.47					
3.3.8 Control variables	.47					
3.4 Common method variance (CMV)	.48					
3.5 Reliability and validity	.50					
3.6 Pre-test	.52					
3.7 Ethical considerations	.53					
3.8 Language for the survey	.53					
3.9 Data collection	.54					
3.10 Sample profile	.55					
3.11 Correlation analysis	.56					
3.12 Multiple regressions	.58					
3.13 Chapter summary	.59					
Chanton 4. Analysis and Desults	८ ೧					
4.1 Dringingl component analysis	.00					
4.1 Finicipal component analysis	.00					
4.2 Contrations	.02					
4.3 1 Opportunity exploration stage, generating and experimenting ideas	.05					
4.5.1 Opportunity exploration stage, generating and experimenting ideas	66					
A A Chapter summary	.00					
4.4 Chapter summary	. / 1					
Chapter 5: Discussion	.72					
5.1 Supervisor–employee relationship	.72					
5.2 Internal communication	.73					
5.3 External communication	.75					
5.4 Reward and recognition	.76					
5.5 Training	.78					
5.6 Effects of organisational factors on stages of employee process						
innovation	.80					
5.6.1 Opportunity exploration stage	.82					
5.6.2 Idea generation and experimentation stage	.83					
5.6.3 Idea promotion stage	.85					
5.7 Chapter summary	.87					
Chapter 6: Conclusion	.88					
6.1 Implications	.88					
6.1.1 Theoretical implications	.88					
6.1.2 Practical implications	.90					
6.2 Limitations of the study	.91					
6.3 Future research directions	.93					
6.4 Summary	.95					
•						

Reference	
Appendix A: Survey Instrument	
Appendix B: Survey Instrument (Chinese Version)	

List of Tables

Table 3.1: Summary of results of Cronbach's alpha test 50	0
Table 3.2: Summary of results of validity test 52	2
Table 4.1: KMO and Bartlett's Test 6	1
Table 4.2: Means, standard deviations and Pearson correlations among	
variables (N=1,299)64	4
Table 4.3: Results of regression analyses on opportunity exploration	7
Table 4.4: Results of regression analyses on generating and experimenting	
ideas68	8
Table 4.5: Results of regression analyses on idea promotion	9
Table 4.6: Summary of the hypotheses testing results 70	0

List of Figures

Figure	2.1: F	Proposed	the	oretical model		•••••	•••••	•••••	••••		.14
Figure	5.1:	Effects	of	organisational	factors	on	different	stages	of	the	
employee innovation process					.81						

List of Abbreviations

CMV	common method variance
EFA	exploratory factor analysis
FDI	foreign direct investment
НК	Hong Kong
HR	human resources
HREC	Human Research Ethics Committee
HRM	human resource management
КМО	Kaiser–Meyer–Olkin LMX
	leader–member exchange
PCA	principal component analysis
SPSS	Statistical Package for the Social Sciences
VIF	variance inflation factor

Chapter 1: Introduction

This chapter provides an introduction to the thesis. It commences with a discussion of the rationale of this thesis. Next, the broad research opportunity is briefly discussed. The chapter concludes with a discussion of the structure of the thesis.

1.1 Background of research on innovation

Innovation is essential for organisational competitiveness and success (McAdam & Keogh 2004; Edwards, Delbridge & Munday 2005). It is widely claimed to have beneficial influences on the effectiveness and long-term survival of organisations (e.g. Amabile 1988; Ancona & Caldwell 1987; Kanter 1988; Mumford 2000; Woodman, Sawyer & Griffin 1993), and it has been widely recognised as an important source of competitive advantage for organisations that want to prosper and grow (Damanpour & Schneider 2006; Tellis, Prabhu & Chandy 2009; Yuan & Woodman 2010). Therefore, it is not surprising that a significant amount of literature has been generated on the subject of innovation, which has become a broad concept that can be defined in many ways (Rowe & Boise 1974; Dewar & Dutton 1986; Rogers 1983; Utterback 1994; Afuah 1998; Fischer 2001; Garcia & Calantone 2002; Kuczmarski 2003; McDermott & O'Connor 2002; Pedersen & Dalum 2004; Frascati Manual 2004).

The literature on innovation is diverse and complex, and it covers many subject and research fields such as organisational behaviour, education and sociology (Smith et al. 2008). Due to the complexities associated with innovation research, it is difficult to generate one grand theory or best practice of innovation (Tidd 2001). Meanwhile, the theories discussed hold true in different circumstances relating to the types (Damanpour 1987; Damanpour, Szabat & Evan 1989) and stages of innovation (Gopalakrishnan & Damanpour 1994). To clarify the interpretation and usage of the construct of innovation, it is necessary to define the term _innovation' and what it is concerned with here. The common features that underlie innovation include the concepts of novelty (King & Anderson 2002), potential usefulness to the organisation (West & Farr 1990) and implementation. Ideas are considered novel if they are unique relative to other ideas currently available; ideas are viewed as potentially useful if they have the potential for direct or indirect value to the organisation (Amabile 1996; Zhou & Shalley 2003). Innovation includes both idea generation and implementation (West 1997; West & Farr 1990), where implementation involves making innovations a part of regular work processes (Kleysen & Street 2001) and developing, testing and modifying them (e.g. Kanter 1988).

In addition, innovation can be radical or incremental in processes (Atuahene Gima 1996; March 1991), related to products or processes (Abernathy & Utterback 1978; Davenport 1993; Han, Kim & Srivastava 1998), and it can be administrative or technical in scope (Damanpour 1991). A brief review of the typology of innovation types is presented below (Damanpour 1991).

For the purposes of this research, the first typology of innovation types relates to the degree of innovation. The types of innovation can range from radical to incremental (Ettlie, Bridges & O'Keefe 1984; Normann 1971). It has been widely agreed that significant differences exist between radical and incremental innovations in terms of their influence on organisations (e.g. Dosi 1982; Christensen 2000).

Radical innovation relates to fundamental changes that represent revolutionary changes in technology; they represent clear departures from existing practice (Ettlie 1983; Ettlie, Bridges & O'Keefe 1984; Urabe 1988). Such changes refer to discontinuous events that result from deliberate research and development activity (Freeman & Perez 1988). In particular, radical innovation can be considered new technology or a combination of technologies that are induced commercially to meet users' or markets' needs (Abernathy & Utterback 1990). Radical innovation typically includes aspects regarding high market and technological uncertainty, new market creation and current product cannibalisation (Hurmelinna-Laukkanen,

Sainio & Jauhiainen 2008). Radical innovation tends to be closely relevant to unexpected changes in technology for products or services that might create high levels of uncertainty in organisations.

In contrast, incremental innovation refers to changes that mainly reinforce the existing capabilities of organisations (Dewar & Dutton 1986; Ettlie, Bridges & Keefe 1984; Henderson & Clark 1990). While radical innovation regarding major breakthroughs in the development of new products tends to originate in research and development (R&D) (Stamm 2003), incremental innovation concerning incremental modifications in procedures at work (Mumford & Gustafson 1988) typically tends to be generated and developed within a cross-functional context (Stamm 2003).

However, few theorists explicitly research small-scale incremental innovations, that is, looked like less significant and minor changes and improvements in the work. This presents a research opportunity to examine the minor changes and improvements in employees' daily work.

Innovations can vary in content; specifically, previous research has proposed that innovation can range from product innovation to process innovation (e.g. Patterson & Lightman 1993). Product innovation or new product development refers to the development of new or improved products. Product innovation has been a central topic in innovation research (e.g. Brown & Eisenhardt 1995; Clark & Fujimoto 1991; Cohen & Levinthal 1990; Cooper & Kleinschmidt 1986; Leonard-Barton 1992; Sanchez & Mahoney 1996; Teece 1986; Ulrich & Eppinger 2008; Zirger & Maidique 1990). Conversely, process innovation corresponds to the sequences and nature of production processes. According to Afuah (1998), process innovation is concerned with introducing new elements into an organisation's operations—for example, changing task specifications, improving work and information flows, and producing a product or service using equipment.

From the above discussion, innovation can range from small-scale ideas relating to improvements in daily work processes and work designs to the development and implementation of new ideas that innovate theories, practices or offerings regarding products and/or services across the organisation (Axtell et al. 2000). This implies that, in addition to the development of new offerings such as improved products or services to the market, the contents of innovation can be relevant to the methods of doing jobs and changes relating to organisational characteristics such as the structure of organisations.

Compared with product innovation, process innovation is usually more difficult to detect, but it is also very important (Propris 2002). Process innovation that relates to the development of new or improved processes is related to employee's daily work performance. However, there is a lack of understanding regarding what the employee perceives as motivating for engaging in innovative behaviour towards work processes. Process innovation has received much less attention (e.g. Davenport 1993; Ettlie & Reza 1992; Hatch & Mowery 1998; Pisano 1997; Reichstein & Salter 2006). This raises a question regarding whether radical product innovation theory works in the same way when applied to the area of innovation towards small-scale improvements and refinements in existing work processes. The next section reviews the literature on process innovation.

1.2 Process innovation of employees

The importance of employee innovation is not limited to R&D employees or knowledge workers, and employees may generate innovative ideas in any job (e.g. Madjar, Oldham & Pratt 2002; Shalley, Gilson & Blum 2000; Shalley, Zhou & Oldham 2004). Everyone has the potential to be innovative, and innovation can come from anywhere within an organisation (Amabile 1996). For example, employees need to be innovative to solve day-to-day problems and to provide input to organisations' innovations (Genaidy et al. 2010). Thus, the importance of the incremental process innovation of employees concerning non-technological improvements in their daily work should not be ignored.

Whether radical or incremental, innovative behaviour by employees is at the root of every innovation (Glynn 1996). To meet the needs of global competition and the uncertain environment, organisations need their employees to undertake not only basic duties, but also make extra efforts to act beyond the routine performance of their duties (Axtell et al. 2000) in order to engage in innovative behaviour in work processes. Employee innovative behaviour has been proposed as an important asset that enables an organisation to succeed in a dynamic business environment (Tortoriello & Krackhardt 2010). This illustrates that the innovative behaviour of employees deserves research attention.

Specifically, ideas generated by employees can range from small, incremental refinements in procedures to radical, major breakthroughs in the development of new products (Dewar & Dutton 1986; Madjar, Greenberg & Chen 2011; Mumford & Gustafson 1988). Among these, technological product innovations regarding significant technological improvements of products were the most researched (e.g. Chandy & Tellis 1998; Flor & Oltra 2004), but only a few studies explicitly examined non-technological process innovation—that is. incremental modifications to everyday work processes. Moreover, some innovation studies did not specify which type of innovation they examined (Damanpour 1991), which can lead to erroneous conclusions if theory in employee innovation work in different ways for different types of innovation. Consequently, while few studies explicitly examined innovation towards small-scale improvements and refinements in existing work processes (Axtell et al. 2000), the present study exclusively examines incremental innovation regarding small-scale refinements in everyday work processes.

1.3 Background of employee innovation of China

China is the largest transition economy in the world (Prasad 2004). In the 2000s, China reported double-digit annual gross domestic product (GDP); thus, by 2012, annualised GDP growth was 7.8% (National Bureau of Statistics of the People's Republic of China 2008: 865). China was generally viewed in terms of its astonishing growth as a manufacturing economy, and it has emerged as an exporting country (Girma, Gong & Gorg 2009). Additionally, China has experienced phenomenal growth in foreign direct investment (FDI) (Warner 2004). However, the environment has recently been changing dramatically. In particular, FDI projects showed a rapid downward trend from 2008 to 2009 (An, Shi & Chi 2010). A further increase of exports is limited by the world's market demands, which are subject to fluctuations in the world economy and intensifying competition from other countries.

In such a changing environment, it is important for China to identify new drivers in order to maintain its economic growth. Innovation is another facilitating force to improve efficiency and competitiveness in an organisation through innovative activity. Innovation allows organisations to develop new processes to produce existing goods more efficiently or to develop new products that allow them to grow (Girma, Gong & Gorg 2009). According to Jiang, Wang and Zhao (2012), western organisations increasingly have business relations or establish branches in China, and these organisations are interested in how to facilitate employee innovation. Innovation in Chinese organisations requires more research.

However, although existing literature indicates that organisational factors are related to employee innovative behaviour (e.g. Zhang & Bartol 2010; Wei, Liu & Herdon 2011), research on employee innovative behaviour in the workplace has rarely been conducted in a non-western context. As Jackson and Bak (1998) noted, it would appear that management is not sufficiently aware of which factors may and may not work in a different context. Specifically, few recent empirical studies have examined the link between organisational factors and employee process innovation towards incremental refinements in daily work in the Chinese context. It has been proposed that western theories might not work in China (Jackson & Bak 1998), but current work has not gone any further to discuss how or why they might differ. Applying existing western theories in China can determine whether there are differences so that people can understand which theories, if any, might not hold in China. In particular, questions remain regarding whether western theory works in the same way when applied in the Chinese context. Do western practices matter in terms of the influence on employee innovation in China?

Thus, it is worthwhile to examine whether western theories of employee innovation can be utilised in the Chinese context and whether employee innovation can be affected by organisational factors. By doing so, this study expects to provide suggestions for practicing managers in China who are keen to encourage employee innovation and therefore improve the competitiveness of organisations.

1.4 Research opportunity for the thesis

The present study addresses the research need by extending research on the influence of organisational factors on employee innovation—in particular, on employees' behaviour in innovation processes. Specifically, in line with the preceding discussion, the research opportunity for investigation in this thesis is the influence of organisational factors on employees' behaviour towards refinements and increasing the efficiency of existing practices and processes in daily work in China. By doing so, this study is expected to advance the understanding of what motivates employee innovative behaviour in process innovation. It is one of the first studies to apply western innovation theories to China.

1.5 Structure of the thesis

This thesis proceeds in six chapters. Chapter 2 provides a theoretical background of employee process innovation and its link with organisation-related factors. Chapter 2 also discusses the specific research questions to be addressed in this thesis, as well as hypothesis developments. Chapter 3 outlines the research method adopted in this study. Chapter 4 presents the results of the quantitative analysis of the data gathered in this research. Chapter 5 provides a discussion of the key findings as they relate to the proposed hypotheses and research questions posed in Chapter 2. Chapter 6 presents the theoretical and practical implications, and it recommends future research opportunities.

Chapter 2: Literature Review and Hypotheses Development

The objectives of this chapter are three-fold. First, it provides an account of the theoretical development of the stages of employee innovation perspectives, and it briefly outlines the main findings in this research area. Second, it discusses the link between employee innovation and supervisor–employee relationships, internal communication, external communication, monetary reward, non-monetary recognition and training. Finally, as an integration of these theoretical developments, the chapter presents the specific research focus addressed in this thesis, as well as the hypotheses tested in this research.

2.1 Stages of employee innovation

Innovation is a long and cumulative process (Drazin, Glynn & Kazanjian 1999; Shi 2012) that ranges from generating new ideas to implementing them into a new product, process or service (Urabe 1988). In this way, employee innovative behaviour can be viewed as a process of behaviour towards generating and developing new ideas for incremental refinements and improvements in everyday work processes, regardless of the ultimate success of these new phenomena (e.g. Scott & Bruce 1994; Clegg et al. 2002).

Accordingly, models of employee innovation (e.g. Scott & Bruce 1994; Janssen 2000; Rank et al. 2004; Holman et al. 2012) have highlighted the key stages of the innovation process. There is strong evidence to support the existence of at least two broad overlapping innovation stages: idea generation and idea implementation (Amabile 1988; Wolfe 1994; Mumford et al. 2002; Axtell et al. 2000).

Specifically, employee innovation begins with the generation of novel and useful ideas (Amabile et al. 1996; Kanter 1988; Woodman, Sawyer & Griffin 1993). At this stage, employees working in groups or on their own are expected to come up

with ideas in response to a problem or a perceived need for innovation (Magadley & Birdi 2012). Incongruities, discontinuities, perceived work-related problems and emerging trends are often instigators of the generation of novel ideas (Drucker 1985). This stage can be an important prerequisite for innovation (Scott 1995).

The next stage is idea implementation, which has been viewed as _the transition period during which targeted organisational members ideally become increasingly skilful, consistent and committed in their use of an innovation' (Klein & Sorra 1996: 1057). This stage includes making innovations a part of regular work processes (Kleysen & Street 2001) and developing, testing and modifying them (e.g. Kanter 1988).

Some studies (e.g. Janssen et al. 2004; Oldham & Cummings 1996; Scott & Bruce 1994, 1998) further expanded this stage perspective by proposing that once an employee has generated an idea, he or she must engage in social activities to find sponsors surrounding an idea or to build a coalition of supporters who provide the necessary power behind it (Galbraith 1982; Kanter 1988).

However, as Kleysen and Street (2001) argued, measures of employee innovative behaviour often assessed only one dimension of the construct. To develop a multidimensional measure of employee innovative behaviour, Kleysen and Street (2001) conducted a systematic literature review and identified five dimensions of employee innovative behaviour: opportunity exploration, generativity, formative investigation, championing and application. Specifically, opportunity exploration refers to extensive exploration into innovative opportunities in order to discover more related knowledge; generativity refers to behaviour that facilitates reforms to members, products, work procedures, services and organisations; formative investigation refers to different concepts, solutions and comments that are compiled, surveyed and tested; championing refers to different social and political behaviours in the innovation process; and application refers to innovations that are utilised into daily routines within an organisation. Unfortunately, Kleysen and Street's (2001) study did not lend empirical support to a multi-dimension measure with five distinct subscales according to the dimensions derived from the literature. However, their findings suggest that employee innovation includes more than one dimension, and future research could continue to reflect the richness of the construct of employee innovative behaviour.

The review of the existing research on the stages of the innovation process indicates that the employee innovation process typically encompasses a broad set of behaviours related to the generation of ideas, creating support for them and helping their implementation (e.g. Scott & Bruce 1998; Janssen 2000; Rank et al. 2004). These behaviours were dealt with previously as key steps in the process of individual innovation (Axtell et al. 2000; Krause 2004; Dorenbosch et al. 2005). However, previous studies often assessed only one dimension of employee innovation, whereas Kleysen and Street (2001) noted that employee innovation includes more than one dimension.

Further, although it has been proposed that employee innovation process can be seen as a process, existing research tends to combine different stages of the innovation process, resulting in an erroneous conclusion regarding the effects of organisational factors on different stages of innovation (Magadley & Birdi 2012). Little attention has been devoted to the possibility that the factors that affect the idea generation stage may differ from those that promote their implementation (Axtell et al. 2000; Magadley & Birdi 2012). Future research is needed to separately examine the stages of the employee innovation process.

Additionally, the majority of studies on innovation have focused on the idea generation stage (Axtell et al. 2000; West 2002; Magadley & Birdi 2012). However, innovation does not occur when a new idea is generated; it occurs when the new idea is put into practice (Damanpour 1987; Choi & Chang 2009). In support, recent researchers (e.g. Axtell et al. 2000; Dul & Ceylan 2010; Mumford 2003; Oldham & Cummings 1996; Scott & Bruce 1994; Unsworth 1999) have proposed that the scope should be extended and for more attention to be paid to other stages, such as the implementation of ideas. A comprehensive perspective on the employee innovation process is needed.

In response to the above issues, process perspective is taken by the present study. Following Dul and Ceylan (2010), employee innovation is viewed here as a multistage process from opportunity exploration to idea generation and experimentation, and to idea promotion. These are not limited in space and time. Further, as the work environment is a highly important influence in terms of facilitators or inhibitors of creative behaviour (Amabile 1983; Farmer, Tierney & Kung-McIntyre 2003; Mumford & Gustafson 1988; Sternberg & Lubart 1996), organisations may encourage innovation by using their resources. This study envisions a role for organisations to encourage and motivate employee innovation by leveraging common organisation-related resources. The literature on the influence of organisations on employee innovation is reviewed in the following sections.

2.2 Organisation and employee innovation in workplaces

2.2.1 Introduction

Employee innovative behaviour has been widely proposed to be crucial for the effective functioning and long-term survival of organisations (e.g. Amabile 1988; Ancona & Caldwell 1987; Kanter 1988; Oldham & Cummings 1996; Shalley 1995; Van de Ven 1986; West & Farr 1989, 1990; Woodman, Sawyer & Griffin 1993). In view of evidence, employee innovation is essential for organisations (e.g. Lu & Li 2010), and it is highlighted that the generation of new ideas and the translation of these ideas into useful practices are functions of interactions between employees and situations (Amabile 1997; Scott & Bruce 1994; Hunter, Bedell & Mumford 2007). In this way, theory and research have sought to understand how innovation can be fostered or encouraged in the workplace by identifying personal and contextual factors that influence employee innovation (e.g. Amabile 1988; Kanter 1988; Mumford 2000), including personal differences and characteristics (Bunce & West 1995; Barron & Harrington 1981; Feist 1999), organisation culture and climate, relationships with supervisors (Yuan & Woodman 2010), job characteristics (Oldham & Cummings 1996) and social context (Yuan & Woodman 2010).

Among these, a large body of literature has focused on identifying personal characteristics and attributes associated with innovative achievement (Amabile et al. 1996; Barron & Harrington 1981; Davis 1989; Martindale 1989; Williams & Yang 1999), and generation of ideas is viewed as individual asset (Sternber & Lubart 1999). These studies (e.g. Amabile 1983; Oldham & Baer 2011) have found that a stable set of core personal characteristics, such as aesthetic sensitivity, toleration of ambiguity and self-confidence, relate positively to innovative performance. However, while studies on innovation at the employee level mainly focus on the characteristics of creative people, personal characteristics such as the intelligence or personality of employees may be predetermined before they join organisations, which could be difficult for organisations to change and control.

Meanwhile, for innovation to occur in organisations, management needs to support and promote it, as they have considerable influence over the context within which idea generation can occur (Shalley & Gilson 2004). More recent research has started to examine the effects of contextual factors on employees' innovative performance (e.g. Shalley 1991, 1995). In this way, the context in which an employee performs a task influences his or her intrinsic motivation, which in turn affects innovative achievement (Amabile 1988). Specifically, previous studies suggested that employees generate many ideas for improving their organisation when they have complex and challenging jobs (Hatcher, Ross & Collins 1989) and when they participate in decision-making (Axtell et al. 2000). In addition, it suggests that employees are most likely to engage in activities that promote their ideas when both a supportive work context is present and they have a proactive personality (Seibert, Kraimer & Crant 2006).

The literature reviewed above suggests that management should consider these factors if they are interested in maximising their employees' innovative behaviour in the workplace, and a supportive work context plays an important role in supporting and encouraging employees' innovative efforts (Anderson 1975; Pelz 1956; Scott & Bruce 1994; Taylor, 1963, 1972). Accordingly, this study next reviews six organisational factors that may influence employee innovation, including, supervisor–employee relationship, internal and external communication, monetary reward and non-monetary recognition, and training suggested by

previous theory and research as important factors for innovation. These six factors also are seen as the common organisation resources for most organisations to leverage. The proposed model shown in Figure 2.1 summarises these relationships, and the present study reviews the logic of the hypothesised model in the following sections.





Figure 2.1: Proposed theoretical model

2.2.2 Leadership

2.2.2.1 Introduction

According to De Jong and Den Hartog (2007), leadership can be viewed as the process of influencing others towards achieving desired outcomes. The importance of an effective leader for influencing employee innovation has been recognised (e.g. Basadur 2004; De Jong & Den Hartog 2007). The role of leaders

is to provide a structure for the innovation process for employees (Isaksen & Tidd 2006).

Specifically, previous research on the relationship between a leader's behaviour and employee innovation has investigated different leadership styles, including transformational leadership, participative leadership and leader-member exchange (LXM) theory. These studies demonstrated a positive relationship between transformational leadership and creativity (Kahai et al. 2000; Shin & Zhou 2003), and a positive relationship between employee innovative behaviour and employee participation (Axtell et al. 2000).

While the influence of leaders' behaviours on employee innovation has been discussed in general (e.g. Oldham & Cummings 1996), there is little research on how leaders may facilitate the cognitive and emotional processes of daily work in employees (Isaksen & Tidd 2006). In addition, it should be noted that leadership can be a complex interaction between the designated leader and the social and organisational environments (Fiedler 1996). Some scholars suggest that leadership implies an interpersonal interaction procedure between leaders and subordinates (Jacobs 1970; Bass 1985; Robbins & DeCenzo 2001). In this way, leadership can be viewed as a social process that engages everyone in the community (Barker 1997; Drath & Palus 1994; Wenger & Snyder 2000).

Speaking of such an influence, the LMX theory proposes that the quality of the leader–follower relationship influences outcomes such as employee satisfaction, supervisor satisfaction, performance and commitment (Yukl 2002). In particular, for its influence on employee innovation, extant literature suggested that leader–subordinate relationships have positive effects on innovative behaviour (e.g. Janssen & Van Yperen 2004). Leaders support innovation by providing support in risky and challenging tasks, the provision of task-related resources and recognition, which all facilitate individual innovation. More recent meta-analytical findings showed that in terms of supervisor support, LMX is positively related to the innovative behaviour of followers (Harrison et al. 2006). This is because when leaders and followers have good exchanges or high-quality, effective LMX relationships, they share mutual trust, respect and obligations

(Graen 1976; Graen & Schiemann 1978; Graen, Novak & Sommerkamp 1982a, 1982b), positive support, common bonds, open communication, shared loyalty (Dansereau, Graen & Haga 1975; Dienesch & Liden 1986; Graen & Uhl-Bien 1995) and affection (Liden, Wayne & Stilwell 1993).

The following section reviews the literature regarding the general nature of LMX theory for employee innovation. This is followed by a review of the literature in relation to its influence on process innovation. Finally, this study will research the influence of supervisor–employee relationships according to the stages of the employee innovative process.

2.2.2.2 LMX theory and employee innovation

Graen et al. developed the LMX theory (Dansereau, Graen & Haga 1975; Graen 2006; Graen & Uhl-Bien 1995; van Breukelen, Schyns & Blanc 2006). According to Graen et al. (2004) and Schriesheim et al. (1992), the term _leader-member relationship' implies a relationship between a manager and his or her direct-report employees. This term may refer to a relationship between high-level personnel and low-level personnel who directly report to him or her. The term _supervisor-subordinate relationship' in this study can be categorised as one application of the leader-member relationship given that the supervisor is a higher level of authority than the employee who usually reports directly to him or her.

This LMX theory is widely accepted in the management literature to explain the development of a leader–employee relationship and the quality of such a relationship (e.g. Berneth et al. 2007b; Gerstner & Day 1997; Makela 2005; Scherbaum et al. 2006; Sias 2005; van Breukelen, Schyns & Blanc 2006). It has been widely used to evaluate the quality of the supervisor–subordinate relationship (e.g. Fix & Sias 2006; Lee 2001; Sias 2005).

Recent research indicates that LMX relationships are related to employee creativity (e.g. Tierney, Farmer & Graen 1999). According to Tierney, Farmer and Graen (1999), effective LMX relationships are positively associated with employee creativity where creativity refers to the production of novel and useful

ideas by an individual or a group of individuals working together (Amabile 1988; Madjar, Oldham & Pratt 2002; Shalley, Gilson & Blum 2000; Zhou & Shalley 2003). This finding is supported by related research (e.g. Scott & Bruce 1994). Janssen and Van Yperen (2004) also found that high-quality relationships have a positive effect on the broader construct of innovative behaviour.

Preliminary support for the LMX–creativity link was provided (Scott & Bruce 1994), and studies were conducted to investigate the influence of supervisor–employee relationship quality on the generation of ideas (e.g. Tierney, Farmer & Graen 1999; Shalley & Gilson 2004); however, its influence on the employee innovative process by stages (i.e. opportunity exploration, idea generation and experimentation of employees, and the idea promotion of employees) has received little attention. As Denti and Hemlin (2012) suggested, this poses the problem that if the influence of supervisor–employee relationship quality on the stages of the employee innovative process differs from each other, this will not be evident. In response to this issue, they suggested that the stages of the innovation process are one direction for future research on the influence of leaders on innovation. Given the dominant role of leaders in the workplace, research is still needed to identify the myriad of interacting leader and employee factors that may influence the different stages of the employee innovative process.

2.2.2.3 LMX theory for process innovation

The distinction between incremental innovation (i.e. expanding and refining existing products or processes) and radical innovation (i.e. creating new and novel products) may be an important issue in future research and in theoretical modelling of leadership and innovation (e.g. Isaksen & Tidd 2006). For example, Jansen, Vera and Crossan (2009) found that transactional leadership is positively related to incremental types of innovation (i.e. innovation towards refinements of existing products and increasing the efficiency of existing practices and processes) but negatively related to radical innovation. Along these lines, Denti and Hemlin (2012) suggested that examining the influence of leadership research for specific types of innovation might be one direction for future research on innovation and leadership.

Few theorists explicitly research the effect of supervisor–employee relationship quality on process innovation, and empirical studies that explicitly examined its effect on process innovation types are still scarce. Therefore, the present study exclusively examines the influence on employee process innovation towards refinements and increasing the efficiency of existing practices and processes in the daily work, as well as attempts to fill the gap in the existing work by investigating the influence on the different stages of the employee innovative process.

2.2.2.4 Hypotheses development

Supervisor-employee relationship quality refers to the quality of the interpersonal exchange relationship between an employee and his or her supervisor (Graen & Uhl-Bien 1995). Following Graen and Uhl-Bien (1995) and others (e.g. De Jong & Den Hartog 2007), the present study views the quality of the supervisoremployee relationship as a continuous variable that is defined as employees' perceived quality of work relationship with his or her supervisor. Following Liden, Wayne and Sparrowe (2000), the quality of this relationship determines the amount of physical or mental effort, material resources, information and social support that is exchanged between the supervisor and the employee.

The quality of the supervisor–employee relationship is expected to be positively associated with employee process innovation for the following reasons. Several theorists have proposed that the nature of LMX appears to be compatible with employees' innovative actions because LMX quality focuses on the dyadic social exchange between supervisors and employees (Graen & Cashman 1975; De Jong and Den Hartog 2007). Previous research (Dunegan, Tierney & Duchon 1992; Scott & Bruce 1994) showed that high LMX employees believe they operate in a context that is supportive of innovative work. High LMX employees are more likely to engage in job-related risk-taking (Graen & Cashman 1975) and to receive more task-related resources (Graen & Scandura 1987) and recognition (Graen & Cashman 1975; Grean & Scandura 1987; De Jong & Den Hartog 2007). Employees in such relationships frequently interact with their supervisors and have their supervisors' support, confidence, encouragement and consideration. In addition, they take on added duties or expend extra effort to achieve work goals

beyond contractual or transactional expectations (Sparrowe & Liden 1997). As a result, they tend to be more innovative in their daily work process (i.e. towards incremental process innovation).

Additionally, we would more closely examine the influence of supervisoremployee relationship quality on the employee innovation process stage by stage for а number of reasons. Previous studies have indicated that employees' innovative behaviour can be significantly influenced by interactions with their leaders (Yukl 2002; Basadur 2004). Following De Jong and Den Hartog (2007), leaders influence employees' innovative behaviour both through their deliberate actions, which aim to stimulate idea generation and application, as well as by their more general, daily behaviour. This indicates that future quantitative research could further examine whether leaders' practices might have differing influences on different stages of the employee innovative process (e.g. employees' idea generation and/or application behaviour) (De Jong & Den Hartog 2007). Along this direction, Axtell et al. (2000) suggested that employees with more supportive managers were more likely to have their ideas implemented.

While scholars have long commented on the influence of leadership—specifically the supervisor–employee relationship on employee innovation—few empirical studies have been conducted to test this influence on employee process innovation stage by stage. As a result, on the basis of theoretical arguments and previous studies that demonstrate links between the quality of the supervisor–employee relationship and the stages of the employee innovative process towards incremental improvements in daily work processes, this study includes the following hypotheses:

H1a: Quality of the supervisor–employee relationship is positively associated with the opportunity exploration of employees.

H1b: Quality of the supervisor–employee relationship is positively associated with the idea generation and experimentation of employees.

H1c: Quality of the supervisor–employee relationship is positively associated with the idea promotion of employees.

2.2.3 Communication

2.2.3.1 Introduction

With regard to the organisational factors related to the capacity of a company to innovate, there exists a widely accepted agreement concerning the importance of communication, including communication with outsiders and communication within the organisation (Allen 1977; Tushman 1977; Miller & Friesen 1982; Souder 1987; King 1990; Damanpour 1991; Souder & Sherman 1993). Many researchers have proposed that the innovation process essentially refers to communication and information-processing activities (e.g. Allen 1985; Brown & Utterback 1985; Ebadi & Utterback 1984; Fidler & Johnson 1984; Souder & Moenaert 1992; Tushman 1979a, 1979b; Nadler & Tushman 1980), and that innovation is fundamentally a collaborative effort (Subramaniam & Youndt 2005). A large number of empirical studies have revealed that higher levels of communication and information gathering are associated with organisational innovation in general (Aiken & Hage 1971; Evan & Black 1967; Kanter 1982, 1988a; Tjosvold & McNeely 1988) and higher levels of performance in R&D project groups in particular (Katz 1982; Keller 1986; Keller & Holland 1983; Pelz & Andrews 1966).

As can be seen, existing work has proposed that communication could promote an organisation to innovate. At the employee level, recent research has suggested that employee creativity and innovation can be viewed as a social process (e.g. Amabile 1983; Madjar, Oldham & Pratt 2002; Axtell et al. 2000; Woodman, Sawyer & Griffin 1993). In this way, previous studies have focused on the consequences of social networks for employee creativity and innovation (e.g. Obstfeld 2005; Perry-Smith 2006).

In this respect, a large body of literature has stressed the importance of social networks for innovation and pointed to its importance for information flow, the exchange of ideas and access to resources from diverse contacts (Hemphälä &

Magnusson 2012). The rationale for the importance of innovation is that diversity within teams implies greater creativity and innovation due to the differences in knowledge, experience or information among members (Harrison & Klein 2007). Drawing on the social network theory, Perry-Smith and Shalley (2003) also suggested that contacts with diverse associates or external to the firm were expected to enhance important creativity-relevant skills, which refer to the ability to think creatively, such as to generate alternatives. They found that employees with more diverse networks were more exposed to non-redundant information, which would boost deviant behaviour such as opportunity and championing ideas (Perry-Smith & Shalley 2003). Obstfeld (2005) also showed that engineers with dense social networks were more engaged in developing new products or processes. This indicates that having contacts with associates that are both internal and external to the organisation could enhance employee creativity and innovative work behaviour.

Along with the social network perspective, previous studies have showed that employee innovative behaviour can be influenced by interpersonal communications and interactions with others at work (e.g. De Jong & Den Hartog 2010). The literature on organisational culture suggested that open and transparent communication that is based on trust will have a positive influence on promoting creativity and innovation in general (Barret 1997; Robbins 1996). Previous research (Binnewies, Ohly & Sonnentag 2007; Perry-Smith 2006; Perry-Smith & Shalley 2003) has showed that communication about ideas can serve different functions for creativity. Several theorists have proposed that the communication of ideas and information with others should enhance exploring and generating ideas and lead to higher levels of creativity (Amabile 1988, 1996; Ford 1996; Gilson 2001; Kanter 1988; Perry-Smith & Shalley 2003; Woodman, Sawyer & Griffin 1993, 2003) where such diversity is particularly relevant to employee creativity and includes differences in terms of background, areas of specialisation and work responsibilities (e.g. Amabile et al. 1996; Andrews 1979; Payne 1990).

Specifically, extant literature highlighted the importance of communicating outside organisations, such as with customers for creativity and innovation (e.g. Damanpour 1991). It has been suggested that external communication assists

managers' development of knowledge about the external environment, leading to proposals for new ideas for adoption (Damanpour & Schneider 2006).

It is worth noting that, while preliminary support for diverse work contact networks-creativity has been provided (e.g. Perry-Smith & Shalley 2003; Perry-Smith 2006; De Jong & Den Hartog 2010), the influence of having such work contacts on the stages of the employee innovation process received little attention. Most studies tended to focus on either creativity (e.g. Gilson 2001) or employee innovative behaviour in general (e.g. De Jong & Den Hartog 2010). This poses the problem that, if the influence on the stages of the employee innovative process of having such work contacts differs from each other, this will not be evident.

In response to this issue, Perry-Smith and Shalley (2003) suggested that future research needed to empirically explore the social side of individual innovation. Further, Perry-Smith (2006) suggested that it would be helpful for future research to study the effect of social network relationships on various stages from idea generation to implementation. In particular, as Hemphälä and Magnusson (2012) suggested, because different innovations depend on different contingencies, in terms of social networks, it is important for future research to make further distinctions between different stages of innovation. Thus, one direction for making further distinctions is idea generation and idea implementation. This suggests that future quantitative research could more closely examine whether the effect of communication networks might be different stage by stage.

The following section reviews the literature on both external and internal communication research for employee innovation. This is followed by a review of the influence on process innovation in particular. Finally, the study investigates the influence of communication according to the stages of the employee innovative process.

2.2.3.2 Internal communication and employee innovation

According to Frank and Brownell (1989), employee communication is viewed as the communication transactions between individuals and/or groups at various levels and in different areas of specialisation within organisations, which are intended to design and redesign organisations, to implement designs and to coordinate day-to-day activities. As Cummings and Oldham (1997) suggested, creative employees need to be surrounded by colleagues who stimulate them to become passionate about their work but who do not distract them from it. This implies that communication with colleagues within organisations might be important for stimulating innovation in employee work processes.

In this respect, extant literature has addressed the link between relationships with colleagues and creativity (e.g. Perry-Smith & Shalley 2003). Perry-Smith and Shalley (2003) suggested that employees who have frequent interactions with their colleagues tend be less innovative than those who maintain less frequent relationships. The rationale for this is that while frequent interactions develop between similar individuals, less frequent interactions with colleagues are more likely to connect people with differing points of view, varying interests and different ways of tackling problems, and these favour creativity. This corresponds to the fact that, as Muñoz-Doyague and Nieto (2012) suggested, exposure to various approaches and perspectives could stimulate creative abilities such as the capacity to generate alternative approaches, while access to more information could increase the type of knowledge that is relevant for creativity.

Along these lines, prior research stressed the importance of communication among employees at work for employee innovation by highlighting sharing and the exchange of information and diverse ideas. For example, according to Van de Ven (1986), effective internal communication allows employees to share their knowledge and past experiences, and to exchange and discuss ideas, which is especially significant for the generation of new ideas. Similarly, following Andrews and Smith (1996), interactions with other functional areas enhances creativity. Monge, Cozzens and Contractor (1992) also found that group communication is positively related to the generation of innovative ideas. This implies that creative performance may be enhanced as others' views are brought into the discussion via communication within the organisation.
2.2.3.3 External communication and employee innovation

According to De Jong and Den Hartog (2005, 2010), external work contact relates to the frequency of employees' contact with individuals or groups outside the organisation who may form a relevant source of information, inspiration or innovation, such as contact with customers (Kanter 1988), professionals (Kimberly & Evanisko 1981) and scientists (Kasperson 1978).

The literature has emphasised the influence of external contact networks on innovation (e.g. Tidd, Bessant & Pavitt 2001). Hemlin and Olsson (2011) suggested that external contacts might stimulate innovative endeavours. De Jong and Den Hartog (2010) found that the frequency of such contacts relates to employee innovative behaviour—that is, the more external contacts at work, the better knowledge, and supervisors rated their employees on their innovative behaviour. Specifically, previous research (e.g. Damanpour 1991) has underlined the importance of customers as a source of generating new ideas. In addition, Kimberly and Evanisko (1981) found that contact with professionals outside the organisation is related to the increased adoption of innovations.

As can be seen, two issues were highlighted by the above literature review. First, there is a need for future research to examine the effect of communication on employee innovation according to different stages. As Ohly, Kase and Škerlavaj (2010) suggested, future research on idea-related communication and creativity needs to distinguish between the different stages of the innovation process. Second, researchers argued that most innovation studies underscore that individual-to-individual communication is an important factor for innovation (Carmona-Lavado, Cuevas-RodrÃguez & Cabello-Medina 2010; Poolton & Barclay 1998). However, as communication within organisations and outside organisations could have different influences on employee innovation, more research should be conducted to investigate the effect by separating internal and external communication on different stages of the employee innovation process.

2.2.3.4 Communication for process innovation

Silva and Oldham (2012) pointed out that incremental innovative ideas usually represent minor refinements to existing ways of doing things, which are significantly different from radical innovations concerning major breakthroughs in the development of new products. In this respect, as Hemphälä and Magnusson (2012) showed, organisations cannot simultaneously maximise both incremental and radical innovations; they must choose between them or try to optimise, as different innovations depend on different contingencies in terms of social networks. Thus, while they suggested that social networks are a direction for future research in both cases, it is important to make distinctions between different types of innovation. The literature revealed that some studies were conducted for radical product innovation; for example, Kline and Rosenberg (1986) suggested that maintaining external contacts is inevitable to adequately produce a service and be informed about new trends and developments. However, few researchers explicitly clarified the type of innovation they examined or specified the focus on process innovation concerning minor, incremental refinements in procedures. In addition, given that general support was found for communication-employee innovation, few studies explicitly examined internal and external communication for employee process innovation separately.

More attention should be paid to the influence of facilitating employees to communicate with others on incremental idea exchange and practice. The need to identify different types of innovation in research should be highlighted, as each type might require different resources in order to affect employees.

Therefore, the current study exclusively examined the influence on employee process innovation towards refinements and increasing the efficiency of existing practices and processes in the daily work. It also examined the influence of internal communication and external communication separately.

2.2.3.5 Hypotheses development: Internal communication

Internal communication is expected to be positively associated with employee process innovation for the following reasons. Some theorists have proposed that innovation is largely a social and communicative process (Leonard & Sensiper 1998). From such a social perspective on communication for innovation, interactions with colleagues might consequently determine employees' behaviour and innovative performance (Hirst, Van Dick & Van Knippenberg 2009; Hirst, Van Knippenberg & Zhou 2009). Such interactions between individuals might constitute an important aspect of innovation activities, potentially influencing the emergence of ideas as well as their further refinement and implementation (Hemphälä & Magnusson 2012). As a result, employees with internal communication might be more innovative in their daily work processes.

Following Hirst, Van Dick and Van Knippenberg (2009), Hirst, Van Knippenberg and Zhou (2009), Länsisalmi, Peiro and Kivimaki (2000) and Shalley and Perry-Smith (2008), the present study views internal communication as a continuous variable that relates to the frequency of work contacts inside organisations. In this way, internal communication implies the flow of information between employees and others within the organisation, such as work groups.

New empirical evidence is needed for the stage-by-stage effect on employee process innovation. As a result, on the basis of theoretical arguments and existing studies demonstrating links between communication and the stages of the employee innovation process towards incremental refinements in daily work processes, the following hypotheses are included.

H2a: Frequency of internal communication is positively associated with the opportunity exploration of employees.

H2b: Frequency of internal communication is positively associated with the idea generation and experimentation of employees.

H2c: Frequency of internal communication is positively associated with the idea promotion of employees.

2.2.3.6 Hypotheses development: External communication

Following Von Hippel (1988), Leppalahti and Akerblom (1991), we view external communication as a continuous variable that relates to the frequency of work contacts outside the organisation. Such work contacts refers to contacts with customers (Kanter 1988), other firms, consultants, universities and research centres (von Hippel 1988; Leppalahti & Akerblom 1991).

External communication is expected to be positively associated with employee process innovation for a number of reasons. First, it has been widely accepted that communication enables the sharing of information, which is a major source of innovation in general (e.g. Aiken & Hage 1971; Kanter 1982, 1988a; Evan & Black 1967; Tjosvold & McNeely 1988). Previous research (Lawson & Samson 2001) has showed that communication facilitates knowledge sharing by combining a wide variety of experiences, opening dialogue, building on others' ideas and exploring issues that are relevant to innovation. Accordingly, the extant literature (e.g. Kanter 1988; Perry-Smith & Shalley 2003) suggested that the likelihood that an employee has knowledge or can obtain knowledge of different approaches to a given problem increases if he or she has contacts with a diverse group of people at work. For example, Martin and Horne (1995) suggested that employees that have intensive contacts with their customers could use information about customers' experiences to improve themselves. Therefore, as it is tougher to be innovative when one is isolated or surrounded only by people from inside the organisation (De Jong & Den Hartog 2010), employees with external communication tend to be more innovative in their daily work processes.

New empirical evidence is needed for the stage-by-stage effect on employee process innovation. As a result, on the basis of theoretical arguments and existing studies that demonstrate links between communication and the stages of the employee innovation process towards incremental refinements in daily work processes, the following hypotheses are included.

H3a: Frequency of external communication is positively associated with the opportunity exploration of employees.

H3b: Frequency of external communication is positively associated with the idea generation and experimentation of employees.

H3c: Frequency of external communication is positively associated with the idea promotion of employees.

2.2.4 Reward and recognition

2.2.4.1 Introduction

Human resource management (HRM) scholars have long commented on the linkage between human resource practices and innovation (e.g. Chen & Huang 2009; Jiang, Wang & Zhao 2012). For example, Kozlowski (1987) argued that HRM could play a proactive role in facilitating innovation within a changing environment. The role of HRM has been highlighted as an effective facilitator for innovation (Wang & Zang 2005).

Previous studies suggested that if an organisation intends to be innovative, it should adopt practices that encourage its employees to act innovatively (Amabile et al. 1996). In this respect, organisations should provide an environment where employees feel motivated and committed to knowledge sharing and innovating (Velasco, Zamanillo & Del Valle 2012). Thus, it suggests that human resource practices aim to increase employee incentives to engage in innovative activities (Rammer, Czarnitzki & Spielkamp 2009; Shipton et al. 2006).

In particular, Jiang, Wang and Zhao (2012) found that when employees perceive that the organisation values them by sharing profits (incentive rewards), they reciprocate by making more effort, being more willing to provide suggestions and experimenting with new ways of doing their jobs. This implies that the reward system might play a role in motivating and encouraging employees to innovate at work. Along these lines, in addition to the effect of the reward system on organisational innovative performances proposed by the existing work (e.g. Fan, Hong & Ruan 2011), extant literature (e.g. Shalley, Zhou & Oldham 2004) has also stressed the link between reward system–employee creativity and innovation. For example, it suggests that extrinsic incentives such as monetary compensation could motivate the innovative behaviour of employees (Winston & Baker 1985; Edwards 1989).

The following section reviews the literature relating to the effects of the reward system. This is followed by a review of the influence on process innovation in particular. Lastly, this study reviews recent studies that examined the influence on the stages of the employee innovation process.

2.2.4.2 Rewards system and employee innovation

Rewards are a central component of motivation theories (Latham & Huber 1991). According to Cabrera and Bonache (1999), organisational rewards indicate what the organisation values, and they shape employee behaviour. Reward management is a key function in HRM systems, and it plays an essential role in attracting, retaining and motivating employees (Milkovich & Newman 2005).

Specifically, organisational rewards can range from monetary rewards such as increased salary and bonuses to non-monetary recognition such as prizes (Davenport & Prusak 1998; Hargadon 1998). Similarly, rewards can also be categorised into extrinsic and intrinsic rewards. As Aldrich (1999) suggested, extrinsic rewards refers to tangible rewards that are often money-related. Intrinsic rewards refer to the feelings employees obtain from performing or mastering a task. The terms _monetary reward' and _non-monetary recognition' are used interchangeably in this study with extrinsic and intrinsic rewards, given that monetary is related to money and tangible items, and non-monetary refers to intangible items.

The effect of a reward system consisting of monetary incentives and recognition of employee creativity has been researched (e.g. Amabile et al. 1996; Eisenberger 1992). Scholars have proposed several explanations for a positive link between

reward systems and innovative behaviour by employees. They argued that such rewards encourage creativity because they have informational value and recognise employees' personal competencies (Eisenberger 1992; Eisenberger & Armeli 1997). Gupta and Singhal (1993) argued that financial rewards provide employees with the freedom and support, and appropriate financial rewards are essential to boost creative employees. Similarity, Jiang, Wang and Zhao (2012) suggested that rewards influence both the ability and the motivation of employees to be innovative.

Additionally, extant literature indicates that it is reasonable to assume that the influence of reward systems on stages of the employee innovation process might differ from each other. In support, Amabile (1988) suggested that monetary reward might be used to trigger the effective application behaviour of employees, as non-monetary reward is not a prerequisite for the effective implementation of innovation. Similarly, Lu and Zhang (2007) found that motivation related to an individual's engagement with the task positively affects employee idea generation and idea implementation, whereas motivation related to external factors such as compensation only has a positive effect on innovative idea implementation. They proposed that management should focus on the use of monetary reward and non-monetary recognition to encourage idea implementation.

As discussed, there is a research knowledge gap related to employee innovation and reward systems. Whether and how rewards contribute towards more employee innovative behaviour has not been sufficiently investigated, nor has the kind of reward been fully researched; that is, monetary reward or non-monetary recognition that employees find motivating with regard to being involved in innovative behaviour stage by stage.

In response to the above limitations, as Fernandez and Pitts (2011) suggested, future research should determine how intrinsic and extrinsic rewards could be used separately or in tandem to influence innovative behaviour. That is, one direction for future research might be to make distinctions between monetary reward and non-monetary recognition in order to study their influence on employee innovation. Additionally, as there is a possibility that the influence on the employee innovative process might differ by stages, research is still needed to more closely examine the influence on different stages of the employee innovative process.

Therefore, this study exclusively examines the influence of monetary reward and non-monetary recognition on the employee innovation process towards incremental refinements and increasing the efficiency of existing practices and processes in the daily work. In addition, it attempts to fill the gap in the current research by studying the influences on the different stages of the employee innovation process.

2.2.4.3 Hypotheses development

Reward refers to all tangible benefits and provisions that an employee obtains as a part of his or her employment relationship (Milkovich & Newman 2005). In accordance with Gross and Friedman (2004), we view monetary reward as a continuous variable that refers to financial benefits received from the organisation, consisting of compensation (including base pay and short- and long-term incentives) and benefits (including health and insurance, retirement and work/life). Meanwhile, following Anonymous (2009), we view non-monetary recognition as a continuous variable as intangible things such as non-financial recognition, including praise, awards (Dul & Ceylan 2010) and promotion. Non-monetary recognition is believed to give an employee a certain status within an organisation (Danish & Usman 2010).

First, monetary reward is expected to be positively associated with employee process innovation for a number of reasons. Several scholars have proposed that employee behaviour is driveable and changeable, and that monetary reward can bring expected behaviour and performance (e.g. Maltzman 1960; Pryor, Haag & O'Reilly 1969). Empirical studies provide support for the monetary reward–creativity and innovative behaviour link, and monetary reward has been viewed as necessary to motivate employee creativity (Amabile & Gryskiewicz 1989; Eisenberger, Armeli & Pretz 1998; Eisenberger, Haskins & Gambleton 1999; Eisenberger, Pierce & Cameron 1999). Rewarding divergent thinking leads to

higher levels of creativity (Winston & Baker 1985; Edwards 1989). Higher financial compensation correlates positively with increased employee innovative behaviour (Eisenberger & Cameron 1996; Eisenberger & Armeli 1997). Consequently, employees who are rewarded for their ability and courage to think differently about a problem tend to be more innovative in their daily work processes.

Second, non-monetary recognition is expected to be positively associated with employee process innovation for several reasons. Scholars have suggested that intrinsic motivations can be seen as effective instruments to improve creativity and innovative performance (Deci & Ryan 1985; Amabile 1997; Hennessey & Amabile 1998b). Recent studies have proposed that non-monetary recognition might play a role in enhancing employee creativity (e.g. Amabile 1997; Baer & Frese 2003; Martins & Terblanche 2003). In this respect, management could facilitate employee self-motivation as being beneficial for innovative behaviour by recognising them with praise (Ryan & Deci 2000), which might result in employees being more innovative in their daily work processes.

In addition, this study would take a further step to investigate the effect on stageby-stage employee process innovation for a number of reasons. Extant literature shows that reward systems have been commonly cited as an organisational antecedent that might affect employees' innovative performance (Eisenberger & Cameron 1996; Janssen 2000; Mumford 2000). It is believed that efficient reward systems can be good motivators for employees' performance and behaviour. For example, Bowen and Ostroff (2004) suggested that a good reward system drives performance by motivating employees to achieve new levels of performance. Yusuf, Gunasekaran and Dan (2007) also found that reward and recognition are effective motivators and stimulators for desired performance. Some theorists argue that it is important to distinguish between the suggestion and implementation of ideas when considering innovation (e.g. Unsworth 1999; Unsworth & West 1998). In this respect, Axtell et al. (2000) suggested that factors might have different effects at different stages of innovation. This shows that future empirical research could examine whether reward systems that consist of monetary reward and non-monetary recognition might have differing influences

32

on different stages of the employee innovation process. In addition, Axtell et al. (2000) proposed that reward structures that recognise employees when they make suggestions and have them implemented are likely to help the suggestion stage of innovation.

Finally, we make a distinction between the effect of monetary reward and nonmonetary recognition because, as discussed, the influences on employee innovative behaviour can motivate employees to innovate by providing incentives (e.g. Gupta & Singhal 1993) and encouraging employees to be innovative via enhancing personal feelings such as satisfaction and accomplishments (Danish & Usman 2010; Deci 1996). Extant literature suggests that such influences might differ. For example, Block and MacMillan (1993) proposed that non-financial incentives might be more important than financial incentives. Kanter (1984) also suggested that employees are most satisfied when they receive recognition from their superiors. Hafiza, Shah and Jamsheed (2011) found a positive relationship between financial reward and employee motivation, whereas non-financial rewards such as recognition, appreciation and empowerment had a weaker effect on employee motivation.

Consequently, drawing from earlier studies and intuitive thinking, this study proposes that monetary reward and non-monetary recognition are positively associated with employee process innovation. This study expects that if employees receive monetary reward and non-monetary recognition, they are more likely to develop a greater willingness to generate, practice and suggest new ideas for incremental improvements in their daily work processes. Based on the above discussion, the following hypotheses are included.

H4a: Monetary reward is positively associated with the opportunity exploration of employees.

H4b: Monetary reward is positively associated with the idea generation and experimentation of employees.

H4c: Monetary reward is positively associated with the idea promotion of employees.

H5a: Non-monetary recognition is positively associated with the opportunity exploration of employees.

H5b: Non-monetary recognition is positively associated with the idea generation and experimentation of employees.

H5c: Non-monetary recognition is positively associated with the idea promotion of employees.

2.2.5 Training

2.2.5.1 Introduction

Innovation should be viewed as a complex process that involves a set of investment possibilities. In this investment perspective, knowledge must be considered a sort of capital (Carneiro 2000). Prior work (Harari 1994; Nonaka 1994) suggested that organisations that are able to stimulate and improve the knowledge of their human capital are more prepared to face today's rapid changes and to innovate in the domain where they decide to invest and compete. Most innovative firms provide their staff with more training than less innovative firms (Laursen & Foss 2003; Freel 2005). Theorists have proposed a broad application of training in order to develop the employee skills and knowledge needed for innovation (Beatty & Schneier 1997; Cascio 1990; Mabey & Salaman 1995; Schuler & Jackson 1987).

In particular, as Mumford (2000) suggested, employees' innovative work requires the progressive acquisition of skills and expertise. In this way, training that involves providing employees with the basic knowledge and skills they need to perform their duties (Costen & Salazar 2011) is believed to be instrumental in increasing the knowledge and competence of employees (Johannessen & Olsen 2003). It can enhance employees' knowledge and skills, which are important to increase innovative thought processes and provide opportunities that enhance task-domain expertise (Lau & Ngo 2004).

The following section reviews the literature in relation to the general nature of training theory for employee innovation. This is followed by a review of the literature on process innovation in particular. Finally, the study examines the influence of training according to the stages of the employee innovation process.

2.2.5.2 Training and employee innovation

Recent research shows that training is related to employee creativity. Researchers have argued that employee creativity is a function of knowledge acquisition and motivation (Lawler 1994). In this respect, extant literature suggests that one approach for organisations to enhance employee creativity and innovation is training in creative thinking techniques (Bottger & Yetton 1987; Mansfield, Busse & Krepelka 1978). Recent studies (e.g. Hennessey & Amabile 1998; Shipton et al. 2006) suggested that providing training resources and comprehensive learning opportunities might enhance employee creativity. For example, Jiang, Wang and Zhao (2012) proposed that training in divergent thinking and creative problemsolving skills would facilitate employee creativity.

In addition to enhancing creativity by improving the general creative thinking abilities of employees (Nickerson 1999; Rose & Lin 1984; Scott, Leritz & Mumford 2004), some theorists in the field of individual creativity highlighted the importance of effective features such as motivation to innovate from training (e.g. Amabile 1988; Sternberg & Lubart 1996). In this respect, training employees is believed to bring positive attitudinal improvements (Basadur, Wakabayashi & Graen 1990). The literature on organisational commitment and HR theory points out that providing training might facilitate creative positive employee attitudes and commitment (Benson, Finegold & Mohrman 2004). Nickerson (1999) argued that such attitudinal improvements would be more important than domain-specific knowledge of creativity-enhancing techniques.

While the support for the training–employee creativity link has been provided in terms of cognitive and attitudinal improvements (Birdi 2007), and studies were conducted to examine the influence of training on idea generation (e.g. Licuanan, Dailey & Mumford 2007), other researchers took a broader perspective by focusing on the influence on the idea implementation stage of innovation. For example, training is believed to expose employees to a broader palette of ideas that can be brought to bear on problems requiring novel solutions (Damanpour 1991; Thompson 1965; Fernandez & Pitts 2011; Katz & Tushman 1981). Similarity, as Basadur (2004) suggested, while training has been shown to make people more willing to actively diverge to generate more and better ideas for solving problems, it also improves skills in evaluating ideas and improves the ability to separate idea generation from evaluation through deferral of judgment.

However, it is worth noting that, as Birdi (2007) suggested, the issue of the differing influence of training on idea generation versus idea implementation requires future exploration. Given that it is possible that training might be limited to mainly facilitating employees to generate ideas, whether those ideas are put into practice might be more relevant to other factors that are not under the influence of employees (Van de Ven, Angle & Poole 1989). Birdi (2007) argued that there is little empirical evaluation of training for the management of innovation. This implies that a more comprehensive perspective is needed to investigate the influence of training on different stages of the employee innovation process.

2.2.5.3 Training and process innovation

Carneiro (2000) proposed that knowledge development is a fruitful background where incremental innovation might be attempted. Similarly, Schmierl and Köhler (2005) suggested that training is especially important in a non-R&D intensive environment because daily work requires hybrid qualifications that are not usually offered by the market. This implies that training might play a role in influencing employee innovative behaviour towards incremental innovation concerning daily work processes. In support, Beugelsdijk (2008) suggested that training is positively associated with incremental innovation.

However, it finds that few researchers explicitly specify the focus on process innovation towards incremental improvements in everyday working processes. Most studies are limited to knowledge employees, such as in R&D functions (e.g. Wang & Horng 2002), and address radical innovation in products (e.g. O'Connor, Paulson & DeMartino 2008).

Consequently, in accordance with previous studies (e.g. Birdi 2007) and taking a further step, the present study exclusively examines the effect of training on the employee innovation process towards incremental improvements in daily work processes from idea generation to putting them into practice.

2.2.5.4 Hypotheses development

Training is used to provide employees with the knowledge and skills needed for their present job (Fitzgerald 1992). This study expects training to be positively associated with employee process innovation for several reasons. Several theorists have suggested that the nature of training appears to be compatible with employee innovative behaviour because the focus of training courses tends to be on generating and choosing solutions using creative thinking techniques as well as opportunity- or problem-finding and the implementation of ideas (Basadur, Graen & Green 1982; Leavitt 1975; Wang & Horng 2002). Previous research (Shalley & Gilson 2004) revealed that employees try to be more innovative in their work if they are offered training opportunities that can increase their knowledge base or creativity-relevant skills. This implies that by offering training opportunities, employees' work might benefit from an increased knowledge base and skills. In addition, by offering training opportunities, employee motivation to innovate at work is enhanced (e.g. Basadur, Graen & Green 1982). Therefore, employees tend to be more innovative in their day-to-day work processes.

In addition, this study would more closely examine the influence on stage-bystage employee process innovation for a number of reasons. Following Amabile (1988), innovation is considered to not only cover the generation of ideas that are novel and useful to the workplace, but also to include their implementation in ways of working. In this respect, Birdi (2007) suggested that the skills needed for idea generation are likely to be different to those required for idea implementation (e.g. planning). This implies that future research could examine whether the effect of training on the employee innovation process differs between different stages. This research direction is consistent with the theoretical view that factors that influence the generation of ideas differs from those that influence their implementation (Axtell et al. 2000). This perspective can also be informed by findings from the general training literature, which shows that the application of trained skills at work is influenced by external factors such as the opportunity to use skills and the degree of social support (Tannenbaum & Yukl 1992; Warr, Allan & Birdi 1999). Consequently, another aim of this study is to investigate the influence on everyday innovation with respect to different stages. Birdi (2007) found that creativity training is only slightly more strongly related to the generation of ideas as opposed to their implementation.

Given that empirical studies that explicitly examined the effect of training on the process innovation type at the employee level are still scarce, based on the theoretical arguments and previous studies that demonstrate the link between training and employee creativity and innovation, the present study includes the following hypotheses.

H6a: Training is positively associated with the opportunity exploration of employees.

H6b: Training is positively associated with the idea generation and experimentation of employees.

H6c: Training is positively associated with the idea promotion of employees.

2.3 Chapter summary

This chapter has presented a review of the literature relevant to developing a theoretical framework for understanding the relationships between employee innovation and six organisation-related factors. The proposed framework

emphasises stages of employee innovation concerning incremental improvements of processes—that is, from idea generation and idea implementation to idea promotion at the employee level—as the key aspects for examining the influences of corporate-level factors on those stages. While the current body of research regarding the influences of the six corporate-level factors on employee innovative behaviour has been well identified, the current literature has paid little attention to their influences on the stages of employee innovative behaviour towards process innovation, thereby leaving the door open for future investigation.

Chapter 3: Research Methodology

This chapter presents the methodology for the research. It begins with a description and justification of the chosen method. This is followed by a discussion of the instrument development, pre-test, sampling method, data collection and data analysis techniques. The quality of the data is also discussed in this chapter. Finally, the chapter concludes with a summary.

3.1 Research methodology

When designing the research method, the nature of the research questions and objective of the study should be considered because particular research methodologies are more suited to particular research problems (Babbie 2005). This study attempted to expand the understanding of the links between corporate-related factors and employee innovation by emphasising the importance of employee innovative behaviour on incremental innovation (e.g. Axtell et al. 2000; Scott & Bruce 1994), which relates more to minor changes/modifications in daily work processes. Therefore, this research is theory extension rather than theory generation. It follows a hypothetical–deductive approach (Burrell & Morgan 1979; Deshpande 1983), where deductive refers to research that starts with a hypothesis and designs a study to systematically test them (Carpiano & Daley 2006).

Due to the theory extension nature of this study, a positivist approach is deemed appropriate (Burrell & Morgan 1979; Deshpande 1983). A positivist approach relies on a reductionist view in its search for universal mechanistic rules that are not contextually bounded, and it seeks to verify hypotheses (Phillips & Burbules 2000). That is, positivism is associated with deductive theory testing (Burrell & Morgan 1979; Deshpande 1983), which is an approach that researchers use to reach generalisable conclusions (Creswell 1994; Deshpande 1983). Using the deductive method, the researcher begins with hypotheses development and follows with observations that are relevant to testing the hypotheses. Finally, the researcher decides whether to accept or reject the hypotheses by comparing the observations and hypotheses (Babbie 2005).

Accordingly, a quantitative research methodology was employed because it relies on numerical data to test hypotheses. The quantitative methodology involves data collection and the analysis of numerical data (Hesse-Biber & Leavy 2006; Veal & Ticehurst 2005). Quantitative research allows for the collection of answers to a number of questions from a large sample of the population. It allows the researcher to collect and explore information that describes the experience being studied, and it helps prevent bias in collecting and presenting data; for example, a questionnaire survey might avoid interview bias and assure anonymity (Neuman 2003; Sarantakos 2005); therefore, quantitative research could avoid subjectivity (Sarantakos 2005).

A questionnaire survey was the most appropriate way to obtain the necessary data for the present study to test the proposed hypotheses. It is evident that a survey questionnaire, when constructed and used properly, is a powerful scientific instrument for collecting data (Borg & Gall 1989; Shaughnessy & Zechmeister 1997). A survey provides an opportunity to contact a large audience with moderate cost in time and money (Van Riel et al. 1998). Similarly, Babbie (2005) suggested that surveys—especially self-administered ones—could make large samples feasible. A questionnaire is a powerful scientific instrument for measuring different variables (Shaugnessy & Zechmeister 1997). In addition, surveys are flexible because they allow researchers to ask many questions on a given topic, thus giving them considerable flexibility in the analysis (Babbie 2005).

In particular, a self-administered questionnaire could provide greater accessibility to a relatively large number of participants in a wide geographic area (Neuman 2003; Hoyle, Harris & Judd 2002). The use of a structured, self-rated questionnaire can help to obtain perceptual data collected confidentially and inexpensively (Shaughnessy et al. 2000). Moreover, according to Shaughnessy et al. (2000), survey research can be used to measure the nature of people's thoughts, opinions and feelings. As this thesis is interested in the personal behaviour of employees and the influences of organisation-related factors from the employee's perspective, a self-administered survey was employed in this study. The researcher chose to survey individuals about their own innovative behaviour because this behaviour could be unknown to the supervisors or other colleagues. In addition, their innovative behaviour could be considered threatening by supervisors and colleagues (Thornberry 2003). Therefore, it could be hard to obtain true views about employees' innovative behaviour from others' evaluations.

Therefore, the researcher collected data using a self-administered questionnaire survey conducted in China within a period of five months between November 2010 and March 2011. The detailed research plan is presented in the following sections.

3.2 Sampling

The sampling method is part of the research plan that indicates how cases are selected for investigation (Singleton & Straits 1999). Sampling methods can be classified as either probability sampling or non-probability sampling (Singleton & Straits 1999). According to Trzesniewski, Donnellan and Robins (2008), convenience sampling means that data are not sampled randomly from the general population. Zikmund (2003) referred to convenience sampling as the process of obtaining information about people who are the most conveniently available. Convenience sampling was used in this study for several reasons. First, in this study, anyone employed in an organisation could participate (i.e. the population of the present study). This qualification would ensure that participants understand the workplace within their organisations, making the survey items easy for them to accomplish. In addition, this qualification would not introduce sampling bias or errors. Given this, as the population is quite large in this study, it is extremely difficult to identify every employee of an organisation, and it is also not necessary for this study to do so.

Specifically, 1,850 questionnaires were sent over a five-month period in China. After checking all returned questionnaires, a final usable sample size of 1,299 was obtained. According to Hair et al. (1998), the sample size suitable for most multivariate analysis approaches should have a minimum ratio of at least five respondents for each estimate item. The present study has 42 variables (excluding variables for demographic characteristics). Therefore, given the final sample size of 1,299, the researcher obtained the ratio of 30 respondents for each item, which was found to be above the minimum sample size for analyses, as the extant literature suggested (Hair et al. 1998).

3.3 Survey instrument

The survey instrument consisted of an information sheet and the survey questionnaire, as illustrated in Appendix A (English Questionnaire) and Appendix B (Chinese Questionnaire). The information sheet describes the purpose of the study, the structure of the questionnaire and other relevant information such as why the participant had been invited, voluntary participation and confidentiality for the potential participants. Below are descriptions of the scales included in the survey instrument.

The questionnaire consisted of five sections. Section 1 aimed to gather background information about the respondents and the organisations. Section 2 was designed to collect respondents' perceptions about their innovative behaviour in the workplace. Section 3 focused on respondents' views about their relationship with their supervisor. Sections 4 and 5 focused on collecting data that showed the condition of internal and external communication at work. Sections 6 and 7 focused on respondents' perceptions about monetary reward and non-monetary recognition that organisations provide. Section 8 aimed to gather data relating to respondents' perceptions about available training.

The survey used measures that were developed based on insights from the extant literature from both published questionnaire items (for measuring employee innovative behaviour and the quality of the supervisor–employee relationship) and the extant literature providing theoretical definitions and domains of the other constructs of interest (for measuring internal and external communication, monetary reward, non-monetary recognition and training). All items were worded positively in the questionnaire. Respondents were asked to rate the items for each construct on a seven-point Likert scale. The following discussion is framed in relation to each construct of interest for the present study.

3.3.1 Employees' innovative behaviour

The present study used the instrument proposed by Kleysen and Street (2001) to measure employee innovative behaviour. Sixteen items were used based on the work of Kleysen and Street (2001) to assess the three stages (opportunity exploration, idea generation and experimenting, and idea promotion) of employee behaviour in process innovations. Items were scored on a seven-point Likert scale that ranged from _1=never' to _7=always'. Among the 16 items, employees provided self-rated assessments on three items for opportunities exploration (i.e. _In your current work, how often do you recognise opportunities to make a positive difference in your work?'); eight items for generating and experimenting ideas (i.e. _In your current work, how often do you experiment with new ideas?'); and five items for idea suggestion (promotion) (i.e. _In your current work, how often do you put ideas forward to supervisor-level people so they have a chance to become implemented at the corporate level?').

Three original items were reworded to fit better with this study's situation and to make it easier for participants to understand. As this study focused on process innovation, the researchers reworded the original item _improve an existing process, technology, product, service or work relationships' to _improve an existing process' (see item 11 of questionnaire). The original item _take the risks to support new ideas' was broadened into three items: _take the risks to support other colleagues' new ideas' and _take the risks to promote own new ideas' to ensure that the construct's domain was adequately covered (see items 25–26). The researchers separated the original item _push ideas forward so that they have a chance to become implemented' into two items to distinguish two levels (i.e. peer level and supervisor level) for where to push ideas forward as _push ideas forward to the colleagues' and _push ideas forward to supervisor-level people'.

In this study, all 16 measurement items were arranged in a logical flow; that is, from idea generation to implementation, followed by suggestion and promotion, which is different from the original sequence. By doing so, the measures are presented in the order consistent with the conceptualisation of employee process innovation in this study. This helped participants to evaluate their behaviour more easily.

3.3.2 Quality of the supervisor-employee relationship

The quality of the supervisor–employee relationship was assessed by a slightly modified version of the seven-item LMX instrument proposed by Graen and Uhl-Bien (1995). LMX-seven is concise and measures three dimensions of leader–member working relationships—respect, trust and obligation—and it is an extensively pre-tested instrument that has proven to be valid and reliable (Graen & Uhi-Bien 1995).

In this study, the researcher used the same items proposed by Graen and Uhi-Bien (1995) but worded affirmatively (i.e. original items were adapted from questions such as _Do you know where you stand with you leader' into statements such as _You know where you stand with your supervisor'). Using a seven-point scale, respondents rated the extent of their agreement with statements about the quality of their relationship with their supervisors. The scale ranges from _1=totally disagree' to _7=totally agree'. Among the seven items, the researchers separated one original item into two to ensure that each item only included one (items 28–29 of questionnaire). Therefore, an eight-item scale based on LMX-seven was used to measure this construct (see Appendix A).

3.3.3 Internal communication at work

This variable was measured with four items that were based on the extant literature, which provides theoretical definitions and domains of this construct (see Appendix A). These four items (item 39–42) asked respondents to rate their frequency of communication at work with employees from the same and different work areas with the same and different job tasks. Participants responded to these

four items using a seven-point Likert-type scale ranging from _1=not at all' to _7=yearly'.

3.3.4 External communication at work¹

This variable was measured with four items that were derived from the existing literature (Von Hippel 1988; Leppalahti & Akerblom 1991) (see Appendix B). These four items asked respondents to rate their frequency of external contacts at work, including with conferences, customers, other companies and institutions. Participants responded to these four items using a seven-point Likert-type scale ranging from _1=not at all' to _7=yearly'.

3.3.5 Monetary reward

As there is no existing scale for this construct of interest for this study, this variable was measured with four items that were generated for the purpose of this study (i.e. _How often do you receive the following rewards: increased wages; bonuses; various allowances; social wage?'). Participants responded to these five items using a seven-point Likert-type scale ranging from _1= never' to _7= always'.

3.3.6 Non-monetary recognition

Based on the current literature (e.g. Bartol 2002; Danish & Usman 2010), the quality of non-monetary recognition is defined as an employee's self-rating of views on which he or she thinks about non-monetary recognition in terms of the _promotion', _honour', _symbolic support' that he or she can receive from the current organisation.

¹ There were 35 participants who did not use any external communication mechanisms, and that this represents such small proportion of the sample (1,299 in total) that it does not affect the results or conclusions.

As there is no existing scale for this construct of interest for this study, this variable was measured with three items that were generated for the purpose of this study (i.e. _How often do you receive the following recognition: promotion; honour; symbolic support?'). Participants responded to these five items using a seven-point Likert-type scale ranging from _1=never' to _7=always'.

3.3.7 Training

Using a seven-point scale, participants rated the extent of their agreement with statements about the quality of training at work ranging from _1=totally disagree' to _7=totally agree'.

3.3.8 Control variables

The demographic characteristics were ascertained for each participant with direct and straightforward questions included on the questionnaire in the first section. The researcher collected the demographic data such as gender, age and education level because the researcher planned to check for differences in the employee innovative behaviour to socio-demographic variables. As such, it is expected to gain in-depth understanding based on the collected data. By collecting information on important demographic characteristics, the survey sample in the present study can be accurately described.

The control variables fall into two different categories. The first category consists of the demographic characteristics of the respondent, including gender and education. The second category includes measures of the organisation size and job characteristics.

The current study used gender (1=female, 2=male) and the level of education (1=below undergraduate, 2=undergraduate and above) as control variables to check for differences in employee innovative behaviour related to sociodemographic variables. Kanter (1984) suggested that under certain circumstances women would behave more innovatively than men. Pinchot (1985) claimed that educational level could affect innovative behaviour by employees. Moreover, the educational level of an employee has been suggested to be important to innovation (Mumford & Gustafson 1988; Tierney & Farmer 2002), and it might affect the ability of an employee to identify and generate new ideas.

In addition, Pinchot (1985) suggested that work experience from technicaloriented work would improve the likelihood of innovative behaviour by employees. Moreover, innovative behaviour by employees may be more common among employees who hold jobs in certain functional areas (Pinchot 1985). Therefore, because work area might influence employee innovative behaviour, work area (1=specialised technical relevant, 2=less specialised functional) was included as a control. Further, according to Pinchot (1985), job rotation will improve the innovative behaviour of the employees. Lee and Wong (2004) also argued that organisational tenure is positively related to innovation behaviour. Thus, the length of service time (1= <= five years, 2= >five years) was included as a control. Finally, according to Kanter (1984), large organisations offer more opportunities for learning experiences, which could be implemented in another functional or technical arena. Thus, organisation size (1=medium and small, 2=large) was also used as a control variable.

3.4 Common method variance (CMV)

In typical survey studies in which the same rater responds to the items in a single questionnaire at the same point in time, data are likely to be susceptible to CMV (Kemery & Dunlap 1986; Lindell & Whitney 2001). As described by Fiske (1982), CMV refers to the variance that is attributable to the measurement method rather than to the constructs that the measures represent. Such a variance may occur as a result of factors such as social desirability, the halo effect and selective memory brought about by the self-reporting method, and it can threaten the internal validity of conclusions about the predictive relationships between measures (Campbell & Fiske 1959; Howard 1994; Spector 1994; Podsakoff et al. 2003). Using a single source to measure variables—usually self-rated surveys—may inflate correlations among variables. As suggested by Kaynak (1997), a researcher should therefore plan how to overcome CMV.

In this study, as all data were self-rated and collected through the same questionnaire during the same period with cross-sectional research design, CMV that is attributed to the measurement method rather than the constructs of interest may cause systematic measurement error and thus further bias the estimates of the true relationship among theoretical constructs (Avolio, Yammarino & Bass 1991; Podsakoff et al. 2003). Therefore, CMV may cause concern. The present study thereby utilised other techniques that could be helpful in controlling for CMV, as suggested by Podsakoff et al. (2003). The assessment of CMV is presented in this section.

Harman's Single Factor test (Podsakoff et al. 2003) was performed to examine whether CMV was a problem in the current study. This statistical method is one of the most widely used techniques to address the issue of CMV (Podsakoff & Organ 1986; Podsakoff et al. 2003). Researchers using this technique traditionally load all variables in their research into an exploratory factor analysis (EFA) and examine the unrotated factor solution to determine the number of factors that are necessary to account for variance in the variables (e.g. Andersson & Bateman 1997; Aulakh & Gencturk 2000; Organ & Greene 1981; Schriesheim 1979). If a substantial amount of CMV is presented, either a single factor will emerge from the factor analysis or one general factor will account for the majority of the covariance among the variables (e.g. Anderson & Bateman 1997; Aulakh & Gencturk 2003).

The present study used Harman's one-factor test (Schriesheim 1979) to address the common method bias issue. A principal components factor analysis on items in the nine variable measures was performed to examine if common method bias was a serious problem in this study. The factor analysis extracted nine factors with an eigenvalue greater than 1.0 (together accounting for 71.61% of the total variance), and the first (largest) factor accounted for only 15.92% of the variance. The results of this test indicate that CMV was not a problem in this study. Not a single factor, nor one general factor, emerged from the factor analysis to account for most of the covariance in the dependent and independent variables (Podsakoff & Organ 1986); thus, it is unlikely to confound the interpretations of results.

3.5 Reliability and validity

It must be noted that a central issue in research using survey questionnaires is internal reliability, referring to the extent to which all items within one scale capture the same construct (Hair et al. 1998). A common indicator of internal reliability is the Cronbach's alpha coefficient (α), which ideally should be greater than 0.70 to show an acceptable internal reliability level (Nunally & Bernstein 1994; Pallant 2007). The higher the value of Cronbach's alpha coefficient (α), the higher the internal consistency of the item measurement of each construct, thus a high internal reliability of the survey instrument (Hair et al. 1998). Table 3.1 outlines the Cronbach's alpha coefficient (α) values for each construct in this study. The results suggest that the reliability of measures in this study is satisfactory (alphas from .701 to .933).

Construct	Cronbach's alpha	
Opportunities exploration	0.721	
Generating and experimenting ideas	0.933	
Idea promotion	0.886	
Quality of the supervisor–employee relationship	0.869	
Internal communication	0.842	
External communication	0.701	
Monetary reward	0.811	
Non-monetary recognition	0.827	
Training	0.920	

Table 3.1: Summary of results of Cronbach's alpha test

Validity refers to the extent to which a measurement procedure truly measures what it is intended to measure (Peter 1979; O'Leary 2004). The present study employed construct validity checks for the measurement items, which is used to examine the extent to which a scale measures a theoretical variable of interest (Woo Bock & Kim 2002). In particular, construct validity is established by convergent validity and discriminate validity (Straub 1989; Chi 2005), convergent validity is concerned with the extent to which multiple attempts to measure the same concept with different methods are in agreement, and discriminate validity is

the extent to which a concept differs from other concepts (Fornell & Larcker 1981; Hair et al. 1998).

Correlations between the latent constructs in factor analyses can be checked to establish convergent and discriminate validity (Hair et al. 1998). In this study, EFA, which concerns relationships among variables, was used to test the convergent validity of measurement scales (Hair et al. 1998). Specifically, the researcher evaluated the item-to-total correlation; that is, the correlation of each item to the sum of the remaining items. The scores for each single item within an instrument may be correlated with scores on the total test (Gregory 1996). The rationale for this check is that an instrument with a high level of internal consistency would consist of items that are reasonably homogeneous and that demonstrate high item-total correlations. Everitt (2002) and Field (2005) suggested that a correlation value of less than .20 or .30 indicates that the corresponding item does not correlate very well with the scale overall, and it may be dropped.

The researcher also assessed the mean inter-item correlation. A small item correlation provides empirical evidence that the item is not measuring the same construct by the other items included. In this study, the item-total correlation test was performed to check if any item in the sets was inconsistent with the average behaviour of the others. The recommended critical values are .30 for the mean inter-item correlation and .20 for any item-total correlation (Cortina 1993; Nunnally 1967). Table 3.2 shows the results of the convergent validity test, which suggest that no mean inter-item correlations in this study were lower than .30 and no item-to-total correlations were lower than .20, indicating that they achieved the acceptable level for the assessment of convergent validity.

	Mean inter-	Range item-
Construct	item	total
	correlation	correlations
Opportunities exploration	0.472	0.451-0.610
Generating and experimenting ideas	0.634	0.669–0.820
Idea promotion	0.605	0.588-0.807
Quality of the supervisor–employee relationship	0.456	0.480-0.751
Internal communication	0.584	0.605–0.783
External communication	0.373	0.381-0.586
Monetary reward	0.533	0.571-0.715
Non-monetary recognition	0.613	0.579–0.776
Training	0.748	0.766–0.889

Table 3.2: Summary of results of validity test

Discriminate validity was assessed using the factor loading values. The validity was supported by the Principal Component Analysis (PCA), in which the value of the factor loading of each item into its relative principal component should not be less than .40 (Straub, Boudreau & Gefen 2004). The results of the PCA indicate that all of the loadings were above .50, which means that the acceptable level of standardised loading estimates for the assessment of validity was achieved. Detailed results computed via PCA are presented in Chapter 4. Moreover, following Kline (2005), correlations between constructs should not exceed .85 in order for the constructs to have discriminate validity. The correlation results reveal that correlations between constructs in this study were below .85. In summary, the reliability and validity of the variables were found to be acceptable.

3.6 Pre-test

According to the current literature (Davis & Cosenza 1985; van der Velde, Jansen & Anderson 2004), once a survey instrument is completed in draft form, it should be subject to a pre-test in order to ensure readability and detect any potential ambiguity regarding the wording and format of survey items. This procedure is expected to help researchers to minimise errors due to unclear wording or improper design (Schwab 2005; Zikmund 2003). Therefore, before collecting data, the questionnaire was pre-tested in order to discover ambiguous questions. The basic goals of the pre-test conducted for the current research include evaluating

the competency of the questionnaire and estimating the length of the survey (Larossi 2006).

The researcher asked two academics in the School of Management of the University of Tasmania to screen the questionnaire for problems with question wording and bias due to the question sequence. Based on their feedback, minor changes to the wording of a few questions were made to ensure that the questionnaire was easy to understand. The questionnaire was then pre-tested on 10 employees from different work functional areas in the service industry in the cities of Tai'an and Shanghai, China, to ensure that the questions were clear (their responses were not used in the final study). Feedback from the pre-test showed that the questions were clearly and easily understood and that there was no need to further revise the questionnaire. The format was logical and clear, and the questionnaire could be completed within 15 minutes, which is a suitable time limit.

3.7 Ethical considerations

This study was conducted with the approval of the University of Tasmania's Human Ethics Research Committee (approval number: H11495). To fulfil the considerations of ethical approval, the questionnaire included an information sheet to invite voluntary participation, to explain the purpose of the study, and to address issues relating to anonymity and confidentiality. This ensured that respondents knew that all answers to questions in this questionnaire would remain anonymous and all information would be treated confidentially; there were no right or wrong answers. It also advised them how to complete and return the survey.

3.8 Language for the survey

As the current research was conducted in a Chinese-speaking context, the questionnaire was translated from English into simplified Chinese following back-translation procedures (note that simplified Chinese and traditional Chinese are different variations of the written Chinese language, where simplified Chinese is

the official writing system in mainland China). This procedure has been used to adapt a large number of English measures into other languages (Mallinckrodt & Wang 2004), which can improve the research's validity in different languages (Sartorius & Kuyken 1994; Sekaran 2003). The researcher, a native Chinese speaker, first designed the survey questions in English and then translated the survey from English to simplified Chinese. Following that, the Chinese version of the questionnaire was then translated back into English by a Chinese bilingual person with a Ph.D. Finally, the original and back-translated items were compared to see if differences in meaning existed. No differences in meanings were found between the original version and the back-translated version. Therefore, the simplified Chinese version of the questionnaire was used in the survey.

3.9 Data collection

As presented earlier, data for this thesis were gathered through a questionnaire survey conducted in China. The detailed data gathering procedure is presented in this section. The researcher followed a two-step procedure.

Using relationship networks, the researcher initially contacted existing contacts from a variety of organisations in China to solicit their participation in this research. The researcher then passed on hard copies of the Chinese (simplified) version of the questionnaire to the contacts, who then physically distributed hard copies of the questionnaire to their contacts in a range of organisations.

Next, the questionnaire was distributed to employees through the contacts. All participants were informed in the information sheet that participation was voluntary. They were told that their individual responses would remain confidential and that the data would only be used for statistics. They were informed that their consent to participate in this study would be implied by the return of the completed questionnaire through the contacts that distributed the questionnaire. They were informed that if they wished to be involved in this study, they needed to complete the questionnaire, place it into the envelope provided and return it to the person in charge of distributing and collecting the questionnaires.

Then all received envelops from employees would be placed into one envelop provided for the distributor and then returned to the researcher. This provided an opportunity to discuss with some of the contacts how the process of distribution had been completed. Additionally, informal feedback was also obtained on how the respondents had reacted. No significant problems were reported.

3.10 Sample profile

Frequency tables were generated to provide descriptive statistics about the profile of the sample in the present study. To understand the characteristics of each construct, descriptive statistical analysis was employed to illustrate frequencies, means and standard deviations of every construct.

The results of descriptive statistics for demographic information collected from the respondents are presented in this section, including gender, highest education level, the work area of the participant, the length of time the participant has been employed in the current organisation, the industry of the participant's organisation, the ownership type of the participant's organisation, the number of employees in the participant's organisation and the organisation's years of operation. They were conducted to provide simple summaries about the sample and the measures for this thesis.

Of the 1,299 completed questionnaires, 42.1% were male and 57.9% were female. The age of the participants ranged from 18 to 55. Most of the participants (61.4%) had bachelor degrees (61.4%), very few completed high school and below (1.2%) and 6.9% had a Master's degree or above. The participants were spread across nine different work areas, including 23.7% in sales, 20.9% in general administration, 12.4% in technical areas, 11.5% in advertising and marketing, 8.9% in human resources, 7.2% in finance and accounting, 7.0% in R&D, 6.1% in production and operation, and the remainder (2.4%) in _other' work areas. Regarding the respondents' length of service in years, the majority of the respondents had 1–3 years' service in their current organisation (40.3%). Only 1.8% had 10–20 years' service and 8.6% had less than three months' service.

In regards to the industrial classification of organisations, the majority of participating organisations were from the services industry (62.1%). The other three industries included infrastructure (19.3%), manufacturing (14.6%) and energy (3.8%). In addition, the ownership types of organisations included in the sample were Limited Liability Corporation (32.9%), state-owned enterprise (27.6%), private enterprise (20.6%), Sino-foreign cooperative enterprise (8.2%), wholly foreign or Hong Kong (HK), Taiwan or Macao-owned enterprise (7.6%), Sino-foreign or HK, Taiwan or Macao equity joint venture (2.6%) and collective-owned enterprise (.3%).

About 43.6% of the organisations had fewer than 100 employees, while 38.1% had between 101 and 499 employees. About 10.3% of the organisations possessed between 501 and 1,000 employees, and 7.9% possessed >=1001 employees. Moreover, regarding the years that the participating organisation had operated for, about 32.8% of the participating organisations had been in business for 1–5 years. Of the organisations participating, 28.1% had between six and 10 years in business, 18.8% had between 11 and 20 years in business, 18.2% had been in business for less than one year.

The results of the descriptive statistics above provide the demographic characteristics of respondents, showing the diversity of the participants' profiles in this study to have a better understanding of their responses and resulting conclusions for the research.

3.11 Correlation analysis

In order to test the 18 hypotheses for this study, the inferential tests used include the Pearson Product–Moment Correlation Coefficient, Multiple Regression Analysis. The upper level of statistical significance for null hypothesis testing was set at 5%, which is the most widely used level of significance (Hair et al. 1998). All statistical test results were computed at the two-tailed level of significance in accordance with the non-directional hypotheses presented (Sekaran 2001).

The Statistical Package for the Social Sciences (SPSS) Version 18 was used to store and analyse the data. The SPSS was sufficient for the types of tests that were used.

First, as preliminary analysis, principle component analysis with Varimax rotation was performed. PCA could be used to concisely describe and understand the relationships among observed variables (Tabachnick & Fidell 2001) and to explain these variables in terms of their common underlying dimensions (Hair et al. 1998). The Varimax method—an orthogonal rotation method—was used because it is the most widely used rotation for factor loadings (Norusis 1993) regarding its ability to secure a simple structure effectively.

Correlation coefficients indicate both the direction of the relationship and its magnitude (Hameed & Amjad 2011). The Pearson Product-Moment Correlation Coefficient is a statistic that indicates the degree to which variables are related to one another (Hair et al. 1998). The sign of a correlation coefficient (+ or -) shows the direction of the relationship between -1.00 and +1.00. A positive correlation indicates a direct and positive association between two variables (Leary 2004). As the present study aimed to investigate the relationship between the dependent and independent variables, the Pearson Product-Moment Correlation Coefficient was appropriate for hypotheses testing. It was used to test whether the independent variables (quality of the supervisor-employee relationship, frequency of internal communication, frequency of external communication, quality of monetary reward, quality of non-monetary recognition and quality of training) were positively associated with the three dependent variables (opportunity exploration, idea generating and experimenting, and idea promoting of employee process innovative behaviour), respectively. The correlation analysis was employed here to test whether the variables were reliable, and the Pearson Product-Moment Correlation Coefficient r was examined to indicate the strength of the relationship between the independent variable and dependent variable (Hair et al. 1998).

3.12 Multiple regressions

Multiple regressions are employed for hypothesis testing. Given that the correlation coefficient r does not give the magnitude of the variance in that dependent variable that will be explained when several independent variables are theorised to simultaneously affect it (Im 2003), it is worth noting that the dependent variable, for example opportunity exploration, may be explained by a range of independent demographic and other variables. Thereby, multiple regressions are excellent tools to evaluate the relationship between a set of independent variables and a dependent variable (Tabachnick & Fidell 2001). Hair et al. (1998) suggested that multiple regression analysis can be used to analyse the relationship between a single dependent variable and several independent variables.

As the current thesis was interested in the relationship between the dependent variables (opportunity exploration, idea generating and experimenting, and idea promoting of employee process innovative behaviour) and the independent variables (quality of the supervisor–employee relationship, frequency of internal communication, frequency of external communication, quality of monetary reward, quality of non-monetary recognition and quality of training), multiple regressions were appropriate. Specifically, the adjusted R² in the multiple regressions shows how much of the variance in the behaviour is explained by the independent variables. The standardised beta values in multiple regressions indicate the relative strength and direction of the independent variables on the examined behaviour (Åmo & Kolvereid 2005; Hair et al. 1998).

In this study, in order to test if the demographic variables including gender, education level, work area, organisation size and work length affected the variables of interest, a hierarchical regression was employed to examine the relationship between a set of independent variables and the dependent variables, after controlling for, or taking into account, the influence of the control variables on the dependent variable. First, for each of the dependent variables, including opportunity exploration, idea generating and experimenting, and idea promoting, a regression with only control variables was run, including measures describing the individual and organisation. By doing so, the researcher aimed to investigate the influence of the control variables on the relationships proposed between employee process innovative behaviour and corporate factors. The regression then included the measures of corporate-level factors in addition to the control variables, which was run for each dependent variable. Specifically, the regression, controlling for gender, education level, service length, work area and organisation size, was conducted with the six corporate-level factors as the predictors and opportunity exploration, idea generating and experimenting, and idea promoting as the criterion, respectively.

3.13 Chapter summary

This chapter presented the methods used for the analyses, of which the rationale is also provided. A quantitative methodology was deemed suitable for investigating the influences of corporate-level factors on employee process innovation at work. This chapter addressed all major aspects of the research design, measuring instruments, sampling method, data collection and statistical methods that were employed to analyse the data.
Chapter 4: Analysis and Results

This chapter presents the results of the data analysis and hypothesis testing. First, results from the factor analysis are presented. This is followed by the inferential statistical analysis, including correlation and multiple regression analyses for hypotheses testing. The chapter concludes with a summary.

4.1 Principal component analysis

In this study, all observable variables were entered into a factor analysis using PCA with Varimax rotation to determine the number of factors that are necessary to account for the variance in the variables. It was conducted to extract the anticipated factors regarding the employee innovation process and organisational factors. Generally, the components of a construct are considered the principal components if it has an eigenvalue greater than 1.0 and is used for further analysis (Straub, Boudreau & Gefen 2004). Factor loadings were also checked in order to examine how each item loaded into its corresponding factor (Hair et al. 1998). The value of factor loading of each item into its relative principal component should not be less .40 (Straub, Boudreau & Gefen 2004).

Before the extraction process, as indicated in Table 4.1, the Kaiser–Meyer–Olkin (KMO) values ranged from .646 to .930, which were greater than the acceptable level of .50 suggested by Kaiser (1974), supporting the adequacy of the sample for factorisability. The Bartlett's test of sphericity was highly significant, which indicated that the correlation matrix was not an identity, supporting the factor analysis (Stewart 1981).

	KMO Measure	Bartlett's Test of	Bartlett's Test
Constructs	of Sampling	Sphericity Chi-	of Sphericity
	Adequacy	square	Sig.
All items of the constructs	.857	40077.668	.000
Opportunity exploration	.646	869.99	.000
stage			
Generating and	.930	7469.95	.000
experimenting ideas stage			
Promoting ideas stage	.812	3859.78	.000
Quality of the supervisor-	.816	5201.96	.000
employee relationship			
Internal communication	.717	2632.16	.000
External communication	.730	933.04	.000
Monetary reward	.747	1974.50	.000
Non-monetary recognition	.670	1639.12	.000
Training	.779	4474.62	.000

Table 4.1: KMO and Bartlett's Test

The results reveal the presence of nine distinct factors with eigenvalues greater than 1.0. The values of loadings of all items in this study vary from .542 to .903. Based on the nature of the variables, the nine factors extracted were accordingly termed as _Generating and experimenting ideas stage', _Quality of the supervisor–employee', _Non-monetary recognition', _Internal communication', _External communication', _Opportunity exploration stage', _Monetary reward' and _Promoting ideas stage'.

Specifically, most of the question items well loaded on the relative factors as expected. Only one cross-loading item was found: the item relating to _promotion' cross-loaded on the factors of monetary reward and non-monetary recognition (the values of the loadings: .665 and .506). As the researcher aimed to distinguish between monetary reward and non-monetary recognition, taking into account the meaning of promotion (Frank 1984; Robbins 2001), the cross-loading was ignored and this item was grouped into one form of non-monetary recognition.

The following sections present the results of the inferential statistical techniques used in the present study. To test the 18 proposed hypotheses, the Pearson Product–Moment Correlation Coefficient was computed, as well as the Standard Multiple Regression.

4.2 Correlations

Correlations and descriptive statistics variables are reported in Table 4.2. Cronbach's alphas, shown along the diagonal, indicate that the measures are reliable (alphas from .701 to .933).

Table 4.2 reveals that there is a statistical correlation between most of the variables of organisational factors and the three stages of employee process innovation, except for the relationship between internal communication and generating and experimenting ideas stage of employee process innovation, and the relationship between monetary reward and opportunity exploration stage.

The results indicate that the quality of the supervisor–employee relationship correlates with opportunity exploration stage (r=.277, p<.01), idea generating and experimenting stage (r=.111, p<.01), and idea promotion stage (r=.111, p<.01). This supports the hypotheses that the quality of the supervisor–employee relationship is associated with opportunity exploration stage, idea generating and experimenting stage, and idea promotion stage of employee process innovation (Hypotheses 1a, 1b and 1c).

A correlation is shown to exist between the frequency of internal communication and the opportunity exploration stage (r=-.066, p<.05), and idea promotion stage (r=227, p<.01), supporting the hypotheses that there is an association between the frequency of internal communication and the opportunity exploration stage (Hypothesis 2a) and idea promotion stage (Hypothesis 2c). There was also a correlation between the frequency of external communication and the opportunity exploration stage (r=.111, p<.01), generating and experimenting ideas stage (r=137, p<.01), and idea promotion stage (r=.266, p<.01). This supports the hypotheses that the frequency of external communication is associated with the opportunity exploration stage, generating and experimenting ideas stage, and idea promotion stage of employee process innovation (Hypotheses 3a, 3b and 3c).

A correlation also exists between quality of monetary reward and the generating and experimenting ideas stage (r=176, p<.01), and idea promotion stage (r=.240, p<.01), providing support that there is an association between the quality of monetary reward and the generating and experimenting ideas stage (Hypothesis 4a), and idea promotion stage (Hypothesis 4c).

A correlation was found between non-monetary recognition and the opportunity exploration stage (r=.088, p<.01), generating and experimenting ideas stage (r=272, p<.01), and idea promotion stage (r=.331, p<.01), supporting that recognition is in explaining the variance in the three stages of employee process innovation (Hypotheses 5a, 5b and 5c).

A correlation is shown to exist between training and the opportunity exploration stage (r=340 p<.01), generating and experimenting ideas (r=.250, p<.01) and the idea promotion stage (r=187, p<.01), supporting the hypotheses that there is an association between quality of training and the three stages of employee process innovation (Hypotheses 6a, 6b and 6c).

Construct	Mean	Std.	1	2	3	4	5	6	7	8	9
Respondent gender	1.58	.494									
Education level	3.61	.833									
The work area	4.23	2.496									
The length of work time	2.89	1.082									
Dependent variables											
Opportunity exploration	10.30	2.549									
Generating and experimenting ideas	29.17	7.698	.507**								
Ideas suggest and promotion	18.06	4.821	.465**	.587**							
Independent variables											
Supervisor–employee relationship	37.62	6.156	.277**	.111**	.111**						
Internal communication	12.47	3.746	066*	.019	.227**	- .203**					
External communication	12.06	6.196	.111**	.137**	.266**	.055*	.224**				
Monetary reward	9.37	3.309	.015	.176**	.240**	.047	.082**	.398**			
Non-monetary recognition	7.52	2.913	.088**	.272**	.331**	.065*	.054	.276**	.702**		
Training	18.32	4.305	.340**	.250**	.187**	.397**	- .226**	.076**	.069**	.166**	1
*Correlation is signific	ant at	the 0.05	level	(2-tailed);	**Corr	elation i	s signifi	cant at	the 0.0	l level	(2-tailed)

 Table 4.2: Means, standard deviations and Pearson correlations among variables (N=1,299)

4.3 Multiple regression analysis

For hypotheses testing, regression analysis was employed in this study. Using regression analysis, the researcher aimed to test the effect of each organisational factor on the three stages of employee process innovation respectively. As discussed in Chapter 3, the demographical features of employees may affect their innovative work behaviour (e.g. Kanter 1984; Pinchot 1985; Mumford & Gustafson 1988; Tierney & Farmer 2002; Lee & Wong 2004). Therefore, gender, education level, work area, length of service time and organisation size acted as controls in this study. Detailed results are presented in the following sections.

It is worth noting that a standardised regression coefficient (beta coefficient) allows for a direct comparison between coefficients regarding their relative explanatory power of the dependent variable (Hair et al. 1998), but the beta coefficient should only be a guide to the relative importance of the independent variables included in the equation, and only for those variables with minimal multicollinearity (Hair et al. 1998). Thereby, prior to interpreting the regression results, the degree of multicollinearity was checked by examining the variance inflation factor (VIF) associated with each independent variable in the regression equation (Hair et al. 1998).

The VIF received its name from the fact that the square root of the VIF is the degree to which the standard error has been increased due to multicollinearity (Hair et al. 1998). It refers to an indicator of the effect that the other independent variables have on the standard error of a regression coefficient (Hair et al. 1998). It suggests that large VIF values also show a high degree of multicollinearity among the independent variables (Hair et al. 1998), but there is no commonly accepted VIF value for determining the presence of multicollinearity. According to Hair et al. (1998), values of VIF that exceed 10 are often regarded as indicating multicollinearity. The results show that VIF values that indicate multicollinearity do not appear to be an issue in this research. The maximum VIF is 1.623, which is significantly below the recommended maximum values of 10 in the literature (Hair et al. 1998). The results of the regression analysis are then presented.

To test the hypotheses, three multiple regressions were run for the dependent variable of opportunity exploration stage, generating and experimenting ideas stage, and ideas suggest and promotion stage, respectively. Each regression was run by regressing all control variables and measures of organisational factors. These regressions were used to examine the effect of organisation-related factors on each stage of employee process innovation, after controlling for the possible effect of respondents' and organisations' characteristics.

4.3.1 Opportunity exploration stage, generating and experimenting ideas stage, and promoting ideas stage

First, as shown in Table 4.3, for the opportunity exploration stage of employee process innovation, the total variance explained by the model as a whole was 24.2%, *F* (11, 1287)=38.63, p<.01. The results indicate that the variables of quality of the supervisor–employee relationship, external communication, non-monetary recognition and training were statistically significant, with the quality of the supervisor–employee relationship recording a higher beta value (β =.166, p<.01) than other organisation-related factors. Therefore, the hypotheses that examined the links between the supervisor–employee relationship (H1a), external communication (H3a), non-monetary recognition (H5a), training (H6a) and opportunity exploration were supported.

Variable	β (N=1,299)
Control variables	
Gender	.083***
Education	.016
Work area	263***
Organisation size	.109***
Work length	088***
Independent variables	
Quality of the supervisor–employee relationship	.166***
Internal communication	.002
External communication	.094***
Monetary reward	086**
Non-monetary recognition	.071**
Training	.163***
Adjusted R ²	.242
F-value	38.63 (sig .000)

Table 4.3: Results of regression analyses on opportunity exploration

Dependent variable: Opportunity exploration stage.

* *p*<.10; ***p*<.05; *** *p*<.01

Second, for generating and experimenting ideas, as shown in Table 4.4, the total variance explained by the model as a whole was 15.1%, *F* (11, 1287)=21.94, p<.001. The results show that only the variables of non-monetary recognition and training were statistically significant. Therefore, the hypotheses that examined the links between non-monetary recognition (H5b), training (H6c), and generating and experimenting ideas were supported.

Variable	β (N=1,299)
Control variables	
Gender	.083***
Education	.016
Work area	202***
Organisation size	.026
Work length	032
Independent variables	
Quality of the supervisor-employee relationship	.012
Internal communication	.041
External communication	.048
Monetary reward	001
Non-monetary recognition	.204**
Training	.155**
Adjusted R ²	.151
F-value	21.94 (sig .000)

Table 4.4: Results of regression analyses on generating and experimenting

ideas

Dependent variable: Generating and experimenting ideas stage.

* *p*<.10; ***p*<.05; *** *p*<.01

Third, as shown in Table 4.5, for promoting ideas, the five control variables explained 10.4% of the variance. By regressing the six corporate-level factors, the total variance explained by the model as a whole was 26.7%, *F* (11, 1287)=44.03, p<.001. The results show that the variables of quality of the supervisor–employee relationship, internal communication, external communication, non-monetary recognition and training were statistically significant, with the non-monetary recognition recording a higher beta value (β =.270, p<.001) than others. Therefore, the hypotheses that examined the links between the supervisor–employee relationship (H1c), internal communication (H2c), external communication (H3c), monetary reward (H4c), non-monetary recognition (H5c), training (H6c) and idea promotion were supported.

Variable	β (N=1,299)
Control variables	
Gender	005
Education	.034
Work area	237***
Organisation size	.093***
Work length	034
Independent variables	
Quality of the supervisor–employee relationship	.082**
Internal communication	.223***
External communication	.131***
Monetary reward	025
Non-monetary recognition	.270***
Training	.070**
Adjusted R ²	.267
F-value	44.03 (sig .000)

 Table 4.5: Results of regression analyses on idea promotion

Dependent variable: Ideas suggest and promotion stage.

* *p*<.10; ***p*<.05; *** *p*<.01

In summary, for opportunity exploration, the most relevant corporate-level factors are the perceived quality of the supervisor–employee relationship, frequency of external communication and perceived quality of training. For generating and experimenting ideas, the most relevant corporate-level factors include perceived quality of non-monetary recognition and perceived quality of training. For promoting ideas, corporate-level factors including the frequency of internal communication, frequency of external communication and perceived quality of non-monetary recognition appeared to be the most relevant predictors. The results from the hypotheses testing are presented in Table 4.6.

Hypotheses	Supported	
H1a: The perceived quality of the supervisor employee relationship	Vac	
is positively associated with opportunity exploration stage of		
employee process innovation.		
H1b: The perceived quality of the supervisor-employee relationship	No	
is positively associated with generating and experimenting ideas stage		
of employee process innovation.		
H1c: The perceived quality of the supervisor–employee relationship	Yes	
is positively associated with ideas suggest and promotion stage of		
employee process innovation.		
H2a: The frequency of internal communication is positively	No	
associated with opportunity exploration stage of employee process		
innovation.		
H2b: The frequency of internal communication is positively	No	
associated with generating and experimenting ideas stage of		
employee process innovation.	X 7	
H2c: The frequency of internal communication is positively	Yes	
associated with ideas suggest and promotion stage of employee		
H2a: The frequency of external communication is positively	Vac	
associated with opportunity exploration stage of employee process	168	
innovation		
H3b: The frequency of external communication is positively	No	
associated with generating and experimenting ideas stage of	110	
employee process innovation.		
H3c: The frequency of external communication is positively	Yes	
associated with ideas suggest and promotion stage of employee		
process innovation.		
H4a: The perceived quality of monetary reward is positively	No	
associated with opportunity exploration stage of employee process		
innovation.		
H4b: The perceived quality of monetary reward is positively	No	
associated with generating and experimenting ideas stage of		
employee process innovation.	N 7	
H4C: The perceived quality of monetary reward is positively	res	
associated with ideas suggest and promotion stage of employee		
H5a: The perceived quality of non-monetary recognition is positively	Vac	
associated with opportunity exploration stage of employee process	105	
innovation		
H5b: The perceived quality of non-monetary recognition is positively	Yes	
associated with generating and experimenting ideas stage of	100	
employee process innovation.		
H5c: The perceived quality of non-monetary recognition is positively	No	
associated with ideas suggest and promotion stage of employee		
process innovation.		

Table 4.6: Summary of the hypotheses testing results

H6a: The perceived quality of training is positively associated with opportunity exploration stage of employee process innovation.	Yes
H6b: The perceived quality of training is positively associated with generating and experimenting ideas stage of employee process innovation.	Yes
H6c: The perceived quality of training is positively associated with ideas suggest and promotion stage of employee process innovation.	Yes

4.4 Chapter summary

This chapter has presented the results of data analysis and hypothesis testing. The next chapter provides a discussion of the findings of the current research and recommendations for future research. Implications and limitations of the present study will also be presented in Chapters 5 and 6.

Chapter 5: Discussion

This chapter discusses the empirical results observed in this study. The discussion is organised to reflect the 18 hypotheses analysed in Chapter 4. As shown in Chapter 2, this study aims to examine the effects of organisational factors, including the quality of the supervisor–employee relationship, internal communication, external communication, monetary reward, non-monetary recognition and training, on the different stages of employee innovation behaviour that produces incremental refinements and improvements in daily work processes. In the following sections, the discussion of the results is framed in relation to employee innovation and the six organisational factors. Finally, a summary is provided at the end of this chapter.

5.1 Supervisor–employee relationship

The regression analysis proposed that the quality of the supervisor–employee relationship is positively associated with opportunity exploration (Hypothesis 1a), idea generation and experimentation (Hypothesis 1b), and idea promotion (Hypothesis 1c) of the employee innovative process. The results show a positive association between opportunity exploration and the supervisor–employee relationship quality (Hypothesis 1a), as well as a positive association between idea promotion and the quality of the supervisor–employee relationship (Hypothesis 1c). However, no relationship was found in the empirical data between idea generation and experimentation with the quality of the supervisor–employee relationship (Hypothesis 1b).

One possible explanation for the finding that the quality of the supervisoremployee relationship was not positively associated with idea generation may be that the different aspects of innovation have different aetiologies. As innovation is a social process, idea promotion is heavily reliant on the involvement of others. While an employee can generate new ideas alone, putting ideas into practice typically depends on the approval, support and resources of others (Axtell et al. 2000). Therefore, the quality of the supervisor–employee relationship might affect the idea promotion stage. In addition, idea generation and experimentation is more dependent on an employee's personal characteristics (Axtell et al. 2000), which is consistent with much of the literature on creativity, which usually proposes that individual-level characteristics are likely to have more of an influence on idea experimentation than organisational factors. This might explain why the association between the quality of the supervisor–employee relationship and idea generation and experimentation stage was not supported.

Additionally, previous studies on the LMX theory argue that a higher-quality supervisor–employee relationship is associated with greater autonomy, to which employees are given latitude to carry out their tasks without excessive supervision (Basu & Green 1997). By providing employees with autonomy at work, supervisors can help create a work environment that encourages free thinking, exchange of information and the latitude to explore the opportunity for new or improved ideas. Employees are expected to have the opportunity to engage in unusual thoughts and behaviour. Hence, this study provides further evidence to support the positive effect of the quality of the supervisor–employee relationship on the opportunity exploration stage of the employee innovation process found in previous studies.

5.2 Internal communication

The current study suggested that internal communication is positively associated with opportunity exploration (Hypothesis 2a), idea generation and experimentation (Hypothesis 2b), and idea promotion (Hypothesis 2c) of the employee innovative process. The results show that the proposed positive association between idea promotion and internal communication (Hypothesis 2c) was supported. However, neither a positive association between opportunity exploration and internal communication (Hypothesis 2a), nor a positive association between idea generation and experimentation and internal communication (Hypothesis 2b), was found in the empirical data.

The finding that internal communication is positively associated with idea promotion may be because while employees can find new opportunities for improvements alone, to put them into practice, they need the support and involvement of others in their organisation (Van de Ven, Angle & Poole 1989; Axtell et al. 2000; Magadley & Birdi 2012). As promoting ideas typically depends on the support and acceptance of others, communication with others might be helpful in facilitating employees to promote their innovative ideas at work. Thus, the proposed association between it and the idea promotion stage of employee process innovation was supported in the present study.

One possible explanation for the finding that internal communication was neither positively associated with idea generation nor with the opportunity exploration stage might be that individual-level factors such as creative self-efficacy are important for the generation of novel ideas (Magadley & Birdi 2012). Prior research (e.g. Axtell et al. 2000) suggested that organisations wishing to encourage employees to generate new ideas should focus on the individual-level factors. Another possible explanation is that as previous studies suggested, a high level of communication frequency can create mutual production blocking, which means a tendency for one individual to inhibit or block other people during a group discussion (e.g. Diehl & Stroebe 1987), which can limit the cognitive capacity of employees (Nijstad 2000). This is supported by prior research (Perry-Smith & Shalley 2003), which argued that too-frequent communication within organisations might discourage employees from generating new ideas. Amabile and Conti (1999) also suggested that if the frequency of communication at work reaches a certain level, the mutual influence becomes so high that it deteriorates the work environment for innovative work performance up to the situation in which group thinking (Janis 1972) occurs. Länsisalmi, Peiro and Kivimaki (2000) also proposed that very frequent interaction among employees might overburden personnel with having to attend too many meetings and overload them with information, which in turn discourages innovative activities.

5.3 External communication

The empirical data also suggested that external communication helps to facilitate opportunity exploration (Hypothesis 3a), idea generation and experimentation (Hypothesis 3b), and idea promotion (Hypothesis 3c) of the innovative process of employees. The results show a positive association between the opportunity exploration stage and external communication (Hypothesis 3a), and a positive association between the idea promotion stage and external communication (Hypothesis 3c). However, the association between the idea generation and experimentation and experimentation stage and external communication (Hypothesis 3c). However, the association between the idea generation and experimentation stage and external communication (Hypothesis 3b) was not found in the empirical data.

The finding that internal communication is positively associated with the opportunity exploration stage may be because employees' diverse contacts at work might expose them to more various perspectives and ideas (De Jong & Den Hartog 2010). With diverse networks, employees are expected to be more likely to obtain various information and ideas, and to discover the opportunity for sparking ideas at work. Prior research proposed that outside contacts such as customers could contribute to innovations (e.g. Von Hippel 1988). This is also in line with the findings of Perry-Smith and Shalley (2003), who confirmed that diverse external networks allow employees to be exposed to non-redundant information and therefore boost deviant behaviour. This would create opportunities for employees in championing ideas.

In addition, as showed earlier, generating ideas is heavily reliant on individuallevel factors such as domain expertise, which includes relevant skills and knowledge (Magadley & Birdi 2012). Individual-level characteristics are likely to have more of an influence on employees experimenting with ideas than organisational factors (Axtell et al. 2000). This might explain why no relationship was found between external communication and idea generation and experimentation. Another possible explanation is that, similar to communication within organisations, very frequent external communication might discourage employees' efforts in generating new ideas. For example, Kratzer, Leenders and Engelen (2004) suggested that very frequent communication might mould the environment in such a way that mutual production blocking occurs or the mutual influence restricts individual attempts to perform creatively. It suggested that both quantity and quality of communication is important for employees generating new ideas (Ohly, Kase & Škerlavaj 2010). This suggests that more research is needed to examine the effect of communication at work on employee innovation from a broader perspective (i.e. investigating both frequency and quality).

5.4 Reward and recognition

It was found that the quality of monetary reward is positively associated with the opportunity exploration stage (Hypothesis 4a), idea generation and experimentation stage (Hypothesis 4b), and idea promotion stage (Hypothesis 4c) of the employee innovative process. The results show that only a positive association between the idea promotion stage and quality of monetary reward (Hypothesis 4c) was supported. However, neither an association between the opportunity exploration stage and quality of monetary reward (Hypothesis 4a), nor an association between the idea generation and experimentation stage and quality of monetary reward (Hypothesis 4a),

The empirical data also suggested that quality of non-monetary recognition is positively associated with opportunity exploration (Hypothesis 5a), idea generation and experimentation (Hypothesis 5b), and idea promotion (Hypothesis 5c) of the employee innovative process. The results show that a positive association between opportunity exploration and quality of non-monetary recognition (Hypothesis 5a) was supported, and a positive association between idea generation and experimentation and quality of non-monetary recognition (Hypothesis 5b) was also supported. However, an association between idea promotion and quality of non-monetary recognition (Hypothesis 5c) was not found in the empirical data.

One possible explanation for the findings is that creativity can be viewed as selfmotivated psychological behaviour that is typically sparked by an intrinsic spiritual reward (Deci & Ryan 1985; Amabile 1997; Hennessey & Amabile 1998). This implies that an intrinsic reward is likely to have more of an influence on employees looking for opportunities for innovation. In this way, employees are expected to have the opportunity to engage in unusual thoughts and behaviour, and to look for new opportunities. This might explain why the proposed association between non-monetary recognition and the opportunity exploration stage was supported.

Another possible explanation for these findings is that although most reward systems are designed with the view that money can motivate employees and higher compensation can improve employee performance (Lawler 1969), prior research suggested that monetary incentives might not have the same effect across different categories of employees. Rynes and Gerhart (2003) confirmed this and suggested that employees might favour different incentives according to their needs, education, social status or circumstances.

In addition, despite the theoretical and empirical evidence showing the positive effect of monetary reward on employee creativity, some previous research (e.g. Deci & Ryan 1985; Amabile et al. 1996; Cooper 1999) revealed that monetary reward might not be the best incentive to stimulate idea generation. Monetary reward might reduce the autonomy and self-motivation of employees, diverting their attention to economic benefits and weakening their proactive innovative behaviour. In this way, consistent with some of the literature on creativity (e.g. Lepper, Greene & Nisbett 1973; Lepper & Greence 1978), excessive monetary reward might depress employee innovative behaviour towards idea generation and experimentation by eroding self-motivation (Zhou, Zhang & Montoro-S ánchez 2011). This might also explain why the association between monetary reward and the idea generation and experimentation stage was not supported.

In contrast, given that generating and experimenting ideas is more dependent on an employee's personal characteristics (Axtell et al. 2000), and intrinsic motivation has been suggested to be related to a person's need for personal development, such as the feeling of enjoyment and accomplishment that accrue spontaneously as a person engages freely in the target activities (Deci, Ryan & Williams 1996), it has been proposed that intrinsic rewards are likely to have more of an influence than monetary reward on experimentation with new ideas.. In support, Lu and Zhang (2007) suggested that a high level of intrinsic work motivation might make employees more self-motivated and therefore more involved in innovative behaviour towards idea generation. This might explain why the association between non-monetary recognition and the idea generation and experimentation stage was supported.

As discussed in the literature review, innovation is a social process. Accordingly, as Lu and Zhang (2007) suggested, once an employee generates a new idea, intrinsic motivation might play a weaker role in influencing the following stages of individual innovation, such as the implementation of the ideas stage. This is because, as discussed earlier, while an employee can generate ideas alone, promoting ideas is heavily reliant on the involvement of others. In this way, extrinsic motivators might have more of an influence on employees compared with self-motivation. This is supported by Lu and Zhang (2007), who suggested that employees might feel less self-motivated when they need to obtain support from the organisation. In this way, a tangible reward is likely to have more of an influence on the idea promotion stage. This might explain why the proposed association between monetary reward and the idea promotion stage was supported, but the association between non-monetary recognition and the idea promotion stage was not supported.

5.5 Training

The regression analysis confirmed that training is positively associated with the opportunity exploration stage (Hypothesis 6a), idea generation and experimentation stage (Hypothesis 6b), and idea promotion stage (Hypothesis 6c) of the employee innovation process towards process innovation. The results show positive associations between quality of training and all three stages of employee innovation process were supported.

The empirical findings indicated that training was associated with the three stages of employee process innovation to different extents. The results indicate that training was more strongly associated with the opportunity exploration stage compared with its associations with the idea generation and experimentation stage, and idea promotion stage. This finding is in line with the stage-specific research direction proposed in the existing literature (e.g. Birdi 2007), which suggested that employees might require different skills, knowledge and capabilities for different stages of the innovation process. Therefore, it is reasonable that training could have different effects by stages of the employee innovation process. Our findings provide empirical evidence supporting the notion that the influence of training on the stages of the employee innovative process may differ.

Additionally, one possible explanation for the finding that training was more strongly associated with opportunity exploration than other stages of process innovation is that following previous literature (e.g. Van de Ven, Angle & Poole 1989), exploring opportunities to innovate is primarily an internal process, whereas putting novel ideas into practice and promoting them to others tends to be a social process that requires the involvement of others. For example, employees often need time, resources, support and agreement from those in authority or those affected by the potential changes. Such environmental factors are not under the influence of the individual employee. Therefore, training might be limited to mainly stimulating employees to explore opportunities for generating ideas, while other features more directly dictate whether those ideas are put into practice. In support, Birdi (2007) suggested that certain creativity training activities can affect as managerial support have a greater influence on whether such ideas are put into practice.

Previous studies on training argue that employees can enhance their capabilities and skills needed for work through training and development (e.g. Basadur 2004). The general support for such an effect on the employee innovation process from idea generation to idea implementation has been provided. For example, training and development have been linked to receptivity to new ideas and innovations as well as to the generation of innovative proposals (Thompson 1965; Katz & Tushman 1981; Damanpour 1991; Hurley & Hult 1998). As training and development improve an employee's ability to diagnose and solve technical problems, the odds are increased that innovative proposals will become effective practices (McGinnis & Ackelsberg 1983; Dewar & Dutton 1986). Along these lines, employees should be able to generate more ideas at work and, as a result, have more of their ideas implemented if they are receiving methods for developing more solutions or ideas (Clegg et al. 2002). The present study provides further evidence to support the positive effects of training on employee process innovation.

5.6 Effects of organisational factors on stages of employee process innovation

This study empirically distinguishes between the different dimensions of employee innovation by identifying that employee innovation could be viewed as a process consisting of the opportunity exploration stage, idea generation and experimentation stage, and the idea promotion stage. This is in line with the pattern identified by previous work (Kanter 1988; Scott & Bruce 1994; Oldham & Cummings 1996; Axtell et al. 2000; Kleysen & Street 2001; De Jong & Den Hartog 2010; Holman et al. 2011). Further, along with the direction proposed that future research should make further distinctions between stages of innovation (Hemphälä & Magnusson 2012), our findings provide empirical evidence supporting the notion that different stages of innovation can have different organisational correlations (Axtell et al. 2000; Birdi 2007; Magadley & Birdi 2012). For innovation management practice, the findings imply that organisations wishing to promote opportunity exploration for innovation, idea generation and experimentation, and idea promotion among employees should focus on different organisational factors, which could help to facilitate employee innovation in daily work processes effectively and efficiently.

Specifically, as shown in Figure 5.1, this research found that the quality of the supervisor–employee relationship, external communication, non-monetary

recognition and training were positively associated with the opportunity exploration stage. Non-monetary recognition and training were positively associated with the idea generation and experimentation stage. The quality of the supervisor–employee relationship, internal communication, external communication, monetary reward and training were positively associated with the idea promotion stage. Each stage of the employee innovation process is discussed in the following sections.



Figure 5.1: Effects of organisational factors on different stages of the employee innovation process

5.6.1 Opportunity exploration stage

This study explicitly found that four organisational factors were positively associated with opportunity exploration. This confirms the value of continuing to develop theories and research that explores how organisational factors are associated with fostering the process innovation of employees by stages and which factors are associated with facilitating employee opportunity exploration specifically.

Opportunity exploration of employees can be viewed as a starting point for employee innovation, which is important because employees need to break out of the routine of doing things and consider making an extra effort for change in the workplace (e.g. Kleysen & Street 2001). More specifically, opportunities to generate ideas often lie in incongruities and discontinuities—things that do not fit expected patterns, such as problems in existing work methods, unfulfilled needs of customers or indications that trends may be changing (De Jong & Den Hartog 2007).

Extant literature suggested that employees are required to be both able and willing to be innovative (e.g. Huhtala & Parzefall 2007). In this respect, this research explicitly found that four organisational factors were positively associated with opportunity exploration: quality of the supervisor–employee relationship, external communication, non-monetary recognition and training.

With respect to abilities, general skills and task- and context-specific knowledge might be important to drive employees who explore opportunities for innovation (e.g. Barron & Harrington 1981). As Shalley and Gilson (2004) suggested, employees need to be more aware of different alternatives and opportunities for innovation by developing a more extensive skill set. This implies the important role of external communication, which was suggested to be one source of new ideas (De Jong & Den Hartog 2010), and training, which could be used to help employees learn more about the skills and knowledge (Basadur 2004) in facilitating employees to explore opportunities for innovation.

Specifically, this study empirically suggests that external communication can affect opportunity exploration because it allows employees to obtain information and ideas, and to discover the opportunity for innovative ideas in the workplace (e.g. De Jong & Den Hartog 2010; Perry-Smith & Shalley 2003). In addition, as training involves providing employees with basic knowledge and skills (e.g. Costen & Salazar 2011; Johannessen & Olsen 2003; Lau & Ngo 2004), and training strategies have been found to enhance individuals' creative thinking skills and problem-solving abilities (e.g. Feldhusen & Goh 1995), this study supports the notion that training can affect the opportunity exploration of employees by ensuring they are able to innovate.

Beyond skills and knowledge, management needs to push employees to persevere in the face of challenges inherent in innovative work (Georgsdottir & Getz 2004; Shalley & Gilson 2004). Current results are in line with this view by supporting the positive effects of the quality of the supervisor-employee relationship and non-monetary recognition. Specifically, based on the extant work, which suggests that it is important to make employees feel encouraged to take risks and break out of safe ways of doing things if management wants them to be willing to explore opportunities for innovation (e.g. Shalley & Gilson 2004), this study takes a further step by explicitly suggesting that the organisational factors of the quality of the supervisor-employee relationship and non-monetary recognition concerning supervisor support and self-motivation might be important to facilitate opportunity exploration. Future theory and research should pay more attention to the issue of whether there is a causal relationship between these specific organisational factors and opportunity exploration, and how these factors influence employee innovative behaviour towards opportunity exploration.

5.6.2 Idea generation and experimentation stage

This study explicitly found that two organisational factors were positively associated with idea generation and experimentation. This confirms the value of continuing to develop theories and research, which explores how organisational factors are associated with fostering the process innovation of employees by stages and which factors are associated with facilitating idea generation and the experimentation of employees specifically.

Idea generation and the experimentation of employees refer to employees generating new ideas and trying them out before promoting them to others (e.g. Kleysen & Street 2001). This research found that only two organisational factors were positively associated with opportunity exploration: non-monetary recognition and training. This confirms the view that employees might not heavily rely on the involvement of others because they can generate new ideas alone (Axtell et al. 2000), and intrinsic motivation could be useful to facilitate idea generation (Amabile 1979; Amabile, Goldfarb & Brackfield 1990).

Specifically, the extant literature proposed that intrinsically motivated employees are more likely to explore new pathways and to take greater risks for generating new ideas (Amabile, Goldfarb & Brackfield 1990) because they need to be sufficiently interested in a certain problem and/or outcome and in finding ways to solve or achieve it (Coelho, Augusto & Lages 2011). In this way, they will be more excited about their work, and this in turn increases their idea generation (Shalley, Zhou & Oldham 2004).

This study further supports this view by identifying the positive associations between idea generation and experimentation and two organisational factors. Specifically, this study suggests that recognition that is linked to intrinsic motivation can affect the idea generation of employees (e.g. Eisenberger & Armeli 1997; McAdam & McClelland 2002). Additionally, as Shalley and Gilson (2004) suggested, by developing a more extensive skill set through training, employees should be more comfortable in trying new things. The findings of this study suggest that with training, employees might be more likely to perform not only their work, but also be curious about other solutions, options or ways of doing their work so that idea generation and experimentation can occur.

Relevant theories and research should further examine the view that intrinsically motivated employees might be more likely to generate new ideas in the workplace.

The issue of which specific organisational factors could drive employees to generate new ideas and how they could influence employee innovative behaviour should be researched further.

5.6.3 Idea promotion stage

This study explicitly found that five organisational factors were positively associated with idea promotion. This confirms the value of continuing to develop theories and research that explores how organisational factors are associated with fostering the process innovation of employees by stages and which factors are associated with facilitating employee idea promotion specifically.

The idea promotion stage is about employees promoting new ideas to others. Extant work suggested that promoting ideas is related to the involvement of others in terms of support and resources from their organisation (e.g. Axtell et al. 2000). This study suggests that the quality of the supervisor–employee relationship, internal communication, external communication, non-monetary recognition and training might help to distribute and promote ideas, whereas monetary reward was not positively associated with the idea promotion of employees. To examine whether these factors can drive employees to promote the new ideas, and to what extent, theories and research should focus on the organisational factors that might foster employees to promote new ideas.

Interestingly, the finding shows that opportunity exploration, idea generation and experimentation, and idea promotion have different correlations. This agrees with the stage-specific perspective on innovation (e.g. Axtell et al. 2000; Magadley & Birdi 2012) and advances the employee innovation literature by addressing the specific influence of organisational factors on different stages of employee innovation respectively. It suggests that the factors identified as antecedents to employees' idea generation and development (e.g. Amabile 1996; Shalley & Zhou 2004; Woodman, Sawyer & Griffin 1993) might need to be investigated in more detail because factors that specifically facilitate opportunity exploration might not affect idea generation or idea promotion.

Additionally, as mentioned in Chapter 5, this study included gender, education level, work area, work length and organisation size as control variables. In doing so, it was expected to gain an in-depth understanding of the organisation's influence on employee innovation in the workplace. This study found that controlling for these factors, a different mix of organisational factors was positively associated with different stages of employee innovation. The results suggest that these associations were not affected by the gender, education level or work length of employees, whereas the variables of work area and organisation size might influence the associations. This indicates that the findings might not hold equally true for employees from specialised technical work areas or less specialised functional areas, or for employees from large-, small- or medium-sized organisations. This finding can be explained by previous work (Kanter 1984; Pinchot 1985), which suggested that work experience from technical-oriented work might improve the likelihood of entrepreneurial behaviour, and large organisations could offer more opportunities for learning experiences for employees. Future theory and research should examine whether the influence of these organisational factors can work in the same way for employees with different demographical characteristics.

In conclusion, as empirical evidence showed that organisational factors could affect different aspects of employee innovation process, this study has drawn attention to the need to extend theoretical understanding of the interactions between the organisational factors and different stages of the employee innovation process. The identified relationships between organisational factors and the stages of employee innovation process should be examined in greater detail to understand how they affect an organisation's ability to facilitate employee innovation (Smith et al. 2008). In addition, it is important to distinguish between different stages, and clearly clarify which of these stages is being investigated when considering innovation models, future theory and research on employee innovation should focus in greater depth on the factors that relate differently to the stages of innovation process.

5.7 Chapter summary

This chapter discussed the empirical results of the data analysis undertaken to address the hypotheses, regarding the effect of the quality of the supervisor– employee relationship, internal communication, external communication, monetary reward, non-monetary recognition and training on employee process innovation stage by stage. The results show that organisational factors differently affect the stages of employee innovation. Specifically, the quality of the supervisor–employee relationship, external communication, non-monetary recognition and training were found to play a significant role to facilitate opportunity exploration while non-monetary recognition and training encouraged idea generation and experimentation. Moreover, the quality of the supervisor– employee relationship, internal communication, external communication, nonmonetary recognition and training helped to distribute and promote ideas. The following chapter will conclude with theoretical and practical implications, as well as the limitations and potential directions for future research.

Chapter 6: Conclusion

This chapter examines theoretical and practical implications, as well as limitations and potential directions for future research. Overall conclusions are provided at the end of this chapter.

6.1 Implications

This study has important theoretical implications for employee innovation research and practical implications for organisations to encourage employee innovation in daily working processes. Detailed discussions are presented in the following sections.

6.1.1 Theoretical implications

By examining the link between different stages of employee innovation concerning incremental refinements and improvements in daily work processes and organisational factors, one key theoretical implication is that different stages of employee innovation are positively associated with a different mix of organisational factors. This implies that future research is needed to discover which organisational factors are more important in facilitating specific stages of the employee innovation process. There has been a need to empirically investigate the specific connection between organisational factors and different stages of employee innovation. Along this line, Denti and Hemlin (2012) found that in the early stages of innovation, leaders may have to take a divergent and explorative approach to problem construction and ideation in which knowledge and ideas are broadly integrated.

In addition, this study contributes to employee innovation literature by explicitly investigating employee process innovation concerning incremental and minor refinements and improvements in employees' daily working processes. Considering the distinction between incremental innovation and radical innovation (e.g. Isaksen & Tidd 2006), this study has taken the first step in extending innovation research by explicitly addressing incremental process innovation rather than radical innovation. Specifically, by examining the incremental innovation of employees, it has showed that six organisational factors—supervisor–employee relationship, internal communication, external communication, monetary reward, non-monetary recognition and training—are relevant to employee innovation regarding expanding and refining existing processes.

Moreover, this study contributes to the literature in its stage-specific perspective on employee innovation. This research confirms the value of adopting a multistage perspective on employee innovation. By doing so, people can explore how each stage can affect different organisational factors.

Further, as this study empirically tested the effects of organisational factors in the context of China, it provides a unique environment for investigating such effects compared with western countries (Zhou, Zhang & Montoro-S ánchez 2011).

Specifically, as Madjar, Oldham and Pratt (2002) suggested, different results could be obtained in different countries. In particular, as Hon (2010) noted, previous findings from western literature may not apply to Asia because the innovative performance of employees depends on social–contextual factors such as leadership behaviour (e.g. Scott & Bruce 1994; Woodman et al. 1993). Additionally, as McClelland (1987) suggested, the achievement motive involving the creation of more efficient ways of doing things and solving problems may not apply to Chinese employees, whereas the avoidance motive might be high in China, as the fear of being punished for mistakes seems to be deep-rooted in the traditional culture of mainland China (McClelland 1987; Schermerhorn & Nyaw 1990). This may lead to passivity in the workplace, where taking risks is avoided (Jackson & Bak 1998). This in turn discourages employee behaviour towards innovation in daily work processes. However, there have been few empirical studies of the application of the employee innovation process in the Chinese context.

Consistent with previous studies (Jackson & Bak 1998; Wang, Xu & Su 2010), this study provides the first step in determining whether Chinese employees may be motivated differently in the workplace compared with the western world, and it enriches the understanding of how to facilitate the employee innovation process in the global context.

6.1.2 Practical implications

With regard to practice, in influencing employee process innovation, organisational factors such as the supervisor–employee relationship, internal communication, external communication, monetary reward, non-monetary recognition and training are important. Specifically, organisations need to recognise that the factors that may facilitate the employee innovation process and the skills required can be different depending on the stage of the innovation. While it has been proposed that organisations wishing to facilitate employees' idea generation should focus on individual-level factors such as self-efficacy and domain expertise, which could be developed by a host of training initiatives (Magadley & Birdi 2012), the findings of this study provide empirical support for the view that there needs to be effective organisational support for ideas to be put into practice. Moreover, this study further comments that while employees can generate new ideas alone, they need the support and involvement of others in their organisation to put them into practice.

From a practical perspective, this study will help business managers to improve their understanding of the different stages of employee process innovation and the role of organisational-level factors in facilitating employee innovation. The findings suggest that business managers could utilise organisational resources to improve the innovative behaviour of employees in their daily work rather than focusing solely on product innovation. Additionally, the findings of this study imply that business managers should also target the correct organisational support for each stage of employee innovative behaviour in order to maximise employee innovative behaviour. In particular, if an organisation focuses on fostering their employees to generate novel ideas at work, they could invest in improving the quality of the supervisor–employee relationship, external communication, nonmonetary recognition and training for their employees. In addition, investing in non-monetary recognition and training might help their employees to generate and experiment with ideas. Moreover, to foster their employees to promote novel ideas, this study recommends investing in the quality of the supervisor–employee relationship, internal communication, external communication, monetary reward and training.

In addition, this study finds that the organisational factors can significantly impact on the incremental process innovation of the employees. The results had no much difference with previous studies in Western cultural setting (e.g. Scott & Bruce 1994). This suggests that organisations can improve the innovation and creativity of the employees by systematically implementing well–designed measures at the organisational level though the cultural values and traditions of China differ from the rest part of the world.

6.2 Limitations of the study

Several limitations of this study should be noted. These limitations can be categorised in terms of the study sample and its method.

One sample limitation of the research lies in its focus on a single country, which restricted the generalisability of the findings in different situations. The study results may reflect its specific characteristic and may not therefore be generalisable to other countries. There is a need for future research outside China to examine whether the findings developed in this study hold true for different settings.

Although the statistical analyses suggested that the sample for this study was acceptable, the convenience sampling method has a limitation in that the findings

cannot be generalised. Thus, it is preferable to use a random sample. Another sample limitation stems from the possibility of sampling selection bias. However, this study paves the way for future research to explore whether the findings could apply to other samples and populations. Future research should also examine whether the theoretical model proposed in this study also holds true for different people.

This study is cross-sectional. The cross-sectional research design only established associations between variables. Therefore, this research cannot make any definitive statements regarding the causality of the included variables (Huang, Rode & Schroeder 2011). For future research, it is suggested that a case study design be used to further examine the underlying logic of how corporate-level factors can facilitate employee process innovation and thus provide evidence for the causal relationship among the variables. According to Saunders, Lewis and Thornhill (2000), interviews provide the most suitable research tool in situations where researchers need to understand the background for respondents' decisions or attitudes. In-depth qualitative methods would enable the gathering of rich data that may reflect personal comments from employees about their feelings and perceptions about the influence of organisational factors on their innovative behaviour.

In addition, this study is based on self-reporting data. All of the included variables were measured with surveys that relied on perceptually based measures. The subjective and potentially idiosyncratic nature of respondents' reporting of their views that might influence their innovative behaviour must be acknowledged. This subjectivity should be kept in mind when interpreting the findings of this study.

It is also noted that, common method variance (Podsakoff & Organ, 1986) could be a potential bias in this study because the respondent is the only data source for independent variables and dependent variables. Though the researcher has tried to use Harman's one–factor test (Schriesheim 1979) to address the common method bias issue, the problem cannot be really solved. Another potential bias concerns social desirability, which is potentially due to the respondents perceiving them to be the _right' answers to the questions offered. To be innovative and helpful towards benefiting one organisation is desirable, and the respondents might be tempted to report such behaviour. In addition, this bias might occur because it should be noted that innovative behaviour deviates from prescribed roles; therefore, in some cases, it might be considered threatening by colleagues and supervisors (Åmo & Kolvereid 2005).

6.3 Future research directions

There are several possible directions for future research. First, as the crosssectional design does not permit the interference of the causal-effect relationship, future researchers might employ a case study research design to provide evidence for causality. There should be more longitudinal studies with comparison groups so that causality can be fully established. Second, future research needs to be based on objective indicators and multiple sources. Model variables should be linked to additional sources of data that do not rely on self-report measures and that include multiple management-level informants and case studies. Multiple sources would allow for multi-level insights into the influence of corporate-level factors on employee innovative behaviour, which single-source surveys cannot reveal.

Third, methodologically, it is recommended to use the mixed research method. The approach taken in this study was quantitative, and while qualitative methods help to gather a rich understanding of the research object (Djojosaputro, Nguyen & Peszynski 2005), the qualitative method could provide a more in-depth understanding to reinforce the explanation of the statistical results of quantitative data (Teddlie & Tashakkori 2003). In-depth qualitative methods would help to gather rich data that might achieve more comprehensive insights into the influence of corporate-level factors on employee innovative behaviour from the perspectives of employees and managers. Moreover, the comments gathered using the qualitative method might reveal new insights regarding influences on

employee innovative behaviour that are not covered in the quantitative research design.

Fourth, this empirical study was conducted in the context of China. This research paves the way for future studies to explore whether the findings also hold true for other settings. Similar studies can be conducted in other countries to test this study's findings and thus determine the generalisability of this construct. More studies are needed in both western and non-western organisations.

A direct comparison of the effects of organisational factors on the employees between China and Western countries would help identify how different cultures may impact on the innovative behaviour of the employees. It is suggested that future studies could examine the effects of cultural values and traditions on the incremental process innovation of employees.

In addition, the regression analysis results of this study indicated that organisation size and the work area of employees could be relevant to the influence of organisational factors on employee innovative behaviour towards incremental innovation. Specifically, it showed that the findings might not hold equally true for employees from specialised technical work areas or less specialised functional areas, or for employees from large-, small- or medium-sized organisations. As the findings cannot be generalised to people or settings other than the ones in this study due to the use of a convenience sample, future research could explore this to examine whether the theoretical model of employees from different work areas or different-sized organisations, and address the issues of how it would differ.

Finally, employee innovation is not only influenced by organisational factors. Future research could consider synthetic interactions of different-level factors, including employee and organisation levels on employee innovation, rather than investigate their influences separately. In particular, it is recommended that future research examine the combined effects of such personal and contextual conditions on employee process innovation towards incremental refinements and increasing the efficiency of existing daily work processes.

6.4 Summary

This study set out to examine the six important organisational factors that facilitate employee innovation towards incremental refinements and improvements in daily work processes stage by stage, namely opportunity exploration, idea generation and experimenting, and idea promotion. The results suggest that organisation-related factors might relate differently to different stages of employee innovation (e.g. Birdi 2007; Axtell et al. 2000).

Recognising such different influences on the stages of employee process innovation is critical for organisations to understand how to make the best use of these common corporate resources to effectively facilitate employee process innovative behaviour. In particular, recognising the importance of employee incremental process innovation at work and paying more attention to the different influences of corporate resources on the stages of employee process innovative behaviour are important to facilitate employee innovation by leveraging corporate resources for organisations.
REFERENCE

Abernathy, WJ & Utterback, JM 1975, _A dynamic model of process and product innovation', *Omega*, vol. 3, pp. 639-56.

Abra, J 1994, Collaboration in creative work: An initiative for investigation', *Creativity Research Journal*, vol. 7, no. 1, pp. 1-20.

Afuah, A 1998, Innovation management: Strategies, implementation, and profits, Oxford University Press, New York.

Aiken, M & Hage, J 1971, The organic organization and innovation', *Sociology*, vol. 5, no. 1, pp. 63-82.

Alberto, C 2000, _How does knowledge management influence innovation and competitiveness⁴, Journal *of Knowledge Management*, vol. 4, no. 2, pp. 87.

Aldrich, HE 1999, Organizations evolving, Sage Publications Limited.

Allen, TJ 1977, Managing the Flow of Technology, Cambridge MA: MIT Press.

Allen, TJ 1985, Managing the Flow of Technology: Technology Transfer and the Dissemination of Technological Information within the R&D Organization, Cambridge, MA: MIT Press.

Amabile, TM 1979 Effects of external evaluation on artistic creativity', *Journal* of Personality and Social Psychology, vol. 37, pp. 221-233.

Amabile, TM 1988, A model of creativity and innovation in organizations. In B.M. Staw & L. L. Cummings (Eds.), *Research in organizational behavior*, vol. 10, pp. 123-167. Greenwich, CT: JAI Press.

Amabile, TM 1993, Motivational synergy: Toward new conceptualizations of intrinsic and extrinsic motivation in the workplace', *Human Resource Management Review*, vol. 3, no. 3, pp. 185-201.

Amabile, TM 1996, Creativity in context: Update to 'the social psychology of creativity', Westview Press.

Amabile, TM 1997, Motivating creativity in organizations: On doing what you love and loving what you do', *California Management Review*, vol. 40, no. 1, pp. 39-58.

Amabile, TM, Conti, R, Coon, H, Lazenby, J & Herron, M 1996, Assessing the work environment for creativity', *Academy of Management Journal*, pp. 1154-1184.

Amabile, TM, Goldfarb, P & Brackfield, S 1990, _Social influences on creativity: Evaluations, coaction, and surveillance', *Creativity Research Journal*, vol. 34, pp. 6-21.

Amabile, TM & Gryskiewicz, ND 1989, _The creative environment scales: Work environment inventory⁴, *Creativity Research Journal*, vol. 2, no. 4, pp. 231252.

Amabile, TM, Hennessey, BA & Grossman, BS 1986, _Social influences on creativity: The effects of contracted-for reward', *Journal of Personality and Social Psychology*, vol. 50, no. 1, pp. 14.

Amabile, TM, Schatzel, EA, Moneta, GB & Kramer, SJ 2004, Leader behaviors and the work environment for creativity: Perceived leader support⁴, The *Leadership Quarterly*, vol. 15, no.1, pp. 5-32.

Åmo, BW 2006, _Employee innovation behaviour in health care: the influence from management and colleagues', *International Nursing Review*, vol. 53, no. 3, pp. 231-237.

Amo, BW 2010, _Corporate entrepreneurship and intrapreneurship related to innovation behaviour among employees', *International Journal of Entrepreneurial Venturing*, vol. 2, no. 2, pp. 144-158.

Åmo, BW & Kolvereid, L 2005, _Organizational strategy, individual personality and innovation behavior⁴, *Journal of Enterprising Culture*, vol. 13, no. 1, pp. 7-19.

Andersson, LM & Bateman, TS 1997, _Cynicism in the workplace: some causes and effects', *Journal of Organizational Behavior*, vol. 18, no. 5, pp. 449-469.

Anderson, N, De Dreu, CK & Nijstad, BA 2004, _The routinization of innovation research: A constructively critical review of the state-of-the-science', *Journal of organizational Behavior*, vol. 25, no. 2, pp. 147-173.

Anderson, N & King, N 1993, Innovation in organizations.

Anderson, N & West MA 1998, _Measuring Climate for Work Group Innovation: Development and Validation of the Team Climate Inventory', *Journal of Organizational Behavior*, vol. 19, no. 3, pp. 235-258.

Anderson, SB 1975, Encyclopedia of Educational Evaluation: Concepts and Techniques for Evaluating Education and Training Programs.

Andrews, FM 1979, *Scientific productivity*, Cambridge University Press, Cambridge, England.

Andriopoulos, C 2001, _Determinants of organisational creativity: A literature review', *Management Decision*, vol. 39, no. 10.

Anne, C & Greg RO, 1997, _Enhancing creativity: Managing work contexts for the high potential employee⁽, *California Management Review*, vol. 40, no. 1.

Armbruster, H, Bikfalvi, A, Kinkel, S & Lay, G 2008, Organizational innovation: The challenge of measuring non-technical innovation in large-scale surveys', *Technovation*, vol. 28, no. 10, pp. 644-657.

Atuahene-Gima, K, 2005, <u>Resolving the Capability: Rigidity Paradox in New</u> Product Innovation', *Journal of Marketing*, vol. 69, no. 4, pp. 61-83.

Aulakh, PS & Gencturk, EF 2000, _International principal-agent relationships control, governance and performance', *Industrial Marketing Management*, vol. 29, no.6, pp. 521–538.

Avolio, BJ, Yammarino, FJ & Bass, BM 1991, <u>Identifying common methods</u> variance with data collected from a single source: An unresolved sticky issue', *Journal of Management*, vol.17, no.3, pp. 571-587.

Axtell, CM, Holman, DJ & Wall, TD 2006, _Promoting innovation: A change study', *Journal of Occupational and Organisational Psychology*, vol. 79, pp. 509-516.

Axtell, CM, Holman, DJ, Unsworth, KL, Wall, TD, Waterson, PE & Harrington, E 2000, _Shopfloor innovation: Facilitating the suggestion and implementation of ideas', *Journal of occupational and organizational psychology*, vol.73, no. 3, pp. 265-285.

Babbie, E 2005, The basics of social research, Thomson Wadsworth, Belmont, CA.

Baer, M & Frese M. 2003, Innovation is not enough: climates for initiative and psychological safety, process innovations, and firm performance', *Journal of Organizational Behavior*, vol. 24, no.1, pp.45-68.

Bailey, DK 1987, <u>Mantle metasomatism</u>—perspective and prospect', *Geological Society, London, Special Publications*, vol. 30, no.1, pp. 1-13.

Barker, RA, 1997, <u>How can we train leaders if we do not know what leadership</u> is', *Human Relations*, vol. 50, no. 4, pp. 343-362.

Baron, RA & Tang, J 2011, _The role of entrepreneurs in firm-level innovation: Joint effects of positive affect, creativity, and environmental dynamism⁴, *Journal of Business Venturing*, vol. 26, no.1, pp. 49-60.

Barret, R 1997, *Liberating the corporate soul*⁴, *HR Focus*, vol. 74, no. 4, pp. 15-16.

Barron, F & Harrington, DM 1981, Creativity, intelligence, and personality', *Annual Review of Psychology*, vol. 32, no. 1, pp. 439-476.

Basadur, M 2004, Leading others to think innovatively together: Creative leadership', *The Leadership Quarterly*, vol. 15, no. 1, pp. 103-121.

Basadur, M, Graen, G & Green, S 1982, 'Training in creative problem solving: Effects on ideation and problem finding and solving in an industrial research organization', *Organizational Behavior and Human Decision Processes*, vol. 30, pp. 41-70.

Basadur, M, Wakabayashi, M & Graen, GB 1990, Individual problem-solving styles and attitudes toward divergent thinking before and after training', *Creativity Research Journal*, vol. 3, no.1, pp. 22-32.

Bass, BM, 1985, Leadership: Good, better, best', Organizational Dynamics; Organizational Dynamics.

Basu, R & Green, SG 1997, Leader-Member Exchange and Transformational Leadership: An Empirical Examination of Innovative Behaviors in Leader-Member Dyads', *Journal of Applied Social Psychology*, vol. 27, no. 6, pp. 477-499.

Beatty, RW & Schneier, CE 1997, _New HR roles to impact organizational performance: From -partnersl to -playersl', *Human Resource Management*, vol. 36, no.1, pp. 29-37.

Becheikh, N, Landry, R & Amara, N 2006, Lessons from innovation empirical studies in the manufacturing sector: A systematic review of the literature from 1993–2003^c, *Technovation*, vol. 26, no. 5, pp. 644-664.

Becker, KL & Matthews, JH, 2008, Linking HRM and innovation: formulating the research agenda.

Benson, GS, Finegold, D & Mohrman, SA 2004, 'You paid for the skills, now keep them: tuition-reimbursement and voluntary turnover', *Academy of Management Journal*, vol.47, no. 3, pp. 315-331.

Berneth, JB, Armenakis, AA, Feild, HS, Giles, WF & Walker, HJ 2007b, Leadermember social exchange (LMSX): Development and validation of a scale', *Journal of Organizational Behavior*, vol. 28, pp. 979-1003.

Beugelsdijk, S 2008, _Strategic Human Resource Practices and Product Innovation', *Organization Studies*, vol. 29, no. 6, pp. 821-847.

Bharadwaj, S & Menon A 2000, <u>Making Innovation Happen in Organizations</u>: Individual Creativity Mechanisms, Organizational Creativity Mechanisms or Both', *Journal of Product Innovation Management*, vol. 17, no. 6, pp. 424-434.

Binnewies, C, Ohly, S & Sonnentag, S 2007, <u>Taking personal initiative and</u> communicating about ideas: What is important for the creative process and for idea creativity', *European Journal of Work and Organizational Psychology*, vol.16, no. 4, pp. 432-455.

Bigoness, WJ & Perreault, WD 1981, _A Conceptual Paradigm and Approach for the Study of Innovators', *Academy of Management journal*, vol.24, no.1, pp. 68-82.

Birdi, K 2007, _A Lighthouse in the desert? Evaluating the effects of creativity training on employee innovation', *The journal of Creative Behaviour*, vol. 41, pp. 249-270.

Bishop, J 1987, _The Recognition and Reward of Employee Performance', *Journal of Labor Economics*, vol. 54, pp. S36-S56.

Block, Z & MacMillan, I 1993, *Corporate venturing: Creating new businesses with the firm*, Boston (Massachusetts).

Borg, WR & Gall, MD 1989, Educational Research, White Plains, NY: Longman.

Bottger, PC & Yetton, PW 1987, Improving group performance by training in individual problem solving', *Journal of Applied Psychology*, vol. 72, no. 4, pp. 651.

Bowen, DE & Ostroff, C 2004, _Understanding HRM-firm performance linkages: The role of the" strength" of the HRM system', *Academy of Management Review*, vol. 29, no. 2, pp. 203-221.

Brennan, A & Dooley, L 2005, <u>Networked creativity: a structured management</u> framework for stimulating innovation', *Technovation*, vol. 25, no. 12, pp.1388-1399.

Brown, JW & Utterback, JM 1985, _Uncertainty and technical communication patterns', *Management Science*, vol. 31, no. 3, pp. 301-311.

Brown, SL & Eisenhardt, KM 1995, _Product development: past research, present findings, and future direction', *Academy of Management Review*, vol. 20, pp. 343-78.

Bullinger, HJ, Auernhammer, K & Gomeringer, A 2004, <u>Managing innovation</u> networks in the knowledge-driven economy', *International Journal of Production Research*, vol. 42, no. 17, pp. 3337-3353.

Bunce, D & West, MA 1995, _Self perceptions and perceptions of group climate as predictors of individual innovation at work', *Applied Psychology*, vol. 44, no. 3, pp. 199-215.

Burrell, G & Morgan, G 1979, _Two dimensions: Four paradigms', *Sociological paradigms and organizational analysis*, Heinemann, pp. 21-40.

Byrne, CL, Mumford, MD, Barrett, JD, & Vessey, WB 2009, Examining the leaders of creative efforts: what do they do, and what do they think about?', *Creativity and Innovation Management*, vol. 18, no. 4, pp. 256-268.

Cabrera, EF & Bonache, J 1999, *An expert HR system for aligning organizational culture and strategy* (No. info: hdl: 10016/7157)⁴, Universidad Carlos III de Madrid.

Campbell, DT & Fiske, D 1959, Convergent and discriminant validation by the multitrait multimethod matrix', *Psychological Bulletin*, vol. 56, pp. 81-105.

Carmona-Lavado, A, Cuevas-Rodr guez, G & Cabello-Medina, C 2010, _Social and organizational capital: Building the context for innovation', *Industrial Marketing Management*, vol. 39, no. 4, pp. 681-690.

Carneiro, A 2000, _How does knowledge management influence innovation and competitiveness⁴, *Journal of knowledge management*, vol. 4, no. 2, pp. 87-98.

Carpiano, RM & Daley, DM 2006, _A guide and glossary on postpositivist theory building for population health', *Journal of Epidemiology and Community Health*, vol. 60, no. 7, pp. 564-570.

Cascio, WF 1990, Strategic human resource management in high technology industry', *Organizational Issues In High Technology Management*, vol. 11.

Cascio, W 1992, Managing human resources: Productivity, Quality of Work Life Profits, McGraw-Hill, New York.

Chandy, RK & Tellis, GJ 1998, _Organizing for radical product innovation: the overlooked role of willingness to cannibalize', *Journal of marketing Research*, pp. 474-487.

Chen, CJ & Huang, JW 2009, _Strategic human resource practices and innovation performance -The mediating role of knowledge management capacity', *Journal of Business Research*, vol. 62, no. 1, pp. 104-114.

Chen, MH & Kaufmann, G 2008, _Employee Creativity and R&D: A Critical Review⁴, *Creativity and Innovation Management*, vol. 17, no. 1, pp. 71-76.

Chi, G 2005, _A Study of Developing Loyalty Model', PhD Thesis, Oklahoma State University.

Choi, JN & Chang, JY 2009, Innovation implementation in the public sector: An integration of institutional and collective dynamics', *Journal of Applied Psychology*, vol. 94, no. 1, pp. 245.

Chow, IHS 1992, _Chinese Workers' Attitudes Towards Compensation Practices in the People's Republic of China', *Employee Relations*, vol. 14, no. 3, pp. 41.

Christensen, CM 1999, *Innovation and the General Manager*, Irwin/McGraw-Hill, Boston.

Christensen, CM & Rosenbloom, RS 1995, _Explaining the attacker's advantage: Technological paradigms, organizational dynamics, and the value network', *Research Policy*, vol. 24, no. 2, pp. 233-257.

Christensen, JF 2000, _Trajectories in Corporate Strategy, Organization and Management of R&D⁴, Paper presented at the *DRUID Summer Conference 2000*.

Clark, KB & Fujimoto, T 1991, _Product Development Performance: Strategy, Organization, and Management in the World Auto Industry', *Harvard Business School Press*, Boston, MA.

Clegg, C, Unsworth, K, Epitropaki, O & Parker, G 2002, Implicating trust in the innovation process', *Journal of Occupational and Organizational Psychology*, vol. 75, no. 4, pp. 409-422.

Coelho, F, Augusto, M & Lages, LF 2011, _Contextual factors and the creativity of frontline employees: The mediating effects of role stress and intrinsic motivation', *Journal of Retailing*, vol. 87, no. 1, pp. 31-45.

Cohen, WM & Levinthal, DA 1990, _Absorptive capacity: a new perspective on learning and innovation', *Administrative Science Quarterly*, pp. 128-152.

Conti, R, Amabile, TM & Pollak, S 1995, _The positive impact of creative activity: Effects of creative task engagement and motivational focus on college students' learning', *Personality and Social Psychology Bulletin*, vol. 21, no. 10, pp. 1107-1116.

Cooper, JR 1998, _A multidimensional approach to the adoption of innovation', *Management Decision*, vol. 36, no. 8, pp. 493-502.

Cooper, RG 1999, From experience: the invisible success factors in product innovation', *Journal of Product Innovation Management*, vol. 16, pp. 115–133.

Cooper, RG & Kleinschmidt, EJ 1986, _An investigation into the new product process: steps, deficiencies, and impact', *Journal of Product Innovation Management*', vol. 3, no. 2, pp. 71-85.

Cortina, JM 1993, _What is coefficient alpha? An examination of theory and applications', *Journal of applied psychology*, vol. 78, no. 1, pp. 98.

Costen, WM & Salazar, J 2011, _The impact of training and development on employee job satisfaction, loyalty, and intent to stay in the lodging industry', *Journal of Human Resources in Hospitality & Tourism*, vol. 10, no. 3, pp. 273-284.

Creswell, JW 1994, _Qualitative and quantitative approaches', *Qualitative and Quantitative Approaches*.

Creswell, JW & Clark, VLP 2007, *Designing and conducting mixed methods research*. Thousand Oaks, Sage Publications, CA.

Cummings, TG 1978, _Self-regulating work groups: A socio-technical synthesis', *Academy of Management Review*, pp. 625-634.

Cummings, TG & Srivastva, S 1977, *Management of work: A socio-technical systems approach*, University Associates.

Daft, L 1978, <u>A Dual-Core Model of Organisational Innovation</u>, *Academy of Management Journal*, vol. 21, no. 2, pp. 193-210.

Damanpour, F 1987, 'The Adoption of Technological, Administrative, and Ancillary Innovations: Impact of Organizational Factors', *Journal of Management*, vol. 13, pp. 675-688.

Damanpour, F 1991, Organizational Innovation: A Meta-Analysis of Effects of Determinants and Moderators', *Academy of Management Journal*, vol. 34, no. 3, pp. 555-590.

Damanpour, F 1992, Organizational Size and Innovation', *Organization Studies*, vol. 13, no. 3, pp. 375-402.

Damanpour, F 1996, Organizational complexity and innovation: developing and testing multiple contingency models', *Management Science*, vol. 42, no. 5, pp. 693-716.

Damanpour, F & Schneider, M 2006, _Phases of the adoption of innovation in organizations: Effects of environment, organization and top Managers1^c, *British Journal of Management*, vol. 17, no. 3, pp. 215-236.

Damanpour, F, Szabat, KA & Evan, WM 1989, _The relationship between types of innovation and organizational performance', *Journal of Management Studies*, vol. 26, no. 6, pp. 587-602.

Damanpour, F, Walker, RM & Avellaneda, CN 2009, _Combinative effects of innovation types and organizational performance: A longitudinal study of service organizations', *Journal of Management Studies*, vol. 46, no. 4, pp. 650-675.

Damanpour, R, Gopalakrishnan, S 2001, _The dynamics of the adoption of product and process innovations in organizations', *Journal of Management Studies*, vol. 38, no. 1, pp. 45-66.

Danish, RQ & Usman, A 2010, _Impact of Reward and Recognition on Job Satisfaction and Motivation: An Empirical Study from Pakistan', *International Journal of Business and Management*, vol. 5, no. 2, pp. 159.

Dansereau, F, Graen, G & Haga, WJ 1975, _A vertical dyad linkage approach to leadership within formal organizations: A longitudinal investigation of the role

making process', *Organizational Behavior and Human Performance*, vol. 13, no. 1, pp.46-78.

Davenport, TH 1993, _Need radical innovation and continuous improvement? Integrate process reengineering and TQM⁴, *Strategy & Leadership*, vol. 21, no. 3, pp. 6-12.

Davenport, T & Prusak, L 1998, *Working knowledge*, Harvard Business School Press, Boston.

Davis, D & Cosenza, RM 1985, *Business research for decision making*, Kent Publishing Company, Boston.

Davis, FD 1989, _Perceived usefulness, perceived ease of use, and user acceptance of information technology⁶, *MIS quarterly*, pp. 319-340.

Deci, EL & Ryan, RM, 1985, Intrinsic motivation and self-determination in human behaviour, Plenum, New York.

Deci, EL, Ryan, RM & Williams, GC 1996, <u>Need satisfaction and the self-regulation of learning</u>, *Learning and Individual Differences*, vol. 8, no. 3, pp. 165-183.

De Dreu, CKW & West, MA 2001, _Minority dissent and team innovation: The importance of participation in decision making', *Journal of Applied Psychology*, vol. 86, no. 6, pp. 1191-1201.

De Jong, JP 2007, Individual innovation: the connection between leadership and employees' innovative work behaviour, EIM.

De Jong, JP & Den Hartog, DN 2005, _Determinanten van Innovatief Gedrag: Een Onderzoek onder Kenniswerkers in het MKB (Determinants of innovative

behaviour: an investigation among knowledge workers in SMEs)', *Gedrag & Organisatie*, vol. 18, pp. 235-259.

De Jong, JP & Den Hartog, DN 2007, <u>How leaders influence employees</u> innovative behaviour', *European Journal of Innovation Management*, vol. 10, no. 1, pp.41-64.

De Jong, JP & Den Hartog, DN 2010, _Measuring innovative work behaviour', *Creativity and Innovation Management*, vol. 19, no. 1, pp. 23-36.

De Jong, JP & Kemp, R 2003, _Determinants of Co-Workers' Innovative Behaviour: An Investigation into Knowledge Intensive Services', *International Journal of Innovation Management*, vol. 7, no. 2, pp. 189.

De Leede, J & Looise, JK 2005, Innovation and HRM: Towards an Integrated Framework', *Creativity & Innovation Management*, vol. 14, no. 2, pp. 108-117.

Denti, L & Hemlin, S 2012, _Leadership And Innovation In Organizations: A Systematic Review Of Factors That Mediate Or Moderate The Relationship', *International Journal of Innovation Management*, vol. 16, no. 3.

Deshpande, R 1983, _Paradigms Lost: On Theory and Method in Research in Marketing⁴, *The Journal of Marketing*, pp. 101-110.

Dewar, RD & Dutton, JE 1986, _The Adoption of Radical and Incremental Innovations: An Empirical Analysis', *Management Science*, vol. 32, no. 11, pp: 1422-1433.

Diehl, M & Stroebe, W 1987, _Productivity loss in brainstorming groups: Toward the solution of a riddle', *Journal of Personality and Social Psychology*, vol. 53, no. 3, pp. 497.

Dienesch, RM & Liden, RC 1986, _Leader-member exchange model of leadership: A critique and further development', *Academy of Management Review*, pp. 618-634.

Djojosaputro, L, Nguyen, L & Peszynski, K 2005, _Cultural dimensions in online learning', in *Proceedings of the 16th Australia Conference of Information Systems*.

Dosi, G 1982, <u>Technological paradigms and technological trajectories</u>: A suggested interpretation of the determinants and directions of technical change', *Research Policy*, vol. 11, no. 3, pp. 147-162.

Drath, WH & Palus, CJ 1994, _Making common sense: Leadership as meaningmaking in a community of practice', *Center for Creative Leadership*.

Drazin, R, Glynn, MA & Kazanjian, RK 1999, _Multilevel Theorizing about Creativity in Organizations: A Sensemaking Perspective', *Academy of Management Review*, vol. 24, no. 2, pp. 286-307.

Drucker, PF 1985, _The discipline of innovation', *Harvard Business Review*, vol. 63, no. 3, pp. 67.

Dul, J & Ceylan C 2010, _Work environments for employee creativity', *Ergonomics*, vol. 54, no. 1, pp.12-20.

Dul, J, Ceylan, C & Jaspers, FPH 2009, 'Environmental effects on creative behavior and the role of person-environment fit', in Renaissance and Renewal in Management Studies EURAM 2009, 9th Annual Conference, University of Liverpool.

Dunegan, KJ, Tierney, P & Duchon, D 1992, _Perceptions of an innovative climate: Examining the role of divisional affiliation, work group interaction, and leader/subordinate exchange', *Engineering Management, IEEE Transactions on*, vol. 39, no. 3, pp. 227-236.

Ebadi, YM & Utterback, JM 1984, _The effects of communication on technological innovation', *Management Science*, vol. 30, no. 5, pp. 572-585.

Edwards, MR. 1989, _Measuring creativity at work: Developing a reward-forcreativity policy', *Journal of Creative Behaviour*, vol. 23, pp. 26-37.

Edwards, T, Delbridge, R & Munday, M 2005, _Understanding innovation in small and medium-sized enterprises: a process manifest⁴, *Technovation*, vol. 25, no.10, pp. 1119-1127.

Egan, TM 2005, _Factors influencing individual creativity in the workplace: An examination of quantitative empirical research', *Advances in Developing Human Resources*, vol. 7, pp. 160-181.

Eisenberger, R 1992, Learned industriousness', *Psychological Review*, vol. 99, pp. 248-267.

Eisenberger, R & Armeli, S 1997, <u>Can salient reward increase creative</u> performance without reducing intrinsic creative interest', *Journal of Personality and Social Psychology*, vol. 72, pp. 652-663.

Eisenberger, R, Armeli, S & Pretz, J 1998, <u>Can the promise of reward increase</u> creativity', *Journal of Personality and Social Psychology*, vol. 74, no. 3, pp. 704.

Eisenberger, R & Cameron, J. 1996, _Detrimental effects of reward: Reality or myth', *American Psychologist*, vol. 51, no. 11, pp. 1153-1166.

Eisenberger, R, Haskins, F & Gambleton, P 1999, _Promised reward and creativity: Effects of prior experience', *Journal of Experimental Social Psychology*, vol. 35, pp. 308-325.

Eisenberger, R, Pierce, WD & Cameron, J 1999, Effects of reward on intrinsic motivation-Negative, neutral, and positive: Comment on Deci, Koestner, and Ryan (1999)⁴, pp. 677.

Ekvall, G 1971, *Creativity at the place of work*, Swedish Council for Personnel Administration.

Ericsson, KA & Charness, N 1994, Expert performance: Its structure and acquisition', *American Psychologist*, vol. 49, pp. 725-747.

Ettlie, JE 1983, Organisational Policy and Innovation Among Suppliers to the Food Processing Sector', *Academy of Management Journal*, vol. 26, no. 1, pp. 27-44.

Ettlie, JE, Bridges, WP & O'Keefe, RD 1984, _Organization strategy and structural differences for radical versus incremental innovation', *Management Science*, vol. 30, pp. 682-695.

Ettlie, JE & Reza, EM 1992, _Organizational integration and process innovation', *Academy of Management Journal*, vol. 35, no. 4, pp. 795-827.

Evan, WM & Black, G 1967, Innovation in business organizations: some factors associated with success or failure of staff proposals', *The Journal of Business*, vol. 40, no. 4, pp. 519-530.

Fagerberg, J & Verspagen, B 2009, <u>Innovation studies-The emerging structure of</u> a new scientific field', *Research Policy*, vol. 38, no. 2, pp. 218-233.

Farmer, SM, Tierney, P & Kung-McIntyre, K 2003, _Employee Creativity in Taiwan: An Application of Role Identity Theory', *Academy of Management Journal*, vol. 46, no. 5, pp. 618-630.

Feist, GJ 1999, _14 The Influence of Personality on Artistic and Scientific Creativity', *Handbook of creativity*, pp. 273.

Feldhusen, JF & Goh, BE 1995, _Assessing and accessing creativity-An integrative review of theory, research, and development', *Creativity Research Journal*, vol. 8, pp. 231-247.

Fernandez, S & Pitts, DW 2011, _Understanding Employee Motivation to Innovate: Evidence from Front Line Employees in United States Federal Agencies', *Australian Journal of Public Administration*, vol. 70, no. 2, pp. 202-222.

Fidler, L & Johnson, J 1984, Communication and innovation implementation', *Academy of Management Review*, vol. 9, pp. 704-711.

Fiedler, FE 1996, _Research on leadership selection and training: One view of the future', *Administrative Science Quarterly*, pp. 241-250.

Filipczak, B 1997, _It takes all kinds: Creativity in the work force', *Training*, vol. 34, no. 5, pp. 32-40.

Fischer, MM 2001, Innovation, knowledge creation and systems of innovation', *Annals of Regional Science*, vol. 35, pp. 199-216.

Fiske, DW 1982, <u>Convergent-discriminant</u> validation in measurements and research strategies', in D. Brinbirg & LH. Kidder (ed.), *Forms of validity in research*, Jossey-Bass, San Francisco, pp. 77-92.

Fitzgerald, W 1992, _Training versus development', *Training and Development Journal*, vol. 5, pp.81-84.

Fix, B & Sias, PM 2006, _Person-centered communication, leader-member exchange, and employee job satisfaction⁶, *Communication Research Report*, vol. 23, no. 1, pp. 35-44.

Flor, ML & Oltra, MJ 2004, _Identification of innovating firms through technological innovation indicators: an application to the Spanish ceramic tile industry', *Research Policy*, vol. 33, no. 2, pp. 323-336.

Ford, CM 1996, _A theory of individual creative action in multiple social domains', *Academy of Management review*, pp. 1112-1142.

Ford, C & Gioia, D 2000, 'Factors influencing creativity in the domain of managerial decision making', *Journal of Management*, vol. 26, no. 4, pp. 705-732.

Fornell, C & Larcker, DF 1981, _Evaluating structural equation models with unobservable variables and measurement error', *Journal of Marketing Research*, pp. 39-50.

Frambach, RT & Schillewaert, N 2002, _Organizational innovation adoption: a multi-level framework of determinants and opportunities for future research', *Journal of Business Research*, vol. 55, no. 2, pp. 163-176.

Frank, AD & Brownell, J 1989, Organizational Communication and Behavior: Communicating to Improve Performance, 2+2, Holt, Rinehart & Winston, New York.

Frascati Manual 2004, _A summary of the Frascati manual. Main definitions and conventions for the measurement of research and experimental development (R&D)⁴, OCDE/GD (94) 84.

Freel, MS 2005, _Patterns of innovation and skills in small firms', *Technovation*, vol. 25, no. 2, pp. 123-134.

Freeman, C & Perez, C 1988, *Structural crises of adjustment, business cycles and investment behaviour*, Pinter.

Frohman, M & Pascarella, P 1990, _Achieving purpose-driven innovation', *Industry Week*, vol. 239, no. 6, pp. 20-26.

Fuglsang, L & Sørensen, F 2011, _The balance between bricolage and innovation: Management dilemmas in sustainable public innovation', *The Service Industries Journal*, vol. 31, no. 4, pp. 581-595.

Galbraith, JR 1982, _Designing the innovating organization⁴, *Organizational Dynamics*, vol. 10 (winter), pp. 5-25.

Garcia, R & Calantone, R 2002, _A critical look at technological innovation typology and innovativeness terminology: A literature review', *Journal of Product Innovation Management*, vol. 19, no. 2.

Genaidy, AM, Rinder, MM, Sequeira, R & Amal, A 2010, _The role of human at work systems in business sustainability: perspectives based on expert and qualified production workers in a manufacturing enterprise', *Ergonomics*, vol. 53, no. 4, pp. 559-585.

George, JM 2007, _Creativity in organizations', *Academy of Management Annals*, vol.1, no. 1, pp. 439-477.

George, JM & Zhou, J 2001, _When openness to experience and conscientiousness are related to creative behaviour: An interactional approach', *Journal of Applied Psychology*, vol. 86, pp. 513-524.

George, JM & Zhou, J 2002, _Understanding when bad moods foster creativity and good ones don't: The role of context and clarity of feelings', *Journal of Applied Psychology*, vol. 87, pp. 687-697. Georgsdottir, AS & Getz, I 2004, _How Flexibility Facilitates Innovation and Ways to Manage It in Organizations', *Creativity and Innovation Management*, vol. 13, pp. 166-75.

Gerstner, CR & Day, DV 1997, _Meta-analytic review of leader-member exchange theory: Correlates and construct issues', *Journal of Applied Psychology*, vol. 82, no. 6, pp. 827-844.

Gilson, LL 2001, _Diversity, dissimilarity and creativity: Does group composition or being different enhance or hinder creative performance', in *Academy of Management Meetings Washington DC*.

Gilson, LL & Shalley, CE 2004, _A Little Creativity Goes a Long Way: An Examination of Teams æTM Engagement in Creative Processes', *Journal of Management*, vol. 30, no. 4, pp. 453-470.

Gilson, LL & Madjar, N 2011, _Radical and incremental creativity: Antecedents and processes', *Psychology of Aesthetics, Creativity, and the Arts*, vol. 5, no. 1, pp. 21.

Girma, S, Gong, Y & <u>Görg</u> H 2009, _What Determines Innovation Activity in Chinese State-owned Enterprises? The Role of Foreign Direct Investment', *World Development*, vol. 37, no. 4, pp. 866-873.

Gopalakrishnan, S & Damanpour, F 1997, _A review of innovation research in economics, sociology and technology management⁴, *Omega*, vol. 25, no.1, pp. 15-28.

Graen, G 1976, _Role-making processes within complex organizations', *Handbook of industrial and organizational psychology, Rand McNally, Chicago.*

Graen, GB 2006, In the eye of the beholder: Cross-cultural lesson in leadership from project GLOBE: A response viewed from the third culture bonding (TCB)

model of cross-cultural leadership', *The Academy of Management Perspectives*, vol. 20, no. 4, pp. 95-101.

Graen, GB & Cashman, J 1975, _A role-making model of leadership in formal organizations: A developmental approach⁴, in J. G. Hunt & L. L. Larson (ed.), *Leadership Frontiers*, Kent, OH: Kent State University Press, pp. 143-166.

Graen, GB, Hui, C & Taylor, ET 2004, <u>A</u> new approach to team leadership: Upward, downward, and horizontal differentiation⁻, in G. B. Graen (ed.), *New frontiers of leadership*, Greenwich: IAP, pp. 33-66.

Graen GB, Liden R & Hoel W 1982a, _Role of leadership in the employee withdrawal process', *Journal of Applied Psychology*, vol. 67, pp. 868-872.

Graen GB, Novak MA & Sommerkamp P 1982b, _The effects of leader member exchange and job design on productivity and satisfaction: Testing the dual attachment model⁴, *Organizational Behavior and Human Performance*, vol. 30, no. 1, pp. 109-131.

Graen GB, Schiemann W, 1978, Leader-member agreement: A vertical dyad linkage approach', *Journal of Applied Psychology*, vol. 63, no. 2, pp. 206-212.

Graen, GB & Uhl-Bien M 1995, _Relationship-based approach to leadership: Development of leader-member exchange (LMX) theory of leadership over 25 years: Applying a multi-level multi-domain perspective', *The Leadership Quarterly*, vol. 6, no. 2, pp. 219-247.

Gross, SE & Friedman, HM 2004, _Creating an effective total reward strategy: holistic approach better supports business success', *Benefits Quarterly*, vol. 20, no. 3, pp. 7-12.

Gupta, AK & Singhal, A 1993, <u>Managing human resources for innovation and</u> creativity', *Research Technology Management*, vol. 36, pp. 41-41.

Hafiza, N, Shah, S, Jamsheed, H & Zaman, K 2011, _Relationship between rewards and employee motivation in the non-profit organization of Pakistan', *Business Intelligence Journal*, vol. 4, no. 2, pp. 327-334.

Hair, JF, Anderson, RE, Tatham, RL & Black, WC, 1998, *Multivariate Data Analysis*, 5th edn, Prentice-Hall International, Englewood Cliffs, NJ.

Hameed, A & Amjad, S 2011, _Students' Satisfaction in Higher Learning Institutions: A Case Study of COMSATS Abbottabad, Pakistan', *Iranian Journal Management Studies*, vol. 4, pp. 63-77.

Han, JK, Kim, N & Srivastava, RK 1998, <u>Market Orientation and Organisational</u> Performance: Is Innovation a Missing Link', *Journal of Marketing*, vol. 62, no. 4, pp. 30-45.

Harari, O 1994. _When intelligence rules, the manager's job changes', *Management Review*, vol. 83, no. 7, pp. 33-35.

Hargadon, A 1998, Firms as knowledge brokers', *California Management Review*, vol. 40, no. 3, pp. 209-227.

Harrison, DA & Klein, KJ 2007, _What's the difference? Diversity constructs as separation, variety, or disparity in organizations', *Academy of Management Review*, vol. 32, no. 4, pp. 1199-1228.

Hartley, J 2005. *Creative Industries*. Oxford: Blackwell, Tsinghua University Press, translated by Cao Shule, Li Hui and Bao Jiannu.

Hatch, NW & Mowery, DC 1998, _Process innovation and learning by doing in semiconductor manufacturing', *Management Science*, vol. 44 (11-Part-1), pp. 1461-1477.

Hatcher, L, Ross, TL & Collins, D 1989, Prosocial behavior, job complexity, and suggestion contribution under gainsharing plans', *The Journal of Applied Behavioral Science*, vol. 25, no. 3, pp. 231-248.

Hemlin, S & Olsson, L 2011, _Creativity-Stimulating Leadership: A Critical Incident Study of Leaders' Influence on Creativity in Research Groups', *Creativity and Innovation Management*, vol. 20, no. 1, pp. 49-58.

Hemphälä, J & Magnusson, M 2012, _Networks for Innovation–But What Networks and What Innovation', *Creativity and Innovation Management*, vol. 21, no. 1, pp. 3-16.

Henderson, RM & Clark, KB 1990, _Architectural innovation: The reconfiguration of existing product technologies and the failure of established firms', *Administrative Science Quarterly*, pp. 9-30.

Hennessey, BA & Amabile, TM 1998b, _Reward, intrinsic motivation, and creativity', *American Psychologist*, vol. 53, pp. 674-675.

Hesse-Biber, SN & Leavy, P 2006, In-depth interview', *The practice of qualitative research*.

Heyes, J & Stuart, M 2007, _Does training matter', *Human Resource Management Journal*, vol. 6, no. 3, pp. 7-21.

Hirst, G, Van Dick, R & Van Knippenberg, D 2009, _A social identity perspective on leadership and employee creativity', *Journal of Organizational Behavior*, vol. 30, no. 7, pp. 963-982.

Hirst, G, Van Knippenberg, D & Zhou, J 2009, _A cross-level perspective on employee creativity: Goal orientation, team learning behavior, and individual creativity', *Academy of Management Journal*, vol. 52, no. 2, pp. 280-293.

Holman, D, Totterdell, P, Axtell, C, Stride, C, Port, R, Svensson, R & Zibarras, L 2011. _Job Design and the Employee Innovation Process: The Mediating Role of Learning Strategies', *Journal of Business and Psychology*, vol. 27, pp. 177-191.

Hon, AH, _When competency-based pay relates to creative performance: The moderating role of employee psychological need', *International Journal of Hospitality Management*, vol. 31, no. 1, pp. 130-138.

Howard, H 1994, Why do people say nasty things about self-reports', *Journal of Organizational Behavior*, vol. 15, no. 5, pp. 399-404.

Howell, JM & Boies, K 2004, Champions of technological innovation: The influence of contextual knowledge, role orientation, idea generation, and idea promotion on champion emergence', *The Leadership Quarterly*, vol. 15, no. 1, pp. 123-143.

Hoyle, RH, Harris, MJ & Judd, CM 2002, *Research methods in social relations*, Fort Worth, TX: Wadsworth.

Hsu, IC 2006, _Enhancing employee tendencies to share knowledge--Case studies of nine companies in Taiwan', *International Journal of Information Management*, vol. 26, no. 4, pp. 326-338.

Huang, X., Rode, JC & Schroeder, RG 2011, _Organizational structure and continuous improvement and learning: Moderating effects of cultural endorsement of participative leadership', *Journal of International Business Studies*, vol. 42, no. 9, pp. 1103-1120.

Huhtala, H & Parzefall, MR 2007, _A Review of Employee Well-Being and Innovativeness: An Opportunity for a Mutual Benefit', *Creativity & Innovation Management*, vol. 16, no. 3, pp. 299-306. Hunter, ST, Bedell, KE & Mumford, MD 2007, Climate for creativity: A quantitative review', *Creativity Research Journal*, vol. 19, no.1, pp. 69-90.

Hunter, ST & Cushenbery, L 2011, Leading for Innovation Direct and Indirect Influences', *Advances in Developing Human Resources*, vol. 13, no. 3, pp. 248-265.

Hurley, RF & Hult, GTM 1998, Innovation, market orientation, and organizational learning: an integration and empirical examination', *The Journal of Marketing*, pp. 42-54.

Hurmelinna-Laukkanen, P, Sainio, LM & Jauhiainen, T 2008, Appropriability regime for radical and incremental innovations', *R&D Management*, vol. 38, no. 3, pp. 278-289.

Iarossi, G 2006, *The power of survey design: A user's guide for managing surveys, interpreting results, and influencing respondents*, World Bank Publications.

Im, HJ 2003, <u>An Exploratory Study of Branding for the State of Oklahoma'</u>, PhD Thesis, Oklahoma State University.

Isaksen, S & Tidd, J 2006, *Meeting the innovation challenge: Leadership for transformation and growth*, Wiley.

Jackson, T & Bak, M 1998, _Foreign companies and Chinese workers: employee motivation in the People's Republic of China', *Journal of Organisational Change Management*, vol. 11, no. 4, pp. 282-300.

Jacobs, TO 1970, *Leadership and exchange in formal organisations*, Human Resources Research Organisation, Alexandria, VA.

Janis, IL 1972, Victims of groupthink: A psychological study of foreign-policy decisions and fiascos, Houghton Mifflin Company, Bosten MA.

Jansen, JJ, Vera, D & Crossan, M 2009, _Strategic leadership for exploration and exploitation: The moderating role of environmental dynamism⁴, *The Leadership Quarterly*, vol. 20, no. 1, pp. 5-18.

Jaskyte, K & Kisieliene, A 2006, _Determinants of Employee Creativity: A Survey of Lithuanian Nonprofit Organizations', *International Journal of Voluntary and Nonprofit Organisations*, vol. 17, no. 2, pp. 133-141.

Janssen, O 2000, _Job demands, perceptions of effort-reward fairness and innovative work behaviour', *Journal of Occupational and Organizational Psychology*, vol. 73, no. 3, pp. 287-302.

Janssen, O 2005, _The joint impact of perceived influence and supervisor supportiveness on employee innovative behaviour', *Journal of Occupational and Organizational Psychology*, vol.78, no. 4, pp. 573-579.

Janssen, O, Van de Vliert, E & West, M 2004, -The bright and dark sides of individual and group innovation: a special issue introduction, *Journal of Organizational Behavior*, vol. 25, no. 2, pp. 129-45.

Janssen, O & Van Yperen, NW 2004, _Employees' goal orientation, the quality of leader-member exchange, and the outcome of job performance and job satisfaction', *Academy of Management Journal*, vol. 47, no. 3, pp. 368-384.

Jaw, B-S & Liu, W 2003, _Promoting organizational learning and self-renewal in Taiwanese companies: The role of HRM⁶, *Human Resource Management*, vol. 42, no. 3, pp. 223-241.

Jiang, J, Wang, S & Zhao, S 2012, _Does HRM facilitate employee creativity and organizational innovation? A study of Chinese firms', *The International Journal of Human Resource Management*, vol. 23, no. 19, pp. 4025-4047.

Johannessen, JA & Olsen, B 2003, <u>Knowledge management and sustainable</u> competitive advantages: The impact of dynamic contextual training', *International Journal of Information Management*, vol. 23, no. 4, pp. 277-289.

Kaiser, HF 1974, <u>An index of factorial simplicity</u>, *Psychometrika*, vol. 39, pp. 31-36.

Kanter, RM 1982, _The middle manager as innovator', *Harvard Business Review*, vol. 60, no. 4, pp. 95.

Kanter, RM 1983, The Change Masters, Simon & Schuster, New York.

Kanter, RM 1984, The Change Masters, Simon & Schuster, New York.

Kanter, RM 1988, _Three tiers for innovation research', *Communication Research*, vol. 15, no. 5, pp. 509-523.

Kanter, RM 1988, _When a Thousand Flowers Bloom: Structural, Collective & Social Conditions for Innovation in Organizations', *Knowledge Management and Organizational Design*, pp. 93.

Kasperson, CJ 1978, _An analysis of the relationship between information sources and creativity in scientists and engineers', *Human Communication Research*, vol. 4, no. 2, pp. 113-119.

Katz, R 1982, _The effects of group longevity on project communication and performance', *Administrative Science Quarterly*, pp. 81-104.

Katz, R & Tushman, M 1981, _An investigation into the managerial roles and career paths of gatekeepers and project supervisors in a major R & D facility', *R&D Management*, vol. 11, no. 3, pp. 103-110.

Kay, J. 1993, Foundations of Corporate Success, Oxford University Press, Oxford.

Kaynak, H 1997, Total quality management and just-in-time purchasing: their effects on performance of firms operating in the U.S., Garland, New York.

Keller, RT 1986, _Predictors of the performance of project groups in r & organizations', *Academy of Management Journal*, vol.29, no. 4, pp. 715-726.

Keller, RT & Holland, WE 1983, <u>Communicators and innovators in research and</u> development organizations', *Academy of Management Journal*, vol. 26, no. 4, pp. 742-749.

Kelloway, EK 1998, Using LISREL for Structural Equation Modeling a Researcher's Guide.

Kemery, ER & Dunlap, WP 1986, _Partialling factor scores does not control method variance: A rejoinder to Podsakoff and Todor', *Journal of Management*, vol. 12, pp. 525-530.

Kimberly, JR & Evanisko, MJ 1981, _Organizational innovation: The influence of individual, organizational, and contextual factors on hospital adoption of technological and administrative innovations', *Academy of Management Journal*, pp. 689-713.

King, N. 1990, Innovation at work: the research literature', in MA West & JL Farr (ed.), *Innovation and Creativity at Work: Psychological and Organisational Strategies*, Wiley, Chichester, pp. 15-60.

King, N 2004, _Using Templates in the thematic analysis of text⁴, in C Cassell & G Symon (ed.), *Essential Guide to Qualitative Methods in Organizational Researc*, Sage, London.

King, N, Anderson, NR & West, MA 1992, _Organizational innovation: A case study of perceptions and processes', *Work and Stress*, vol. 5, pp. 331–339.

Kivim äki, M, Länsisalmi, H, Elovainio, M, Heikkilä, A, Lindström, K, Harisalo, R & Puolimatka, L 2002, <u>Communication as a determinant of organizational innovation</u>, *R&D Management*, vol. 30, no. 1, pp. 33-42.

Klein, KJ & Sorra JS 1996, _The Challenge of Innovation Implementation', *Academy of Management Review*, vol. 21, no. 4, pp. 1055-1080.

Kleysen, RF & Street, CT 2001, _Toward a multi-dimensional measure of individual innovative behavior', *Journal of Intellectual Capital*, vol. 2, no. 3, pp. 284-296.

Kline, RB 2005, *Principles and practice of structural equation modelling*, 2nd edn, Guilford Press, New York.

Kline, SJ & Rosenberg, N 1986, <u>Innovation:</u> an overview', *The Positive Sum Strategy: Harnessing Technology for Economic Growth*, National Academic Press, Washington DC, pp. 275-307.

Knight, KE 1967, _A descriptive model of the intra-firm innovation process', *The Journal of Business*, vol. 40, no. 4, pp. 478-496.

Koberg, CS, Detienne, DR & Heppard, KA 2003, _An empirical test of environmental, organizational, and process factors affecting incremental and radical innovation', *The Journal of High Technology Management Research*, vol. 14, no. 1, pp. 21-45.

Kozlowski, SW 1987, _Technological innovation and strategic HRM: Facing the challenge of change', *Human Resource Planning*, vol. 10, no. 2, pp. 69-79.

Kratzer, J 2001, 'Communication and Performance: An empirical study in innovation teams', Thesis Publishers, Amsterdam.

Kratzer, J, Leenders, OTA & Engelen, JMV 2004, _Stimulating the potential: Creative performance and communication in innovation teams', *Creativity and Innovation Management*, vol.13, no. 1, pp. 63-71.

Lansisalmi, H, Peiro, JM & Kivimaki, M 2000, 'Collective Stress and Coping in the Context of Organizational Culture', *European Journal of Work and Organizational Psychology*, vol. 9, no. 4, pp. 527-559.

Latham, G. P., & Huber, V. L. (1991). Schedules of Reinforcement. *Journal of Organizational Behavior Management*, vol. 12, no. 1, pp. 125-149.

Lau, CM & Ngo, HY 2004, _The HR system, organizational culture, and product innovation', *International Business Review*, vol.13, no. 6, pp. 685-703.

Laursen, K & Foss, NJ 2003, _New human resource management practices, complementarities and the impact on innovation performance', *Cambridge Journal of Economics*, vol. 27, no. 2, pp. 243-263.

Lawler, EE. III 1969, Job design and employee motivation', *Personnel Psychology*, vol. 22, pp. 426-435.

Lawler, EE. III 1994, _Total quality management and employee involvement: Are they compatible', *Academy of Management Executive*, vol. 8, no. 1, pp. 68-76.

Lawson, B & Samson, D 2001, _Developing innovation capability in organisations: a dynamic capabilities approach', *International Journal of Innovation Management*, vol. 5, no. 3, pp. 377-400.

Leavitt, H 1975, _Beyond the analytic manager', *California Management Review*, vol. 17, no. 4, pp.11.

Lee, J 2001, <u>Leader-member exchange</u>, perceived organizational justice, and cooperative communication', *Management Communication Quarterly*, vol. 14, no. 4, pp. 574-589.

Lee, SH & Wong, PK 2004, _An exploratory study of technopreneurial intentions: A career anchor perspective', *Journal of Business Venturing*, vol. 19, no. 1, pp. 7-28.

Leonard-Barton, D 1992, _Core Capabilities and Core Rigidities: A Paradox in Managing New Product Development', *Strategic Management Journal*, vol. 13, pp. 111-125.

Leonard, D & Sensiper, S 1998, _The role of tacit knowledge in group innovation', *California Management Review*, vol. 40, no. 3.

Leppalahti, A & Akerblom, M 1991, Industrial innovation in Finland: an empirical study.

Lepper, MR & Greene, DE 1978, *The hidden costs of reward: New perspectives on the psychology of human motivation*, Lawrence Erlbaum.

Licuanan, BF, Dailey, LR & Mumford, MD 2007, <u>Idea evaluation</u>: Error in evaluating highly original ideas', *The Journal of Creative Behavior*, vol. 41, no. 1, pp. 1-27.

Liden, RC, Wayne, SJ & Sparrowe, RT 2000, _An examination of the mediating role of psychological empowerment on the relations between the job, interpersonal relationships, and work outcomes', *Journal of Applied Psychology*, vol. 85, no. 3, pp. 407.

Liden, RC, Wayne, SJ & Stilwell, D 1993, A longitudinal study on the early development of leader-member exchanges', *Journal of applied psychology*, vol. 78, no. 4, pp. 662-674

Lin, WB 2008, _The effect of knowledge sharing model', *Expert Systems With Applications*, vol. 34, no. 2, pp. 1508-1521.

Lindell, MK & Whitney, DJ 2001, _Accounting for common method variance in cross-sectional research designs', *Journal of Applied Psychology*, vol. 86, no. 1, pp.114.

Liu, Z, Li, J, Zhu, H, Cai, Z & Wang, L 2012, _Chinese firms' sustainable development-The role of future orientation, environmental commitment, and employee training', *Asia Pacific Journal of Management*, pp. 1-19.

Lu, XJ & Li P 2010, '*The Impact of Learning Culture on Individual Innovative Behavior*', in Management and Service Science (MASS), 2010 International Conference on, IEEE, pp. 1-4.

Lu, XJ & Zhang, GL 2007, _The Relationship between Work Motivation and Individual Innovation Behavior', *Soft Science*, vol. 21, no. 32, pp. 124-127.

Lu, Y, Tsang, EW & Peng, MW 2008, _Knowledge management and innovation strategy in the Asia Pacific: Toward an institution-based view', *Asia Pacific Journal of Management*, vol. 25, no. 3, pp. 361-374.

Mabey, C & Salaman, G 1995, *Strategic Human Resource Management*, Oxford: Blackwell, London.

Madjar, N 2008, _Emotional and informational support from different sources and employee creativity', *Journal of Occupational and Organizational Psychology*, vol. 81, no. 1, pp. 83-100.

Madjar, N, Greenberg, E & Chen, Z 2011, _Factors for radical creativity, incremental creativity, and routine, noncreative performance', *Journal of Applied Psychology*, vol. 96, no. 4, pp. 730.

Madjar, N, Oldham, GR & Pratt, MG 2002, _There's no place like home? The contributions of work and nonwork creativity support to employees' creative performance', *Academy of Management Journal*, vol. 45, pp. 757-767.

Magadley, W & Birdi, K 2012, <u>Two Sides Of The Innovation Coin? An Emprical</u> Investigation of The Relative Correlates of Idea Generation And Idea Implementation⁴, *International Journal of Innovation Management*, vol.16, no. 1.

Makela, L 2005, _Pregnancy and Leader-Follower Dyadic Relationships: A research Agenda⁴, *Equal Opportunities International*, vol. 24, no. 3/4, pp.50-72.

Mallinckrodt, B & Wang, CC 2004, _Quantitative methods for verifying semantic equivalence of translated research instruments: A Chinese version of the experience in close relationships scale', *Journal of Counseling Psychology*, vol. 51, no. 3, pp. 368-379.

Maltzman, I 1960, On the training of originality', *Psychological Review*, vol. 67, no. 4, pp. 229.

Mansfield, RS, Busse, TV & Krepelka, EJ 1978, _The effectiveness of creativity training', *Review of Educational Research*, pp. 517-536.

Manual, F 2004, _A summary of the Frascati manual. Main definitions and conventions for the measurement of research and experimental development (R&D)⁴, *OCDE/GD* (94).

March, JG 1991, _Exploration and Exploitation in Organizational Learning', *Organization Science*, vol. 2, no. 1, pp. 71-87.

Martin Jr, CR & Horne, DA 1995, <u>Level of success inputs for service innovations</u> in the same firm', *International Journal of Service Industry Management*, vol. 6, no. 4, pp. 40-56. Martindale, C1989, *Personality, situation, and creativity*, US: Plenum Press, New York.

Martins, EC & Terblanche, F 2003, <u>Building</u> organisational culture that stimulates creativity and innovation', *European Journal of Innovation Management*, vol. 6, no. 1, pp. 64.

McAdam, R & Keogh, W 2004, _Transitioning Towards Creativity and Innovation Measurement in SMEs⁴, *Creativity and Innovation Management*, vol. 13, no. 2, pp.126-139.

McClelland, DC 1987, *Human Motivation*, Cambridge University Press, Cambridge.

McDermott, CM & O'Connor, GC 2002, <u>Managing radical innovation</u>: An overview of emergent strategy issues', *Journal of Product Innovation Management*, vol. 19, no. 6.

McGinnis, MA & Ackelsberg, MR 1983, _Effective innovation management: Missing link in strategic planning', *Journal of Business Strategy*, vol. 4, no. 1, pp. 59-66.

McLean, LD 2005, Organizational Culture's Influence on Creativity and Innovation: A Review of the Literature and Implications for Human Resource Development', *Advances in Developing Human Resources*, vol. 7, no. 2, pp. 226-246.

McLoughlin, I & Harris, M 1997, _Understanding innovation, organizational change and technology⁴, Chapter Introduction, in Mcloughlin, I & Harris, M (ed), *Innovation, Organization Change and Technology*, International Thomson Business Press, pp. 1-26.

Milkovich, GT & Newman JM 2005, *Compensation*, 8th edn, McGraw Hill Iriwn, Boston

Miller, D & Friesen, PH 1982, _Configuration of strategy and structure: towards a synthesis', *Strategic Management Journal*, vol 7, pp. 233-250.

Ming-Huei, C & Kaufmann, G 2008, _Employee Creativity and R&D: A Critical Review⁴, *Creativity & Innovation Management*, vol. 17, no. 1, pp. 71-76.

Mireille, M-C & Wim, JN 2005, _Factors influencing knowledge creation and innovation in an organisation', *Journal of European Industrial Training*, vol. 29, no. 2/3, pp. 135.

Monge, PR, Cozzens, MD & Contractor, NS 1992, _Communication and motivational predictors of the dynamics of organizational innovation', *Organization Science*, vol. 3, no. 2, pp. 250-274.

Montes, FJL, Moreno, AR & Fernandez, LMM 2004, _Assessing the organizational climate and contractual relationship for perceptions of support for innovation', *International Journal of Manpower*, vol. 25, no. 2, pp. 167-180.

Mumford. MD 2000, _Managing creative people: strategies and tactics for innovation', *Human Resource Management Review*, vol. 10, pp. 313-351.

Mumford, MD & Gustafson, SB 1988, _Creativity syndrome: integration, application, and innovation', *Psychological Bulleting*, vol. 103, pp. 27-43.

Mumford, MD & Hunter, ST 2005, <u>Innovation in organizations</u>: A multi-level perspective on creativity', *Multi-Level Issues in Strategy and Methods*, Emerald Group Publishing Limited, Bingley, pp. 9-73.
Mumford, MD, Scott, GM, Gaddis, B & Strange, JM 2002, Leading creative people: Orchestrating expertise and relationships', *Leadership Quarterly*, vol.13, pp. 705–750.

Muñoz-Doyague, MF & Nieto, M 2012, _Individual creativity performance and the quality of interpersonal relationships', *Industrial Management & Data Systems*, vol. 112, no. 1, pp. 125-145.

Nadler, D. A., & Tushman, M. L. (1980). A model for diagnosing organizational behavior. *Organizational Dynamics*, 9(2), 35-51.

Nadkari, S & Herrmann POL 2010, <u>CEO</u> Personality, Strategic Flexibility, and Firm Performance: The Case of the Indian Business Process Outsourcing Industry', *Academy of Management Journal*, vol. 53, pp. 1050-1073.

Nelson, R 2000, _National innovation systems', *Regional Innovation, Knowledge and Global Change*, Pinter: London.

Neuman, WL 2003, Social research methods: Qualitative and quantitative approaches, Allyn & Bacon, Boston.

Neumann, WP & Dul, J 2010, _Human factors: spanning the gap between OM & HRM⁴, *International Journal of Operations and Production Management*, vol. 30, pp. 923-950.

Nickerson, RS 1999, _How we know—and sometimes misjudge—what others know: Imputing one's own knowledge to others', *Psychological Bulletin*, vol. 125, no. 6, pp. 737.

Nijstad, BA 2000, How the group affects the mind: effects of communication in idea generating groups, Universiteit Utrecht.

Nonaka, I 1994, _A dynamic theory of organizational knowledge creation',

Organization Science, vol. 5, no. 1, pp. 14-37.

Normann, R 1971, Organizational innovativeness: Product variation and reorientation⁶, *Administrative Science Quarterly*, vol. 16, pp. 203-215.

Norusis, MJ 1993, SPSS for Windows: Base system user's guide, Release 6. 0, Pearson Education, Chicago.

Nunnally, JC 1967, Psychometric Methods, McGraw Hill, New York.

Nunnally, JC & Bernstein, IH 1994, *Psychometric Theory*, McGraw Hill, New York.

Obstfeld, D 2005, _Social networks, the tertius iungens orientation, and involvement in innovation', *Administrative Science Quarterly*, vol. 50, no. 1, pp. 100-130.

O'Connor, GC & McDermott, CM 2004, _The human side of radical innovation', *Journal of Engineering and Technology Management*, vol. 21, no. 1, pp. 11-30.

O'Connor, GC, Paulson, AS & DeMartino, R 2008, _Organisational approaches to building a radical innovation dynamic capability', *International Journal of Technology Management*, vol. 44, no. 1, pp.179-204.

Ohly, S & Fritz, C 2009, _Work characteristics, challenge appraisal, creativity, and proactive behavior: A multi-level study', *Journal of Organizational Behavior*, vol. 31, no. 4, pp. 543-565.

Ohly, S, Kase, R & Škerlavaj, M 2010, _Networks for generating and for validating ideas: The social side of creativity', *Innovation: Management, Policy & Practice*, vol. 12, no. 1, pp. 41-52.

O'Leary, Z 2004, *The essential guide to doing research*. Sage Publications Limited.

Oldham, GR & Baer, M 2011, _Creativity and the work context⁴, in Mumford, MD (ed.), *Handbook of organizational creativity*, Elsevier, London, pp. 387-420.

Oldham, GR & Cummings, A 1996, <u>Employee Creativity: Personal and</u> Contextual Factors at Work', *Academy of Management Journal*, vol. 39, no. 3, pp. 607-634.

Onwuegbuzie, AJ & Johnson, RB 2006, _The validity issue in mixed research', *Research in the Schools*, vol. 13, no. 1, pp. 48-63.

Organ, DW & Greene, CN 1981, _The effects of formalization on professional involvement: a compensatory process approach', *Administrative Science Quarterly*, vol. 26, no. 6, pp. 237–252.

Oslo Manual 2005, *The Measurement of Scientific and Technological Activities. Proposed Guidelines for Collecting and Interpreting Technological Innovation Data'*, OECD, Paris.

Osman, MNH, Van Peursem, KA & Eggleton, I 2008, _Audit committees and internal auditors: Using LMX for relationship analysis⁴, Department of Accounting.

Pallant, J 2007, *SPSS: survival manual: a step-by-step guide to data analysis using SPSS for Windows (Version 15)*, 3rd end, Allen & Unwin, New South Wales.

Patton, MQ 2002, *Qualitative research and evaluation methods*, 3rd end, Thousand Oaks, CA: Sage Publishers.

Payne, M 1990, Modern social work theory: A critical introduction, Lyceum Books.

Pedersen, CR & Dalum, B 2004, June, _Incremental versus radical change—the case of the digital north Denmark program', *In International Schumpeter Society Conference*.

Pelz, DC 1956, Some social factors related to performance in a research organization', *Administrative Science Quarterly*, pp. 310-325.

Pelz, DC & Andrews, FM 1966, *Scientists in organization: Productive climates for research and development*, J. Wiley.

Perry-Smith, JE 2006, <u>Social Yet Creative: The Role of Social Relationships in</u> Facilitating Individual Creativity', *Academy of Management Journal*, vol. 49, no. 1, pp. 85-101.

Perry-Smith, JE & Shalley CE 2003, _The Social Side of Creativity: A Static and Dynamic Social Network Perspective', *Academy of Management Review*, vol. 28, no. 1, pp. 89-106.

Peter, JP 1979, _Reliability: A review of psychometric basics and recent marketing practices', *Journal of Marketing Research*, pp.6-17.

Phillips, DC & Burbules, NC 2000, *Postpositivism and educational research*, Rowman & Littlefield Pub Incorporated.

Pinchot III, G 1985, Intrapreneuring: Why you don't have to leave the corporation to become an entrepreneur', University of Illinois at Urbana-Champaign's Academy for Entrepreneurial Leadership Historical Research Reference in Entrepreneurship.

Pisano, G 1997, *The Development Factory: Unlocking the Potential of Process Innovation*, Harvard Business School Press: Boston.

Pires, CP, Sarkar, S & Carvalho, L 2008, _Innovation in services-how different from manufacturing^c, *The Service Industries Journal*, vol. 28, no. 10, pp. 1339-1356.

Podsakoff, PM, MacKenzie, SM, Lee, J & Podsakoff, NP 2003, Common method variance in behavioral research: A critical review of the literature and recommended remedies', *Journal of Applied Psychology*, vol. 88, no. 5, pp. 879-903.

Podsakoff, PM & Organ, DW 1986, _Self reports in organizational research: problems and prospects', *Journal of Management*, vol. 12, no. 4, pp. 531-544.

Poolton, J & Barclay, I 1998, _New product development from past research to future applications', *Industrial Marketing Management*, vol. 27, no. 3, pp. 197-212.

Popadiuk, S & Choo, CW 2006, <u>Innovation and knowledge creation</u>: How are these concepts related', *International Journal of Information Management*, vol. 26, no.4, pp. 302-312.

Prajogo, DI & Ahmed, PK 2006, _Relationships between innovation stimulus, innovation capacity, and innovation performance', *R&D Management*, vol. 36, no. 5, pp. 499-515.

Propris, LD 2002, _Types of innovation and inter-firm co-operation', *Entrepreneurship & Regional Development*, vol. 14, no. 4, pp. 337-353.

Pryor, KW, Haag, R & O'Reilly, J 1969, _The creative porpoise: Training for novel behavior', *Journal of the Experimental Analysis of Behavior*, vol. 12, pp. 653-661.

Ramamoorthy, N, Flood, PC, Slattery, T & Sardessai, R 2005, _Determinants of innovative work behaviour: development and test of an integrated model⁴, *Creativity and Innovation Management*, vol. 14, no. 2, pp. 142-150.

Rammer, C, Czarnitzki, D & Spielkamp, A 2009, Innovation success of non-R&D-performers: substituting technology by management in SMEs', *Small Business Economics*, vol. 33, no. 1, pp. 35-58.

Rank, J, Pace, VL & Frese, M 2004, _Three avenues for future research on creativity, innovation, and initiative', *Applied Psychology*, vol. 53, no. 4, pp. 518-528.

Reichstein, T & Salter, A 2006, _Investigating the sources of process innovation among UK manufacturing firms', *Industrial and Corporate Change*, vol. 15, no. 4, pp. 653-682.

Reiter-Palmono, R & Illies JJ 2004, Leadership and creativity: Understanding leadership from a creative problem-sloving perspective', *Organisational Behavior and Human Decision Process*, vol. 55, pp. 120-151.

Rice, G 2006, Individual values, Organizational Context, and self-perceptions of employee creativity: Evidence from Egyptian organizations', *Journal of Business Research*, vol. 59, no. 2, pp. 233-241.

Rich, JD & Weisberg, RW 2004, _Creating all in the family: A case study in creative thinking', *Creativity Research Journal*, vol. 16, no. 2-3, pp. 247-259.

Ridenour, CS& Newman, I 2008, *Mixed methods research: Exploring the interactive continuum*, Southern Illinois University Press.

Robbins, SP 1996, Organizational Behaviour: Concepts, controversies, applications, 7th edn, Englewood Cliffs, NJ: Prentice-Hall.

Robbins, SP & DeCenzo, D 2001, *Fundamentals of Management: Essential Concepts and Applications*, 3rd edn, Prentice Hall, Englewood Cliffs.

Rodan, S & Galunic, C 2004, _More than network structure: how knowledge heterogeneity influences managerial performance and innovativeness', *Strategic Management Journal*, vol. 25, no. 6, pp. 541-562.

Rogers, C 1954, Toward a theory of creativity', *Review of General Semantics*, vol.2, pp. 249- 260.

Rogers, EM 1983, *Diffusion of Innovations*, 3rd end, The Free Press, New York.

Rogers, EM 1995, Diffusion of Innovations, 4th edn, The Free Press, New York.

Rose, LH & Lin, HT 1984, _A meta-analysis of long-term creativity training programs', *The Journal of Creative Behavior*, vol. 18, no. 1, pp. 11-22.

Rowe, LA & Boise, WB 1974, _Organizational Innovation: Current Research and Evolving Concepts', *Public Administration Review*, vol. 34, no. 3, pp. 284-293.

Ryan, RM & Deci, EL 2000, _Self-determination theory and the facilitation of intrinsic motivation, social development, and well-being', *American Psychologist*, vol. 55, no. 1, pp. 68.

Rynes, S & Gerhart, B 2003, *Compensation: Theory, evidence, and strategic implications*. Sage Publications, Incorporated.

Samouel, M, Lewis, P & Thornhill, A 2000, *Research methods for business students*, 3rd edn, Prentice Hall, Harlow.

Sanchez, R & Mahoney, JT 1996, *Modularity, flexibility and knowledge management in product and organization design*, Graduate School of Management, University of Western Australia.

Sarantakos, S 2005, Social Research, Palgrave Macmillan, Basingstoke.

Sartorius, N & Kuyken, W 1994, _Translation of health status instruments', *Quality of Life Assessment: International Perspectives*, pp.3-18.

Scherbaum, CA, Finlinson, S, Barden, K & Tamanini, K 2006, _Application of item response theory to measurement issues in leadership research', *The Leadership Quarterly*, vol. 17, pp. 66-386.

Schermerhorn, JR & Nyaw, MK 1990, <u>Managerial leadership in Chinese</u> industrial enterprises', *International Studies of Management and Organisation*, vol. 20, pp. 9-21.

Schmierl, K & Köhler, HD 2005, _Organisational learning: knowledge management and training in low-tech and medium low-tech companies⁴, *Journal of Mental Changes*, vol. 11, no. 1-2, pp.171-221.

Schriesheim, CA 1979, _The similarity of individual-directed and group directed leader behavior descriptions', *Academy of Management Journal*, vol. 22, no. 2, pp. 345–355.

Schriesheim, CA, Scandura, TA, Eisenbach, RJ & Neider, LL 1992, _Validation of a new leader member exchange scale (LMX-6) using hierarchically nested maximum likelihood confirmatory factor analysis', *Educational and Psychological Measurement*, vol. 52, pp. 983-992.

Schroeder, R. G., Van de Ven, A. H., Scudder, G. D., & Polley, D. (1989). The development of innovation ideas. *Research on the management of innovation: The Minnesota studies*, 107-134.

Schuler, R & Jackson, S 1987, Linking competitive strategies with human resource practices', *Academy of Management Executive*, vol.1, no.3, pp.207-20.

Schwab, DP 2005. *Research methods for organisational studies*, 2nd edn, Lawrence Erlbaum Associates, Mahwah, NJ.

Scott, SG & Bruce, RA 1994, _Determinants of innovative behavior: A path model of individual innovation in the workplace', *Academy of management journal*, vol. 37, no. 3, pp. 580-607.

Scott, SG, Leritz, LE & Mumford, MD 2004, _The effectiveness of creativity training: A quantitative review', *Creativity Research Journal*, vol. 16, no. 4, pp. 361-388.

Seibert, SE, Kraimer, ML & Crant, JM 2006, What do proactive people do? A longitudinal model linking proactive personality and career success', *Personnel Psychology*, vol. 54, no. 4, pp. 845-874.

Sekaran, U 2003, *Research methods for business: A skill-building approach*, Willey, New York.

Shalley, CE 1995, Effects of coaction, expected evaluation, and goal setting on creativity and productivity', *Academy of Management Journal*, vol. 38, pp. 483-503.

Shalley, CE & Gilson, LL 2004, _What leaders need to know: A review of social and contextual factors that can foster or hinder creativity', *The Leadership Quarterly*, vol.15, pp.33-53.

Shalley, CE, Gilson, LL & Blum, TC 2000, _Matching creativity requirements and the work environment: Effects on satisfaction and intentions to leave', *Academy of Management Journal*, vol. 43, no. 2, pp. 215-223.

Shalley, CE, Zhou J & Oldham GR 2004, _The Effects of Personal and Contextual Characteristics on Creativity: Where Should We Go from Here', *Journal of Management*, vol. 30, no. 6, pp. 933-958.

Shane, S 1995, _Uncertainty avoidance and the preference for innovation championing roles', *Journal of International Business Studies*, pp. 47-68.

Shaughnessy, JJ & Zechmeister, EB 1997, *Research methods in psychology*, 4th edn, McGraw-Hill, New York.

Shi, J 2012, Influence of passion on innovative behavior: An empirical examination in Peoples Republic of China', *African Journal of Business Management*, vol. 6, no. 30, pp. 8889-8896.

Shin, SJ & Zhou, J 2003, _Transformational leadership, conservation, and creativity: Evidence from Korea['], *Academy of Management Journal*, vol. 46, no. 6, pp. 703-714.

Shipton, H, West, MA, Dawson, J, Birdi, K & Patterson, M 2006, <u>HRM as a predictor of innovation</u>, *Human Resource Management Journal*, vol. 16, no. 1, pp. 3-27.

Sias, PM 2005, _Workplace relationship quality and employee information experiences', *Communication Studies*, vol.56, no.4, pp. 375-395.

Silva, ND & Oldham GR 2012, _Adopting Employees' Ideas: Moderators of the Idea Generation–Idea Implementation Link', *Creativity Research Journal*, vol. 24, no.2-3, pp.134-145.

Simon, MB, Elango, B, Susan M, Houghton & Sonia Savelli 2002, _The Successful Product Pioneer: Maintaining Commitment while Adapting to Change', *Journal of Small Business Management* vol. 40, no. 3, pp. 187-203.

Simonton, DK 1976, <u>Ideological diversity and creativity</u>: A re-evaluation of a hypothesis', *Social Behavior and Personality: An International Journal*, vol. 4, no. 2, pp. 203-207.

Singleton, RA & Straits, BC, 1999, *Approaches to Social Research*, 3rd edn, Oxford University Press.

Smith, M, Busi, M, Ball, P & Robert VDM 2008, Factors influencing an organisation's ability to manage innovation: A structured literature review and conceptual model', *International Journal of innovation management*, vol. 12, no.4, pp. 655-676.

Sorescu, AB, Chandy, RK & Prabhu, JC 2003, _Sources and financial consequences of radical innovation: Insights from pharmaceuticals', *Journal of Marketing*, pp. 82-102.

Souder, WE 1987, *Managing New Product Innovations*, Lexington Books, Lexington MA.

Souder, WE & Moenaert, RK 1992, _An information uncertainty model for integrating marketing and R&D personnel in new product development projects', *Journal of Management Studies*, vol.29, no.4, pp. 485-512.

Souder, W.E. & Sherman, J.D. 1993 *Managing New Technology Development*. New York: McGraw-Hill.

Sparrowe, RT & Liden, RC 1997, _Process and structure in leader-member exchange', *Academy of management Review*, pp. 522-552.

Spector, PE 1994, Using self-report questionnaires in OB research: a comment on the use of a controversial method', *Journal of organizational behavior*, vol.15, no.5, pp. 385-392.

Stamm, B. von 2003, *Managing innovation, design & creativity*. London Business School, Wiley.

Sternberg, RJ & Lubart, TI 1996, Investing in creativity', *American psychologist*, vol. 51, no.7, pp. 677.

Sternberg, RJ & Lubart, TI 1999, _The concept of creativity: Prospects and paradigms', *Handbook of creativity*, vol. 1, pp. 3-15.

Stewart, DW 1981, _The application and misapplication of factor analysis in marketing research', *Journal of Marketing Research*, vol. 18, no. 2, pp. 51-62.

Stone, DN, Bryant, SM & Wier, B 2010, _Why are financial incentive effects unreliable? An extension of self-determination theory', *Behavioral Research in Accounting*, vol. 22, no.2, pp. 105-132.

Straub, D, Boudreau, MC & Gefen, D 2004, _Validation guidelines for IS positivist research', *Communications of the Association for Information Systems*, vol. 13, no. 24, pp. 380-427.

Subramaniam, M & Youndt, MA 2005, _The Influence of Intellectual Capital on the Types of Innovative Capabilities', *Academy of Management Journal*, vol. 48, no. 3, pp. 450-463.

Tabachnick, BG & Fidell, LS 2001, *Using multivariate statistics*, 4th edn, Allyn & Bacon, Boston, MA.

Tang, HK 1998, <u>An integrative model of innovation in organizations</u>, *Technovation*, vol. 18, no. 5, pp. 297-309.

Tannenbaum, S & Yukl, G 1992 Training and development in work organizations', *Annual Review of Psychology*, vol. 43, pp. 399-441.

Taylor, CW 1963, Clues to creative teaching: The creative process and education', *Instructor*, vol.73, no. 11, pp. 4-5.

Taylor, CW 1972, Climate for creativity: report, vol. 1966, Pergamon.

Teddlie, C & Tashakkori, A 2003, _Major issues and controveries in the use of mixed methods in the social and behavioral sciences', *Handbook of mixed methods in social & behavioral research*, pp. 3-50.

Teece, DJ 1986, _Profiting from technological innovation: Implications for integration, collaboration, licensing and public policy', *Research policy*, vol. 15, no.6, pp. 285-305.

Tellis, GJ, Prabhu, JC & Chandy, RK 2009, <u>Radical innovation across nations</u>: The preeminence of corporate culture', *Journal of Marketing*, vol. 73, no. 1, pp. 3-23.

Terence, J & Mette B 1998, _Foreign companies and Chinese workers: employee motivation in the People's Republic of China', *Journal of Organisational Change Management*, vol. 11, no. 4, pp. 282.

Therrien, P & L éonard A 2003, <u>Empowering employees</u>: A route to innovation', *The Evolving Workplace Series*.

Thompson, VA 1965, _Bureaucracy and innovation', *Administrative Science Quarterly*, pp. 1-20.

Thornton, D, Kagan, RA & Gunningham, M 2003, _Sources of corporate environmental performance', *California Management Review*, vol. 46, no.1, pp. 127-141.

Tidd, J 2001a, _Innovation management in context: Environment, organisation and performance', *International Journal of Management Reviews*, vol. 3, no. 3, pp. 169-183.

Tidd, J, Bessant, J & Pavitt, K 2001, *Managing Innovation: Integrating Technological, Market and Organisational Change*, 2nd edn, John Wiley, Chichester.

Tierney, P & Farmer, SM 2002, _Creative Self-Efficacy: Its Potential Antecedents and Relationship to Creative Performance', *Academy of Management Journal*, vol. 45, no. 6, pp. 1137-1148.

Tierney, P & Farmer, SM 2004, _The Pygmalion process and employee creativity', *Journal of Management*, vol. 30, no. 3, pp. 413-432.

Tierney, P, Farmer, SM & Graen, GB 1999, _An examination of leadership and employee creativity: The relevance of traits and relationships', *Personnel Psychology*, vol. 52, no. 3, pp. 591-620.

Tjosvold D & McNeely, LT 1988, Innovation through communication in an educational bureaucracy', *Communication Research*, vol. 15, no. 5, pp. 568-581.

Toivonen, M & Tuominen, T 2009, _Emergence of innovations in services', *Service Industries Journal*, vol. 29, no. 7, pp. 887-902.

Tortoriello M & Krackhardt D 2010, _Activating cross-boundary knowledge: The role of simmelian ties in the generation of innovations', *Academy of Management Journal*, vol. 53, no. 1, pp. 167-181.

Trist, EL & Bamforth, KW 1951, Some Social and Psychological Consequences of the Longwall Method.

Tushman, ML 1977, _Special boundary roles in the innovation process', *Administrative Science Quarterly*, vol. 22, pp. 587-605.

Tushman, ML 1979, _Work characteristics and subunit communication structure: A contingency analysis', *Administrative Science Quarterly*, pp. 82-98.

Tushman, ML & Nadler, D 1986, Organisation for innovation', *California Management Review*, vol. 28, pp. 74-92.

Ulrich, KT & Eppinger, SD 1995, *Product design and development*, vol. 384, McGraw-Hill, New York.

Unsworth, KL 1999, *Employee innovation: An exploratory test of a two-factor mediated model*⁴, no. 201, IWP Memo, Sheffield: Institute of Work Psychology.

Unsworth, K 2001, Unpacking Creativity', *The Academy of Management Review*, vol. 26, no. 2, pp. 289-297.

Unsworth, KL, Brown, H & McGuire, L 2000, <u>Employee Innovation: The roles</u> of idea generation and idea implementation', *In: Proceedings of the Annual Conference of the Society for Industrial and Organizational Psychology*, New Orleans.

Unsworth, K & West, MA 1998, July, _Employee innovation: Generation, implementation or both', In *International Work Psychology Conference*, University of Sheffield, Institute of Work Psychology.

Urabe, K 1988, Innovation and the Japanese management system⁴, In Urabe K, Child J, & Kagono T (eds.), *Innovation and management international comparisons*, Walter de Gruyter, Berlin.

Utterback, JM 1994, Mastering the Dynamics of Innovation: How Companies Can Seize Opportunities in the Face of Technological Change, Harvard Business School Press, Boston, MA.

Van Breukelen, W, Schyns, B & Blanc, PL 2006, Leader-member exchange theory and research: Accomplishments and future challenges', *Leadership*, vol. 2, no. 3, pp. 295-316.

Van de Ven, AH 1986, _Central problems in the management of innovation', *Management science*, vol. 32, no. 5, pp. 590-607.

Van de Ven, AH, Angle, HL & Poole, MS 1989, *Research on the Management of Innovation*, Harper and Row, New York.

Van der Vegt, GS & Janssen, O 2003, _Joint impact of interdependence and group diversity on innovation', *Journal of Management*, vol. 29, no. 5, pp. 729-751.

Van der Velde, M, Jansen, P & Anderson, N 2004, *Guide to management research methods*, Oxford: Blackwell Publishing.

Veal, AJ & Ticehurst, GW 2005, *Business research methods: A managerial approach*, Pearson/Addison Wesley.

Velasco, E, Zamanillo, I & Del Valle, TG 2012, _Mobilizing company members' full innovative potential', *Human Factors and Ergonomics in Manufacturing & Service Industries*.

Vermeulen, DPA, De Jong, JP & O'shaughnessy, KC 2005, _Identifying key determinants for new product introductions and firm performance in small service firms', *The Service Industries Journal*, vol. 25, no. 5, pp. 625-640.

Von Hippel, E 1988, <u>The sources of innovation</u>, *University of Illinois at Urbana-Champaign's Academy for Entrepreneurial Leadership Historical Research Reference in Entrepreneurship*.

Von Hippel, E 2005, _Democratizing innovation: The evolving phenomenon of user innovation', *Journal für Betriebswirtschaft*, vol. 55, no. 1, pp. 63-78.

Walker, RM 2008, <u>An Empirical Evaluation of Innovation Types and</u> Organizational and Environmental Characteristics: Towards a Configuration Framework', *Journal of Public Administration Research and Theory*, vol. 18, no. 4, pp. 591-615.

Wang, CW & Horng, RY 2002, _The effects of creative problem solving training on creativity, cognitive type and R&D performance', *R&D Management*, vol. 32, no. 1, pp. 35-45.

Wang, H, Law, KS & Chen, ZX 2008, Leader-member exchange, employee performance, and work outcomes: An empirical study in the Chinese context⁴, *The International Journal of Human Resource Management*, vol. 19, no. 10, pp. 1809-1824.

Wang, Z & Zang, Z 2005, _Strategic human resources, innovation and entrepreneurship fit: a cross-regional comparative model⁴, *International journal of Manpower*, vol. 26, no. 6, pp. 544-559.

Warr, P, Allan, C & Birdi, K 1999, 'Predicting three levels of learning outcome', *Journal of Occupational and Organizational Psychology*, vol.72, pp. 351-375.

Weenig, MW 1999, <u>Communication Networks in the Diffusion of an Innovation</u> in an Organization1⁺, *Journal of Applied Social Psychology*, vol. 29, no. 5, pp.1072-1092. Weerawardena, J 2003, _Exploring the role of market learning capability in competitive strategy', *European Journal of Marketing*, vol. 37, no. 3/4, pp. 407-429.

Wenger, EC & Snyder, WM 2000, _Communities of practice: The organizational frontier', *Harvard business review*, vol. 78, no. 1, pp. 139-146.

West, MA 1990, _The social psychology of innovation in groups', In MA West and JL Farr (eds.), Innovation and Creativity at Work: Psychological and Organizational Strategies, pp. 309-335. Chichester: Wiley.

West, M 1997, Developing Creativity in Organizations: Personal and Professional Development, Wiley-Blackwell.

West, MA 2000, _Creativity and Innovation at Work', *Psychology*, vol. 81, pp. 680-693.

West, MA 2002, _Sparkling fountains or stagnant ponds: An integrative model of creativity and innovation implementation in work groups', *Applied Psychology: An International Review*, vol. 51, no. 3, pp. 355-424.

West, MA & Anderson N 1996 Innovation in top management teams', *Journal of Applied Psychology*, vol. 81, no. 6, pp. 680-693.

West, MA & Farr, JL 1989, Innovation at work: psychological perspectives', *Social Behaviour*, vol. 4, pp.15-30.

West, MA & Farr JL 1990, *Innovation and Creativity at Work*. Chichester: Wiley. Willem, A & Buelens, M 2009, _Knowledge sharing in inter-unit cooperative episodes: The impact of organizational structure dimensions⁴, *International Journal of Information Management*, vol. 29, no. 2, pp. 151-160.

Williams, WM & Yang, LT 1999, Organizational creativity', *Handbook of creativity*, pp. 373-391.

Winston, AS & Baker, JE 1985, _Behavior analytic studies of creativity: A critical review', *The Behavior Analyst*, vol. 8, no. 2, pp. 191.

Wolfe, RA 1994, _Organisational innovation: Review, Critique and Suggested research', *Journal of Management Studies*, vol. 31, no.3, pp. 405-431.

Wolfe, RA 1995, _Human resource management innovations: Determinants of their adoption and implementation', *Human Resource Management*, vol. 34, no. 2, pp. 313-327.

Woodman, RW, Sawyer, JE & Griffin, RW 1993, _Toward a theory of organisational creativity', *Academy of Management Review*, vol.18, pp. 293-321.

Xiaojun, L, & Peng, L 2010, August, _The Impact of Learning Culture on Individual Innovative Behavior', In *Management and Service Science (MASS)*, 2010 International Conference on (pp. 1-4). IEEE.

Yang, JT 2007, <u>Knowledge sharing</u>: Investigating appropriate leadership roles and collaborative culture', *Tourism Management*, vol. 28, no. 2, pp. 530-543.

Yang, Y & Konrad, AM, _Diversity and organizational innovation: The role of employee involvement', *Journal of Organizational Behavior*.

Yuan, F & Woodman, RW, 2010 _Innovative Behaviour in the Workplace: The role of Performance and Image Outcome Expectations', *Academy of Management*, vol. 53, no. 2, pp 323-342.

Yukl, G 2002, *Leadership in Organisations*, 5th edn, Prentice-Hall, Englewood Cliffs, NJ.

Yusuf, Y, Gunasekaran, A & Dan, G 2007, Implementation of TQM in China and organisation performance: an empirical investigation', *Total quality management*, vol. 18, no. 5, pp. 509-530.

Zheng, W_A Social Capital Perspective of Innovation from Individuals to Nations: Where is Empirical Literature Directing Us', *International Journal of Management Reviews*, vol.12, no. 2, pp. 151-183.

Zhou, J & Shalley, CE 2003, _Research on employee creativity: A critical review and directions for future research', *Research in personnel and human resources management*, vol. 22, pp.165-217.

Zhou, J & George JM 2001, _When job dissatisfaction leads to creativity: Encouraging the expression of voice', *Academy of Management Journal*, vol. 44, no.4, pp. 682–696.

Zhou, KZ & Li CB 2012, _How knowledge affects radical innovation: Knowledge base, market knowledge acquisition, and internal knowledge sharing', *Strategic Management Journal*, vol. 33, no.9, pp. 1090-1102.

Zhou, Y, Zhang, Y & Montoro-S ánchez, Á 2011, _Utilitarianism or romanticism: the effect of rewards on employees' innovative behaviour', *International Journal of Manpower*, vol.32, no.1, pp. 81-98.

Zikmund, WG, 2003. *Business research methods*, Thomson South-western, Mason, Ohio.

Zirger, BJ & Maidique, MA 1990, _A model of new product development: an empirical test', *Management Science*, vol. 36, no. 7, pp.867-883.

Appendix A: Survey Instrument

QUESTIONNAIRE SURVEY

Facilitate Individual Innovation in Organisations

Dear employees,

You are invited to participate in a research project being conducted in partial fulfilment of a Master's degree for one of the investigators, Jingjing Wang under the supervision of the chief investigator, Dr Fan Liang, and the co-investigator, Dr Megan Woods, of the School of Management at the University of Tasmania. The research aims to identify key corporate factors that facilitate individual innovation at workplace and explore how companies can facilitate individual innovation by leveraging available corporate resources.

This questionnaire consists of six sections and it is anticipated that it will take no more than fifteen minutes to complete. Please tick the most appropriate option using the scale that follows or write your answer if needed. You are encouraged to complete the questionnaire in an honest way as your answers would greatly constitute to our research. Your participation in this study is entirely voluntary. You have the right to decline. If you wish to take part in it, please complete the questionnaire and put it into the envelope provided for you and return it through your managers who are in charge of distributing and collecting the questionnaires.

All answers to questions in this questionnaire remain anonymous and all information will be treated confidentially. The survey data will be used for statistical analyses only and you will not be identified in the research. All the completed questionnaires will be kept securely in a locked cabinet in the office of School of Management at University of Tasmania. All data will be stored for five years and then securely destroyed.

If you would like to discuss any aspect of this study please feel free to contact Jingjing Wang. This study has been approved by the Tasmanian Social Science Human Research Ethics Committee. Any concerns or complaints about this study can be directed to the Executive Officer of the HREC (Tasmania) Network on +61 3 6226 7479 or email to human.ethics@utas.edu.au.You will need to quote ethics reference number H11495.

Thank you for your cooperation.

Return of completed questionnaire through your managers is an indication that you have given your consent to participate in this study.

Kind regards,

Jingjing Wang

Contact Address in China:

Building 103, Residential compound of Shandong Agricultural University

Tai'an city Shandong China 271000

Phone: +86 15165482232

Email: jwang5@postoffice.utas.edu.au

Section A: General information

1.	Your age:					
2.	Gender:					
3.	Your highest educ	ational qualification	n (pleas	e circle o	one):	
	(1) High school		(2) Te	chnical &	& Vocational	Institutions
	(3) Graduate diplo	oma or certificate	(4) Ba	chelor		
	(5) Master		(6) Ph	D		
4.	The industry of yo	our organisation (pl	ease cir	cle one):		
	(1) Manufacturing	5	(2) Se	rvices		
	(3) Wholesale and	l retail trade	(4) Ed	lucation		
	(5) Infrastructure		(6) En	ergy		
5.	The ownership typ	pe of your organisat	tion (ple	ease circl	e one):	
	(1) State-owned e	nterprise	(2) Co	ollective-	owned enter	prise
	(3) Cooperative en	nterprise	(4) Jo	int owne	rship enterpr	rise
	(5) Limited Liabil	ity Corporation	(6) Sh	are-hold	ing Corporat	tions Ltd.
	(7) Private Enterp	rise	(8) Jo	int ventu	re with invest	stment from Hong
	Kong, Macao and	Taiwan & Enterpri	ses with	n foreign	investment	
6.	The number of en	ployees your organ	nisation	has:		
	(1)<5	(2) 5-20	(3) 21	-100		
	(4) 101-200	(5) 201-500	(6) 50	1-100	(7) ≥	1001
7.	Your organisation	was established in	Year	; or it	has operated	for (please circle
	one):					
	(1)<1 year	(2) 1-5 year	s	(3)) 6-10 years	
	(4) 11-20 years	(5) > 20 year	rs			
8.	The area you worl	k in (please circle o	ne):			
	(1) Sales	(2) Advertising &	z marke	ting	(3) Finance	& accounting
	(4) R&D	(5) Human resou	rces		(6) General	administration
	(7) Technical	(8) Production &	operati	on	(9) other (sp	pecify)
9.	How long have yo	bu been employed in	n this or	ganisatio	on (please cir	rcle one)?
	(1) < 3 months	(2) 3 months-1 ye	ear	(3) 1-3y	/ears	
	(4) 3-5 years	(5) 5-10 years		(6) 10-2	20 years	(7) > 20 years

Please circle the most appropriate answer for each question.

Section B: In your current work, how often do you:

10.	. Look for opportunities to improve an existing process at work?					
	(Never)	(Seldom)	(Sometimes)			
	(Often)	(Fairly often)	(Very often)	(Always)		
11.	Recognise oppo	rtunities to make a pos	sitive difference in y	our work?		
	(Never)	(Seldom)	(Sometimes)			
	(Often)	(Fairly often)	(Very often)	(Always)		
12.	Pay attention to	non-routine issues in	your work?			
	(Never)	(Seldom)	(Sometimes)			
	(Often)	(Fairly often)	(Very often)	(Always)		
13.	Generate ideas	or solutions to address	problems you find in	n your work?		
	(Never)	(Seldom)	(Sometimes)			
	(Often)	(Fairly often)	(Very often)	(Always)		
14.	Look at the big	picture to gain greater	insight into them?			
	(Never)	(Seldom)	(Sometimes)			
	(Often)	(Fairly often)	(Very often)	(Always)		
15.	Experiment with	h new ideas?				
	(Never)	(Seldom)	(Sometimes)			
	(Often)	(Fairly often)	(Very often)	(Always)		

16.	Test-out the so	lutions for current pr	oblems?				
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
17.	Evaluate the st	rengths and weaknes	s of new ideas?				
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
18.	Implement cha	nges that seem to ma	ike your job easier o	or more efficient?			
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
19.	Work the bugs	out of new approach	es when applying th	nem to an existing process?			
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
20.	Incorporate ne	w ideas for improvin	g an existing proces	ss into daily routines?			
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
21.	Try to persuade	e others of the import	tance of a new idea	or solution?			
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
22.	Put ideas forwa	ard to colleagues so t	hat they have a char	nce to become implemented			
	by others?						
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
23.	Put ideas forward to supervisor-level people so they have a change to become						
	implemented at the corporate level?						
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
24.	Take the risk to	o promote own new i	deas?				
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
25.	Take the risk to support other colleagues' new ideas?						
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
26.	Take the risk to	o promote own new i	deas but not the risk	to support other colleagues			
	new ideas?	-					
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			
27.	Take the risk to	o support other collea	igues' new ideas bu	t not the risk to generate &			
	promote own n	new ideas?		-			
	(Never)	(Seldom)	(Sometimes)				
	(Often)	(Fairly often)	(Very often)	(Always)			

Section C: Supervisor-employee relationship

28.	3. You know where you stand with your supervisor.					
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)		
	(Neutral agree)	(Strongly agree)	(Totally agree)			
29.	You know how sa	atisfied your supervis	or is with what you o	lo.		
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)		
	(Neutral agree)	(Strongly agree)	(Totally agree)			
30.	Your supervisor u	inderstands your job	problems and needs.			
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)		
	(Neutral agree)	(Strongly agree)	(Totally agree)			
31.	Your supervisor c	an recognise your po	tential.			
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)		
	(Neutral agree)	(Strongly agree)	(Totally agree)			

32.	2. Your supervisor uses his or her power to help you solve problems in your work.							
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)				
	(Neutral agree)	(Strongly agree)	(Totally agree)					
33.	Your supervisor v	will _bail you out' at	his or her expense.					
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)				
	(Neutral agree)	(Strongly agree)	(Totally agree)					
34.	You have enough	confidence in your	supervisor that you	would defend and justify his				
	or her decision if	he or she were not p	present to do so.					
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)				
	(Neutral agree)	(Strongly agree)	(Totally agree)					
35.	You would chara	cterize your working	g relationship with yo	our supervisor as effective.				
	(Totally disagree)	(Strongly disagree)	(Neutral disagree)	(Neither agree nor disagree)				

(Neutral agree) (Strongly agree) (Totally agree)

Section D: Communication at workplace

In the same work area, you: 36. Communicate with employees who have similar job tasks. (Daily) (Not at all) (Weekly) (Monthly) (Quarterly) (Half yearly) (Yearly) 37. Communicate with employees who have different job tasks. (Daily) (Weekly) (Monthly) (Not at all) (Quarterly) (Half yearly) (Yearly) In different work area from yours, you: 38. Communicate with employees who have similar job tasks. (Not at all) (Daily) (Weekly) (Monthly) (Quarterly) (Half yearly) (Yearly) 39. Communicate with employees who have different job tasks. (Not at all) (Daily) (Weekly) (Monthly) (Quarterly) (Half yearly) (Yearly) 40. How many people do you interact with in your daily work within your organisation? (4) 7-9 (6) 13-15 (1) None (2) 1-3 (3) 4-6 (5) 10-12 (7) >15 For communication with outsiders of your organisation for work, you: 41. Visit conferences, trade fairs and/or expositions, on your organisation's behalf. (Not at all) (Daily) (Weekly) (Monthly) (Quarterly) (Half yearly) (Yearly) 42. Keep in touch with customers, on your organisation's behalf. (Not at all) (Weekly) (Monthly) (Daily) (Quarterly) (Half yearly) (Yearly) 43. Keep in touch with other companies in the same industry, on your organisation's behalf. (Not at all) (Daily) (Weekly) (Monthly) (Quarterly) (Half yearly) (Yearly) 44. Keep in touch with research institutions/universities, on your organisation's behalf. (Not at all) (Daily) (Weekly) (Monthly) (Quarterly) (Half yearly) (Yearly)

Section E: Rewards and recognition

- 45. Which of the following tangible rewards does your organisation offer for your innovative contribution work?
 - (1) Increased basic wage
 - (2) Company performance-related compensation (e.g. stock options)
 - (3) Bonuses

- (4) Various allowances (e.g. overtime payment)
- (5) Social wage (in forms of labour insurance and collective welfare benefits)
- (6) Symbolic support (e.g. gift)
- 46. Have you ever received any tangible rewards for your innovative contributions at work?

(1) Yes (2) No

If yes, how often the following rewards are you received for your innovative contribution in the current organisation?

47. Increased basic wages

	(Never)	(Seldom)	(Sometimes)	
	(Often)	(Fairly often)	(Very often)	(Always)
48.	Company pe	erformance-related co	mpensation (e.g. sto	ck options)
	(Never)	(Seldom)	(Sometimes)	
	(Often)	(Fairly often)	(Very often)	(Always)
49.	Bonuses			
	(Never)	(Seldom)	(Sometimes)	
	(Often)	(Fairly often)	(Very often)	(Always)
50.	Various allo	wances (e.g. overtime	e payment)	
	(Never)	(Seldom)	(Sometimes)	
	(Often)	(Fairly often)	(Very often)	(Always)
51.	Social wage	(in forms of labour in bridges)	nsurance and collecti	ve welfare benefits, e.g.

	transport su	Usidies)		
	(Never)	(Seldom)	(Sometimes)	
	(Often)	(Fairly often)	(Very often)	(Always)
52.	Symbolic su	upport (e.g. gift)		
	(Never)	(Seldom)	(Sometimes)	
	(Often)	(Fairly often)	(Very often)	(Always)

53. Which of the following rewards are most attractive to you?

- (1) Increased basic wage
- (2) Company performance-related compensation (e.g. stock options)
- (3) Bonuses
- (4) Various allowances (e.g. overtime payment)
- (5) Social wage (in forms of labour insurance and collective welfare benefits)
- (6) Symbolic support (e.g. gift)
- 54. Which of the following intangible recognition does your organisation offer for your innovative contribution work?
 - (1) Promotion
 - (2) Honour (e.g., public praise, model worker title, compliment, a crowd cheering)
- 55. Have you ever received any intangible recognition for your innovative contributions at work?
 - (1) Yes (2) No

If yes, how often the following intangible recognition are you received for your innovative contribution in the current organisation?

56.	Promotion				
	(Never)	(Seldom)	(Sometimes)		
	(Often)	(Fairly often)	(Very often)	(Always)	
57.	Honour (e.g., p	ublic praise, model	l worker title, compli	ment, a crowd cheering)	
	(Never)	(Seldom)	(Sometimes)		
	(Often)	(Fairly often)	(Very often)	(Always)	
58. Which of the following rewards are most attractive to you?					
	(1) Promotion				

(2) Honour (e.g., public praise, model worker title, compliment, a crowd cheering)

Section F: Training

How often does your organisation:

59.	9. Provide you with on-the-job training, to update your skills?				
	(Never)	(Seldom)	(Sometimes)		
	(Often)	(Fairly often)	(Very often)	(Always)	
60.	Send you to any	sort of course, to upd	ate your skills?		
	(Never)	(Seldom)	(Sometimes)		
	(Often)	(Fairly often)	(Very often)	(Always)	
61.	Provide you wit	h planned activity for	development?		
	(Never)	(Seldom)	(Sometimes)		
	(Often)	(Fairly often)	(Very often)	(Always)	
62.	How much train	ing do you receive in	the past 12 months?		
	(Never)	(Seldom)	(Sometimes)		
	(Often)	(Fairly often)	(Very often)	(Always)	
mke	for your time i	n completing this sur	VAV		

Thanks for your time in completing this survey.

Appendix B: Survey Instrument (Chinese Version)

敬爱的雇员,

- 您被邀请参与由来自澳大利亚塔斯马尼亚大学管理学院的王静静为了完成其博士 课题而和她的两位导师,梁博士和梅根博士组织的问卷调查。本研究课题旨 在探索可以鼓励员工创新工作行为的企业因素,力图帮助企业探求如何通过 企业内可利用资源来促进员工创新行为的方法。
- 本问卷包含六个部分,预计在 15 分钟内完成。请选出您认为最恰当选项,或者在 需要的地方写下您的答案。
- 本次问卷调查是完全自愿的。您完全有权利拒绝参加。如果您愿意参与本次调 查,请您把完成的问卷放入预先提供的信封里,然后归还给负责发放问卷的 负责人即可。
- 本研究对所有参与者保留匿名,对参与者所填写的所有内容予以绝对保密,本次 问卷调查所得到的所有数据仅用于本次研究的统计分析,不会涉及识别个人 隐私,决不另作其他用途。本研究的所有资料将会被安全的锁在位于澳大利 亚塔斯马尼亚大学管理学院的办公室里5年,5年后,所有资料会被安全销 毁。
- 如果您希望就本研究的任何一方面进行讨论,欢迎您随时与王静静联系。本研究 已经经塔斯马尼亚社会研究人类研究道德委员会批准。如果您有任何关于本 研究的要求或投诉,您可以直接与该道德委员会的执行人员联系(电话+61 3 6226 7479 或电邮至 <u>human.ethics@utas.edu.au</u>)。届时您需要提供道德编 号 H11495。
- 非常感谢您的参与。 您归还完成的问卷将被认为是您愿意参

与本次调查的体现。

此致

王静静

调查问卷:企业创新机制

A: 基本信息

1.	您的年龄:	岁	
2.	性别:		
3.	您的最高学历(单选	√):	
	(1)高中及以下	(2)中专或技校	(3)大学专科
	(4)大学本科	(5)硕士	(6)博士
4.	您所在单位所属行业	(单选 √):	
	(1)制造业	(2)服务业	(3)批发零售业
	(4)教育业	(5)基础设施行业	(6)能源
5.	您所在单位的属性(单选 √):	
	(1)国有企业	(2)集体企业	(3)联营企业
	(4)有限责任公司	(5)中外股份合作合资企业	(6)港澳台股份合作合资企业
	(7)外商独资企业	(8)港澳台独资企业	(9)私营企业
6.	您所在单位的员工人类	敋(人)(单选 √):	
	(1)<5	(2) 5-20	(3) 21-100
	(4) 101-200	(5) 201-500	(6) 501-1000
	$(7) \ge 1001$		
7.	您所在单位成立于	_年;或者它已经运营了	(单选 √):
	(1)<1年	(2) 1-5年	(3) 6-10年
	(4) 11-20年	(5)>20年	
8.	您的工作领域(单选	√): 	
	(1)销售	(2)市场或广告	(3)财务
	(4)研发	(5)人力人事	(6)行政
	(7)专项技术	(8)生产或运营	(9)其他
9.	您在现在这个单位的工	工作时间(单选 √):	
	(1)<3个月	(2)3个月-1年	(3) 1-3年
	(4) 3-5年	(5) 5-10年	(6) 10-20年
	(7)>20年		

请回答每道问题,选出您认为最恰当选向(单选√):

B:

五字》 在你的工作由《家女》							
在忽的工作中, 忍多人。	从不	很少	有	经	相	非	总是
10. 寻找可以改进现有工作的机会?					_		_
	1	2	3	4	5	6	7
11. 息识到有纪念工作市米积极受化的 机					_		_
	1	2	3	4	5	6	7
12. 仕工1F中由息非市观的争分;		2	2		-	(7
	1	2	3	4	5	0	1
13.) 土针对芯工作中现任问题的怎么。 武解决方案?	1	2	2	1	5		7
	1	4	3	4	3		/
深层了解?	1	2	3	4	5	6	7
15. 尝试新点子?	1		5				,
	1	2	3	4	5	6	7
16. 检验针对现有问题的解决方案?			-		-	-	
	1	2	3	4	5	6	7
17. 评价新点子的优点和缺点?							
	1	2	3	4	5	6	7
18. 实施那些看起来可以让您的工作更							
简单更有效率的变化?	1	2	3	4	5	6	7
19. 排除在现有工作过程中应用新方法							
时的障碍?	1	2	3	4	5	6	7
20. 应用新点子去改进现有工作过程?							
	1	2	3	4	5	6	7
21. 试图说服别人关于您的新点子或者							
一件状力系的里安性;	1	2	3	4	5	6	7
22. 枳极把新点子推荐给同事使它们有							_
	1	2	3	4	5	6	7
23. 积极把新点于推存给上级使它们有 机会在公司范围内应田?	1	2	2	4	F		-
	1	2	3	4	5	0	/
	1	2	2	А	=	4	7
25 原音承受风险去推行促进自己的新	1	4	3	4	5	0	/
	1	2	3	4	5	6	7
26. 愿意承受风险去支持同事的新点			5		5	0	,
子?	1	2	3	4	5	6	7
	-	-	v	•	÷	v v	,

C:上下级关系:

	强烈	不同	有	既	有	同	强烈同 意
27. 您跟您上司在工作中相互支持。	1	2	3	4	5	6	7
28. 您和您上司的工作关系非常有效 率。	1	2	3	4	5	6	7
29. 您知道上司对您工作表现的满意程 度。	1	2	3	4	5	6	7
30. 您的上司理解您工作中的问题和需 求。	1	2	3	4	5	6	7
31. 您的上司能意识到您的潜能。	1	2	3	4	5	6	7
32. 您的上司会运用他的权利帮您解决 您工作中的问题。	1	2	3	4	5	6	7
33. 您的上司愿意承担责任帮您渡过难 关。	1	2	3	4	5	6	7
34. 您对您上司的决定有信心,即使上 司不在场时您也愿意维护支持他的 决定。	1	2	3	4	5	6	7
35. 您目前与上级的关系有助于您在工作中:							
a. 产生新点子。	1	2	3	4	5	6	7
b. 实施您的新点子。	1	2	2		5	6	7
c. 传播扩散您的新点子。	1	2	3	4	5	6	7

D:工作中的交往:

在同一个工作领域或工作部门,	从不	每天	每周	每月	每季度	半年	一年一次
787. 36. 与有类似工作任务的员工交							
流。	1	2	3	4	5	6	7
37. 与工作任务和你不同的员工交							
流。	1	2	3	4	5	6	7
与你不同工作领域或工作部门的 人,您:							
38. 与有类似工作任务的员工交流							
	1	2	3	4	5	6	7
39. 与工作任务和你不同的员工交							
流。	1	2	3	4	5	6	7
40. 在你的日常工作中,你每天大概跟几个人接触交流?	1=	2=1-	3=4-	4=7-	5=10-	6=13-	7=多
							1 5
与你公司以外的人的工作交流, 您:	从不	每天	每周	每月	每季度	半年	一年一次

41.	代表公司参加会议,展销会或							
	者博览会等活动。	1	2	3	4	5	6	7
42.	代表公司与顾客保持联系。							
		1	2	3	4	5	6	7
43.	代表公司与同行业其他公司保							
	持联系。	1	2	3	4	5	6	7
44.	代表公司与研究机构 / 大学保							
	持联系。	1	2	3	4	5	6	7
45.	您目前工作中的交往有助于您							
	在工作中:	强	不同	有	既	有点	同意	强烈同
	a. 产生新点子。							
		1	2	3	4	5	6	7
	b. 实施您的新点子。							
		1	2	3	4	5	6	7
	c. 传播扩散您的新点子。							
		1	2	3	4	5	6	7

E: 奖赏和表扬:

46.	如射到一	果您工作中有创新,您多久会得 −次如下奖赏和表扬?	从不	很少	有	经	相	非	总是
				10.2					
	а	增加基本工资							
	u.		1	2	3	4	5	6	7
	b.	股票							
			1	2	3	4	5	6	7
	c.	奖金							
		··-	1	2	3	4	5	6	7
	d.	补贴(如加班费)							
			1	2	3	4	5	6	7
	e.	社会保障(如劳动保险,集体							
		福利)	1	2	3	4	5	6	7
	f.	晋升	-			-			
		171	1	2	3	4	5	6	7
	σ	茔喾(如公开表扬 莹动标丘	1						,
	5.	称号、为您举办庆祝活动)	1	2	2	1	5	6	7
	h	各次性农时(加加加尔斯)	1	<u>∠</u>	3	4	5	0	1
	n.	家仙住关励(如他初寺头初欧 励)					_		_
			1	2	3	4	5	6	7
47.	窓い	人刃哪──种奖赏和表扬对您有吸 	고머포네	-	<u>+</u>	07	<i>+</i>		
	517] ?	通烈	个回	月	跣	月	回	通然问
									意
1									
1									

a.	增加基本工资							
		1	2	3	4	5	6	7
b.	股票							
		1	2	3	4	5	6	7
с.	奖金							
		1	2	3	4	5	6	7
d.	补贴(如加班费)							
		1	2	3	4	5	6	7
e.	社会保障(如劳动保险,集体							
	福利)	1	2	3	4	5	6	7
f.	晋升							
		1	2	3	4	5	6	7
g.	荣誉(如公开表扬,劳动标兵							
	称号,为您举办庆祝活动)	1	2	3	4	5	6	7
h.	象征性奖励(如礼物等实物鼓							
	励)	1	2	3	4	5	6	7
48. 您	目前所获得的奖赏和表扬,有助							
于	您在工作中:							
a.	产生新点子。							
		1	2	3	4	5	6	7
b.	实施您的新点子。							
		1	2	3	4	5	6	7
с.	传播扩散您的新点子。							
		1	2	3	4	5	6	7

F:培训:

您的公司:	强烈	不同	有	既	有	同	强烈同意
				-		-	
49. 为您提供在职培训来提高你的业务 水平和技能。	1	2	3	4	5	6	7
50. 派您去参加一些课程来提高业务水 平和技能	1	2	3	4	5	6	7
51 为您提供甘他有利于你职业发展的	1	2	3	4	5	6	7
活动。		-	Ũ	•	č	Ŭ	,
52. 过去一年您公司为您提供的培训机 会满足您的需求。	1	2	3	4	5	6	7
53. 您目前所接受过的培训有助于您在 工作中:							
a. 产生新点子。							
	1	2	3	4	5	6	7
b. 实施您的新点子。							
	1	2	3	4	5	6	7
c. 传播扩散您的新点子。							
	1	2	3	4	5	6	7

谢谢您的热心参与!